DATA ANALYSIS IN COMMUNITY AND LANDSCAPE ECOLOGY

Edited by

R. H. G. JONGMAN, C. J. F. TER BRAAK & O. F. R. VAN TONGEREN



Published by the Press Syndicate of the University of Cambridge The Pitt Building, Trumpington Street, Cambridge CB2 1RP 40 West 20th Street, New York, NY 10011-4211, USA 10 Stamford Road, Oakleigh, Melbourne 3166, Australia

Cambridge University Press 1995

First published by Pudoc (Wageningen) 1987 New edition with corrections published by Cambridge University Press 1995

Printed in Great Britain at Biddles

A catalogue record for this book is available from the British Library

Library of Congress cataloguing in publication data

Data analysis in community and landscape ecology / R.H.G. Jongman, C.J.F. ter Braak, and O.F.R. van Tongeren, editors. – New ed.

p. cm. Includes bibliographical references (p.) and index. ISBN 0 521 47574 0 (pbk.)

 Biotic communities-Research-Methodology. 2. Landscape ccology-Research-Methodology. 3. Biotic communities-Research-Statistical methods. 4. Landscape ecology-Research-Statistical methods. 5. Biotic communities-Research-Data processing.
 Landscape ecology-Research-Data processing. I. Jongman, R. H. G. II. Braak, C. J. F. ter. III. Van Tongeren, O. F. R. QH541,2.D365 1995 574.5'0285-dc20 94-31639 CIP

This reprint is authorized by the original publisher and copyright holder, Centre for Agricultural Publishing and Documentation (Pudoc), Wageningen, 1987.

ISBN 0 521 47574 0 paperback

Contents

	List of Contributors	page xii
	Preface to first edition	yuge kii xiii
	Acknowledgement	xv
	List of symbols	xvi
	Dune Meadow Data	xix
1	Introduction	
1.1	Types of ecological research	1
1.2	Steps in research in community and landscare coolers	1
1.3	Implementing ecological research	2
1.4	Terminology	4
1.5	Historical notes	/
		ð
2	Data collection	10
2.1	Data collection, not an isolated or undirected activity	10
2.2	Objectives and data collection, start by defining the finish	ĨĨ
2.3	Planning of data collection	12
2.3.1	Study design	12
2.3.2	Study type	14
2.3.3	Sampling strategy	15
2.4	Analysis of data and data collection	16
2.4.1	Introduction	16
2.4.2	Measurement scales	16
2.4.3	Frequency distributions	18
2.4.4	Transformations	20
2.5	Interpretation of results	22
2.6	Sources of misinterpretation	23
2.6.1	Incorrect use of statistical analysis	23
2.6.2	Taking detection to be proof	24
2.6.3	Misinterpretation caused by biased sampling	24
2.0.4	Other sources of misinterpretation	25
2.1	Complexity of ecological research	25
2.8	Bibliographical notes	26
2.9	Exercise	27
2.10	Solution to the exercise	27

vii

2.8	Bibliographical notes	26	,	5
2.9	Exercise	27		5.
2.10	Solution to the exercise	27		5.
				5.
3	Regression	29		5.
3.1	Introduction	29		5.3
3.1.1	Aim and use	29		
3.1.2	Response model and types of response variables	30		5.1
3.1.3	Types of explanatory variables and types of response curves	31		5
31.4	Outline of the chapter	33		5
3.2	Regression for quantitative abundance data: least-squares regression	33		01.
321	Nominal explanatory variables: analysis of variance	33		5 '
3.2.1	Stroight lines	27		5
272	Derabalas and Gaussian aurura	37 40		5.
2.4.2	Parabolas and Gaussian curves	40		د سر
3.3	Regression for presence-absence data: logit regression	43		c
3.3.1	Nominal explanatory variables: cni-square test	43		Э С
3.3.2	Sigmoid curves	44		5
3.3.3	Gaussian logit curves	47		5.
3.4	Regression for abundance data with many zero values	49		5.:
3.5	Multiple regression	51		5.
3.5.1	Introduction	51		5.:
3.5.2	Multiple least-squares regression: planes and other surfaces	51		5.:
3.5.3	Multiple logit regression: logit planes and Gaussian logit surfaces	53		5.4
3.5.4	Interaction between explanatory variables	56		5.:
3.5.5	Nominal explanatory variables	58		5.
3.6	Model choice and regression diagnostics	60		5.:
3.7	The method of weighted averaging	61		5.:
3.8	Bibliographic notes	63		5.:
3.9	Standard errors of estimated optimum and tolerance:			5
5.7	confidence interval for the optimum	65		5
3 10	Evercices	66		5
2 1 1	Solutions to everyices	72		5.
5.11	501010113 10 020101865			5.
۰ I	Calibration	70		J.
4 1		70 78		5
4.1	Introduction	70		5.
4.2	Maximum likelihood calibration using response functions	79).
4.2.1	Introduction	79).
4.2.2	Predicting a nominal environmental variable	01		0.5
4.2.3	Predicting a quantitative environmental variable	01		5.5
4.3	Weighted averaging using indicator values	83		5.5
4.4	Inverse regression	80		5.9
4.5	Bibliographic notes	8/		5.
4.6	Exercises	87		5.
4.7	Solutions to exercises	89		
			i.	6
				6.
			7	

	26	5	Órdination	91
	27	5.1	Introduction	91
	27	5.1.1	Aim and usage	91
		5.1.2	Data approximation and response models in ordination	93
	29	5.1.3	Outline of Chapter 5	95
	29	5.2	Correspondence analysis (CA) and detrended correspondence	
	29		analysis (DCA)	95
	30	5.2.1	From weighted averaging to correspondence analysis	95
se curves	31	5.2.2	Two-way weighted averaging algorithm	97
	33	5.2.3	Diagonal structures: properties and faults of correspondence	
quares regression	33		analysis	103
e	33	5.2.4	Detrended correspondence analysis (DCA)	105
	37	5.2.5	Joint plot of species and sites	108
	40	5.2.6	Block structures and sensitivity to rare species	109
ion	43	527	Gaussian ordination and its relation with CA and DCA	110
	43	53	Principal components analysis (PCA)	116
	44	531	From least-squares regression to principal components analysis	116
	47	532	Two-way weighted summation algorithm	110
1169	49	522	Best lines and planes in m-dimensional space	125
400	51	521	Biplot of species and site scores	123
	51	5.3.4	Deto transformation	127 .
metanac	51	525	Data transformation	130
Sullaces	52	5.5.0	K-mode and Q-mode algorithms	131
i logit suffaces	56	5.4	Contribution of ordination with external data	132
	50	5.5	Canonical ordination	130
ş	58	5.5.1	Introduction	136
	00	5.5.2	Canonical correspondence analysis (CCA)	137
	61	5.5.3	Redundancy analysis (RDA)	144
	63	5.5.4	Canonical correlation analysis (COR)	147
ce;		5.5.5	Canonical variate analysis (CVA)	148
	65	5.5.6	Interpreting canonical axes	- 150
	66	5.5.7	Data transformation .	151
	72	5.6	Multidimensional scaling	151
		5.7	Evaluation of direct gradient and indirect gradient analysis	
	78		techniques	153
	78	5.8	Bibliographic notes	156
nctions	79	5.9	Ordination methods in terms of matrix algebra	158
	79	5.9.1	Principal components analysis (PCA)	İ58
	79	5.9.2	Correspondence analysis (CA)	160
	81	5.9.3	Canonical correlation analysis (COR)	161
	83	5.9.4	Redundancy analysis (RDA)	163
	86	5.9.5	Canonical correspondence analysis (CCA)	163
	87	5.10	Exercises	165
	87	5.11	Solutions to exercises	169
	89			
		6	Cluster analysis	174
	'	6 1	Introduction	17/
		0.1		1/+
				•

611	Aims and use	174		741	Introv
612	Types of cluster analysis	175	د	742	Weigh
6.2	A galomerative methods	175		7.4.2	Ontin
621	Introduction	175		7.5	Optin
622	Similarity and dissimilarity	175		7.5	Dicou
623	Properties of the indices	120		7.0	Biblio
621	Transformation standardization and weighting	184	<u>.</u>	78.	Evero
625	A gelomorative eluster electithms	104		7.0	Soluti
6.2.5	Divisive methods	101		1.2	Joint
631	Introduction	101		8	Nume
632	Association analysis and related methods	103		Q 1	Spatie
633	Two Way INdicator Species A Nalveis	103		0.1	the U
6.4	Non-hierarchical clustering	107			an an-
6.5	Ontimality of a clustering	198		811	Data
6.6	Presentation of the results	100		817	Devia
67	Relation between community types and environment	100		8 1 3	Concl
671	The use of simple descriptive statistics	201		8.1.5	Proce
672	The use of tests of significance	201		0.4	south
6.7.2	Pibliography	202		821	Introd
0.0 4 0	Exercises	203		0.2.1 8 2 2	Drana
6.9	Exclusions to everyings	204		8.2.2	Opers
0.10	Solutions to exercises	200		0.2.3 8 7 A	Drolin
7	Spatial aspects of ecological data	213		825	Concl
/ ~71	Introduction	213		83	Peeto
7.1	Trend surface analysis	215		0.0	the N
7.2	Intend-surface analysis	\$ 215		831	Intro
7.2.1	One dimensional trends	215		832	Ecolo
702	Extensions to non normally distributed variables	215		833	Data
7.2.3	Extensions to non-normally distributed variables Non-linear dependence of Z and Y	216		834	Resul
725	Two-dimensional surfaces	210		835	Annli
726	Trend-surfaces as internalators	218		0.2.2	11061
7.2.0	Use of trend surfaces	210		Refer	ences
7.2.1	Local trends	212			011003
73	Snatial autocovariation	222		Index	
7.3	Introduction	222		macx	
7.3.1	Stationarity	224			
7.3.2	Stationality Statistical moments of spatial series	224			
7.5.5	Autonovariagrams and semivariagrams	224			
7.2.4	Frample of the use of outgeorrelation analysis	220			
7.3.3	Example of the use of autocorrelation analysis	227			
7.3.0	The combinerie memory its properties and used	220			
7.3.1	Ine semivariogram, its properties and uses	230			
7.3.0	The comination of a structural tool	225			
1.3.9	An evenue of using active increases to evelve another	ل دیک			
7.3.10	An example of using semivariograms to analyse spatial	235			
71	variation in son	237			
/.4	Spanar interpolation	ا ب سد			

174	, 7.4.1	Introduction	237
175	7.4.2	Weighted moving averages	237
175	7.4.3	Optimum interpolation methods using spatial autocovariance	239
175	7.5	Optimum sampling	242
176	7.6	Discussion	245
180	7.7	Bibliography and historical overview of interpolation	245
184	7.8	Exercises	246
186	7.9	Solutions to exercises	248 -
191		· · · · · · · · · · · · · · · · · · ·	210
191	8	Numerical methods in practice: case-studies	252
193	8.1	Spatial variation in vegetation, soil and precipitation across	
193	•••	the Hawkesbury Sandstone Plateau in New South Wales. Australia	L
197		an analysis using ordination and trend surfaces	252
198	8.1.1	Data collection and analysis	252
199	8.1.2	Deviations from the trends	256
199	8.1.3	Conclusions	257
201	8.2	Processing of vegetation relevés to obtain a classification of	
202		south-west European heathland vegetation	258
203	8.2.1	Introduction	258
204	8.2.2	Preparation of the set of data	258
206	8.2.3	Operations applied to the set of data	259
	8.2.4	Preliminary results	263
213	8.2.5	Concluding remarks	265
213	8.3	Restoration and maintenance of artificial brooks in the Veluwe.	
. 215		the Netherlands	266
215	8.3.1	Introduction	266
215	83.2	Ecological inventory	268
216	833	Data analysis	268
216	834	Results	269
210	835	Application of the results	272
218	0.0.0		
219	Refer	ences	275
222		******	
222	Index		292
222	Index		
224			
224			
224			
220			
227			
220			
250			
224			

- 12

List of sym	ibols	, n	maximum probability of accurrence
		P max	Runding probability of occurrence.
		, P	P value of a statistical test, e.g. $P < 0.05$.
		q	number of explanatory variables (often equal to number of envir- onmental variables).
		r	coefficient of correlation (in a sample).
ANOVA	analysis of variance.	R	multiple correlation coefficient in regression; R^2 is termed the coefficient of determination.
b _k	value of the regression coefficient for the k-th species; used when only one regression coefficient is involved.	$R_{\rm adj}^2$	adjusted R^2 , also termed percentage variance accounted for. A recommended modification of R^2 to adjust for the number of
b_0, b_1, b_2, \dots	coefficients (parameters) in a regression equation; b_0 is usually the intercent	•	parameters littled by regression. For large sample sizes R^2_{adj} is approximately equal to R^2 .
c, <i>c</i> _k	maximum of a response curve. Can be indexed by a species number	S	standard deviation of a sample, or residual standard deviation in regression.
	(<i>k</i>).	5 ²	variance of a sample, or residual variance in regression.
c_0, c_1, c_2, \dots	coefficients (parameters) in a regression equation.	s.e.	standard error.
C(h)	covariance function, i.e. function giving the covariance between $Z(x_1)$ and $Z(x_2)$, where x_1 and x_2 are two points.	s.d.	standard deviation of a sample. In Chapter 5, standard deviation of a unimodal response curve.
d.f.	degrees of freedom.	S.S.	sum of squares in an ANOVA table.
$\exp(x), e^x$	the exponential function or antilog of $x (exp(log(x))=x)$.	1.1.	tolerance, a measure of ecological amplitude, is the parameter for
Ey or $E(y)$	expected value of a random variable y.	- , - K	curve width in the Gaussian logit response model. It can be indexed
F	variance ratio in an ANOVA table.		by a species number (k) .
h	lag, distance.	<i>t</i> , <i>t</i> (v)	t or Student's distribution with v degrees of freedom: shows both the random variable and a particular or observed value $t_{i}(v)$ or
i, j	indices numbering sites in the data $(i = 1, 2,, n; j = 1, 2,, n)$, often used as subscript, e.g. x_i , the value of variable x		t_{α} , is the critical value of a <i>t</i> distribution in a two-sided (two-tailed) statistical test with significance level α .
	in the <i>i</i> -th site.	<i>u</i> , <i>u</i> _k	optimum of a response curve, i.e. the value for which the response
k, l	indices numbering the species in the data $(k = 1, 2,, m; l = 1, 2,, m)$, often used as subscript, e.g. y_k , the value of variable		curve under consideration attains its maximum (when uniquely defined). Can be indexed by a species number (k) .
	y of the k-th species.	var(y)	variance of a random variable y, also denoted by σ^2 or V.
$\log_{e}(x)$	Naperian or natural logarithm of x (for $x > 0$).	vr	variance ratio in an ANOVA table.
m	number of response variables (often equal to number of species).	x	arithmetic mean of the variable x in a sample.
m.s.	mean square in an ANOVA table.	<i>x</i> 0	value of the variable x at a particular site.
n	number of sites (statistical sampling units, objects, etc.).	x_i	value of the variable x at the <i>i</i> -th site.
p	probability of occurrence of a species.	x_1, x_2, \ldots	explanatory variables in a regression equation, often observed
<i>p</i> (<i>x</i>)	probability of occurrence of a species as a function of the variable x .		environmental variables. Also used for latent variables in ordination (i.e. theoretical, environmental variables) or variables indicating spatial position.
xvi			xvii

ALL DISC. O

y _i	value of a particular response variable y in the <i>i</i> -th site, used when it is clear from the context which variable is being considered.
<i>y</i> _{<i>k</i>}	value from the k -th response variable (species) in a particular site, used when it is clear from the context which site is considered.
y _{ki}	value of the k-th response variable (species) in the i-th site.
<i>y_{+i}, y_{k+}</i>	sum of y_{ki} over the index $k = 1,, m$; sum of y_{ki} over the index $i = 1,, n$.
z_j	<i>j</i> -th environmental variable $(j = 1, 2, \ldots, q)$.
Z _{ji}	value of <i>j</i> -th environmental variable in the <i>i</i> -th site.
Z(x)	a spatial random variable, with x denoting the spatial position. z(x) is the value of $Z(x)$ observed in x
γ(<i>h</i>)	semivariance. $2\gamma(h) = var[Z(x_1) - Z(x_2)]$, where x_1 and x_2 are points at distance h apart.
δ _{ij}	dissimilarity between sites <i>i</i> and <i>j</i> .
ε	error term in a regression equation. ε is the random variable y -E(y).
ê	estimator of the parameter θ .
λ, λ _s	eigenvalue of a cross-product matrix; λ_s is the eigenvalue of the <i>s</i> -th axis or eigenvector.
$\mu(x)$	expected value of a random variable.
ν	degrees of freedom.
$\rho(h)$	correlation function $\rho(h) = C(h)/C(0)$.
σ	standard deviation of a random variable.
σ^2	variance of a random variable; often used for the error variance in regression.

- $\sum_{i=1}^{n} y_i \qquad y_1 + y_2 + \ldots + y_n$
- χ^2 (v) Chi-square distribution with v degrees of freedom: shows both the random variable and a particular or observed value. $\chi^2_{\alpha}(v)$ is the critical value of a chi-square distribution in a statistical test with significance level α .

Dune Meadow Data

In this book the same set of vegetation data will be used in the chapters on ordination and cluster analysis. This set of data stems from a research project on the Dutch island of Terschelling (Batterink & Wijffels 1983). The objective of this project was to detect a possible relation between vegetation and management in dune meadows. Sampling was done in 1982. Data collection was done by the Braun-Blanquet method; the data are recorded according to the ordinal scale of van der Maarel (1979b). In each parcel usually one site was selected; only in cases of great variability within the parcel were more sites used to describe the parcel. The sites were selected by throwing an object into a parcel. The point where the object landed was fixed as one corner of the site. The sites measure $2x2 m^2$. The sites were considered to be representative of the whole parcel. From the total of 80 sites, 20 have been selected to be used in this book (Table 0.1). This selection expresses the variation in the complete set of data. The names of the species conform with the nomenclature in van der Meijden et al. (1983) and Tutin et al. (1964-1980).

Data on the environment and land-use that were sampled in this project are (Table 0.2):

- thickness of the A1 horizon
- moisture content of the soil
- grassland management type
- agricultural grassland use
- quantity of manure applied.

The thickness of the Al horizon was measured in centimetres and it can therefore be handled as a quantitative variable. In the dunes, shifting sand is a normal phenomenon. Frequently, young developed soils are dusted over by sand, so that soil development restarts. This may result in soils with several Al horizons on top of each other. Where this had occurred only the Al horizon of the top soil layer was measured.

The moisture content of the soil was divided into five ordered classes. It is therefore an ordinal variable.

Four types of grassland management have been distinguished:

- standard farming (SF)
- biological farming (BF)
- hobby-farming (HF)
- nature conservation management (NM).

The grasslands can be used in three ways: as hayfields, as pasture or a combination of these (intermediate). Both variables are nominal but sometimes the use of the

grassland is handled as an ordinal variable (Subsection 2.3.1). Therefore a ranking order has been made from hay production (1), through intermediate (2) to grazing (3).

The amount of manuring is expressed in five classes (0-4). It is therefore an ordinal variable.

All ordinal variables are treated as if they are quantitative, which means that the scores of the manure classes, for example, are handled in the same way as the scores of the A1 horizon. The numerical scores of the ordinal variables are given in Table 0.2. There are two values missing in Table 0.2. Some computer programs cannot handle missing values, so the mean value of the corresponding variable has been inserted. The two data values are indicated by an asterisk.

хx

Table 0.1. Dune Meadow Data. Unordered table that contains 20 relevées (columns) and 30 species (rows). The right-hand column gives the abbreviation of the species names listed in the left-hand column; these abbreviations will be used throughout the book in other tables and figures. The species scores are according to the scale of van der Maarel (1979b).

00000000011111111112 12345678901234567890

123456789111294567890123456780	Achillea millefolium Agrostis stolonifera Aira praecox Alopecurus geniculatus Anthoxanthum odoratum Bellis perennis Bromus hordaceus Chenopodium album Cirsium arvense Eleocharis palustris Elymus repens Empetrum nigrum Hypochaeris radicata Juncus articulatus Juncus bufonius Leontodon autumnalis Lolium perenne Plantago lanceolata Poa pratensis Poa trivialis Potentilla palustris Ranunculus flammula Rumex acetosa Sagina procumbens Salix repens Trifolium pratense Trifolium repens	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Achr Alot Leity pur all a control and the second control and the sec	mitoano phola pregodeno progodeno progodeno pregodeno prego
28	Unindia (mala and data		(11)	rep
20	vicia (athyroides		Vic	iat
29	Brachythecium rutabulum	. 2226222244. 44.634	Bra	rut
30	Callieroopella cuspidata	2 2 2	C-1	
	router, 20.0000 cosht09(9		Lar	CUS

Table 0.2. Environmental data (columns) of 20 relevées (rows) from the dune meadows. The scores are explained in the description of the Dune Meadow research project above; asterisk denotes mean value of variable.

Sample number	Al horizon	Moisture class	Management type	Use	Manure class
I	2.8	1	SF	2	4
2	3.5	1	BF	2	2
3	4.3	2	SF	2	4
4	4.2	2	SF	2	4
5	6.3	I	HF	1	2
6	4.3	I	HF	2	2
7	2.8	I	HF	3	3
8	4.2	5	HF	3	3
9	3.7	4	HF	1	1
10	3.3	2	BF	1	1
II	3.5	1	BF	3	1
12	5.8	4	SF	2	2*
13	6.0	5	SF	2	3
14	9.3	5	NM	3	0
15	11.5	5	NM	2	0
16	5.7	5	SF	3	3
17	4.0	2	NM	1	0
18	4.6*	I	NM	I	0
19	3.7	5	NM	1	0
20	3.5	5	NM	1	0

5 Ordination

C.J.F. ter Braak

5.1 Introduction

5.1.1 Aim and usage

Ordination is the collective term for multivariate techniques that arrange sites along axes on the basis of data on species composition. The term ordination was introduced by Goodall (1954) and, in this sense, stems from the German 'Ordnung', which was used by Ramensky (1930) to describe this approach.

The result of ordination in two dimensions (two axes) is a diagram in which sites are represented by points in two-dimensional space. The aim of ordination is to arrange the points such that points that are close together correspond to sites that are similar in species composition, and points that are far apart correspond to sites that are dissimilar in species composition. The diagram is a graphical summary of data, as in Figure 5.1, which shows three groups of similar sites. Ordination includes what psychologists and statisticians refer to as multidimensional scaling, component analysis, factor analysis and latent-structure analysis.

Figure 5.1 also shows how ordination is used in ecological research. Ecosystems are complex: they consist of many interacting biotic and abiotic components. The way in which abiotic environmental variables influence biotic composition is often explored in the following way. First, one samples a set of sites and records which species occur there and in what quantity (abundance). Since the number of species is usually large, one then uses ordination to summarize and arrange the data in an ordination diagram, which is then interpreted in the light of whatever is known about the environment at the sites. If explicit environmental data are lacking, this interpretation is done in an informal way; if environmental data have been collected, in a formal way (Figure 5.1). This two-step approach is indirect gradient analysis in the sense used by Whittaker (1967). By contrast, direct gradient analysis, one is interested from the beginning in particular environmental variables, i.e. either in their influence on the species as in regression analysis (Chapter 3) or in their values at particular sites as in calibration (Chapter 4).

Indirect gradient analysis has the following advantages over direct gradient analysis. Firstly, species compositions are easy to determine, because species are usually clearly distinguishable entities. By contrast, environmental conditions are difficult to characterize exhaustively. There are many environmental variables and even more ways of measuring them, and one is often uncertain of which variables the species react to. Species composition may therefore be a more informative



Figure 5.1 Outline of the role of ordination in community ecology, showing the typical format of data sets obtained by sampling ecosystems and their analysis by direct gradient and indirect gradient analysis. Also shown is the notation used in Chapter 5. Point of site in the ordination diagram (\bullet) .

indicator of environment than any given set of measured environmental variables. Ordination can help to show whether important environmental variables have been overlooked: an important variable has definitely been missed if their is no relation between the mutual positions of the sites in the ordination diagram and the measured environmental variables.

Secondly, the actual occurrence of any individual species may be too unpredictable to discover the relation of its occurrence to environmental conditions by direct means (Chapter 3) and therefore more general patterns of coincidence of several species are of greater use in detecting species-environment relations.

Thirdly, for example in landscape planning, interest may from the onset be focused more on the question of which combinations of species can occur, and less on the behaviour of particular species. Regression analysis of single species then provides too detailed an account of the relations between species and environment. The ordination approach is less elaborate and gives a global picture, but - one hopes - with sufficient detail for the purpose in hand.

Between regression analysis and ordination (in the strict sense) stand the canonical ordination techniques. They are ordination techniques converted into multivariate direct gradient analysis techniques; they deal simultaneously with many species and many environmental variables. The aim of canonical ordination is to detect the main pattern in the relations between the species and the observed environment.

5.1.2 Data approximation and response models in ordination

Ordination techniques can be viewed in two ways (Prentice 1977). According to one view, the aim of ordination is to summarize multivariate data in a convenient way in scatter diagrams. Ordination is then considered as a technique for matrix approximation (as the data are usually presented in the two-way layout of a matrix). A second, more ambitious, view assumes from the beginning that there is an underlying (or latent) structure in the data, i.e. that the occurrences of all species under consideration are determined by a few unknown environmental variables (latent variables) according to a simple response model (Chapter 3). Ordination in this view aims to recover that underlying structure. This is illustrated in Figure 5.2 for a single latent variable. In Figure 5.2a, the relations of two species, A and B, with the latent variable are rectilinear. In Figure 5.2c they are unimodal. We now record species abundance values at several sites and plot the abundance of Species A against that of Species B. If relations with the latent variable were rectilinear, we would obtain a straight line in the plot of Species B against Species A (Figure 5.2b), but if relations were unimodal, we would obtain a complicated curve (Figure 5.2d). The ordination problem of indirect gradient analysis is to infer about the relations with the latent variable (Figures 5.2a,c) from the species data only (Figure 5.2b,d). From the second viewpoint, ordination is like regression analysis, but with the major difference that in ordination the explanatory variables are not known environmental variables, but 'theoretical' variables. These variables, the latent variables, are constructed in such a way that they best explain the species data. As in regression, each species thus constitutes a response variable, but in ordination these response variables are analysed simultaneously. (The distinction between these two views of ordination is not clear-cut, however. Matrix approximation implicitly assumes some structure in the data by the mere way the data are approximated. If the data structure is quite different from the assumed structure, the approximation is inefficient and fails.)

The ordination techniques that are most popular with community ecologists, are principal components analysis (PCA), correspondence analysis (CA), and techniques related to CA, such as weighted averaging and detrended correspondence analysis. Our introduction to PCA and CA will make clear that PCA and CA are suitable to detect different types of underlying data structure. PCA relates to a linear response model in which the abundance of any species either increases or decreases with the value of each of the latent environmental variables (Figure 5.2a). By contrast, CA is related, though in a less unequivocal way, to a unimodal response model (Figure 5.2c). In this model, any species occurs in a limited range



Figure 5.2 Response curves for two species A and B against a latent variable x (a, c) and the expected abundances of the species plotted against each other (b, d), for the straight line model (a, b) and a unimodal model (c, d). The numbers refer to sites with a particular value for x. The ordination problem is to make inferences about the relations in Figures a and c from species data plotted in Figures b and d.

of values of each of the latent variables. PCA and CA both provide simultaneously an ordination for the sites and an ordination for the species. The two ordinations may be plotted in the same diagram to yield 'joint plots' of site and species points, but the interpretation of the species points is different between PCA and CA.

PCA and CA operate directly on the species data. By contrast, multidimensional scaling is a class of ordination techniques that operate on a table of dissimilarity values between sites. To apply these techniques, we must therefore first choose an appropriate dissimilarity coefficient to express the dissimilarity in species composition between any two sites (Subsection 6.2.2). After choosing one, we can calculate the dissimilarity values of all pairs of sites required as input for multidimensional scaling. CA and PCA may also be considered as multidimensional scaling techniques, but ones that use a particular dissimilarity coefficient.

5.1.3 Outline of Chapter 5

Section 5.2 introduces CA and related techniques and Section 5.3 PCA. Section 5.4 discusses methods of interpreting ordination diagrams with external (environmental) data. It is also a preparation for canonical ordination (Section 5.5). After a discussion of multidimensional scaling (Section 5.6), Section 5.7 evaluates the advantages and disadvantages of the various ordination techniques and compares them with regression analysis and calibration. After the bibliographic notes (Section 5.8) comes an appendix (Section 5.9) that summarizes the ordination methods described in terms of matrix algebra.

5.2 Correspondence analysis (CA) and detrended correspondence analysis (DCA)

5.2.1 From weighted averaging to correspondence analysis

Correspondence analysis (CA) is an extension of the method of weighted averaging used in the direct gradient analysis of Whittaker (1967) (Section 3.7). Here we describe the principles in words; the mathematical equations will be given in Subsection 5.2.2.

Whittaker, among others, observed that species commonly show bell-shaped response curves with respect to environmental gradients. For example, a plant species may prefer a particular soil moisture content, and not grow at all in places where the soil is either too dry or too wet. In the artificial example shown in Figure 5.3a, Species A prefers drier conditions than Species E, and the Species B, C and D are intermediate. Each of the species is therefore largely confined to a specific interval of moisture values. Figure 5.3a also shows presence-absence data for Species D: the species is present at four of the sites.

We now develop a measure of how well moisture explains the species data. From the data, we can obtain a first indication of where a species occurs along the moisture gradient by taking the average of the moisture values of the sites in which the species is present. This average is an estimate of the optimum of the species (the value most preferred), though not an ideal one (Section 3.7). The average is here called the species score. The arrows in Figure 5.3a point to the species scores so calculated for the five species. As a measure of how well moisture explains the species data, we use the dispersion ('spread') of the species scores. If the dispersion is large, moisture neatly separates the species curves and moisture explains the species data well. If the dispersion is small, then moisture explains less. To compare the explanatory power of different environmental variables, each environmental variable must first be standardized; for example by subtracting its mean and dividing by its standard deviation.

Suppose that moisture is the 'best' single environmental variable measured in the artificial example. We might now wish to know whether we could in theory have measured a variable that explains the data still better. CA is now the technique that constructs the theoretical variable that best explains the species data. CA does so by choosing the best values for the sites, i.e. values that maximize the dispersion of the species scores (Figure 5.3b). The variable shown gives a larger



Figure 5.3 Artificial example of unimodal response curves of five species (A-E) with respect to standardized variables, showing different degrees of separation of the species curves. a: Moisture. b: First axis of CA. c: First axis of CA folded in this middle and the response curves of the species lowered by a factor of about 2. Sites are shown as dots at y = 1 if Species D is present and at y = 0 if Species D is absent. For further explanation, see Subsections 5.2.1 and 5.2.3.

dispersion than moisture; and consequently the curves in Figure 5.3b are narrower, and the presences of Species D are closer together than in Figure 5.3a.

The theoretical variable constructed by CA is termed the first ordination axis of CA or, briefly, the first CA axis; its values are the site scores on the first CA axis.

A second and further CA axes can also be constructed; they also maximize the dispersion of the species scores but subject to the constraint of being uncorrelated with previous CA axes. The constraint is intended to ensure that new information is expressed on the later axes. In practice, we want only a few axes in the hope that they represent most of the variation in the species data.

So we do not need environmental data to apply CA. CA 'extracts' the ordination

axes from the species data alone. CA can be applied not only to presence-absence data, but also to abundance data; for the species scores, we then simply take a weighted average of the values of the sites (Equation 3.28).

5.2.2 Two-way weighted averaging algorithm

Hill (1973) introduced CA into ecology by the algorithm of reciprocal averaging. This algorithm shows once more that CA is an extension of the method of weighted averaging.

If we have measured an environmental variable and recorded the species composition, we can estimate for each species its optimum or indicator value by averaging the values of the environmental variable over the sites in which the species occurs, and can use the averages so obtained to rearrange the species (Table 3.9). If the species show bell-shaped curves against the environmental variable, the rearranged table will have a diagonal structure, at least if the optima of the curves differ between the species (Table 3.9). Conversely, if the indicator values of species are known, the environmental variable at a site can be estimated from the species that it contains, by averaging the indicator values of these species (Section 4.3) and sites can be arranged in order of these averages. But, these methods are only helpful in showing a clear structure in the data if we know in advance which environmental variable determines the occurrences of the species. If this is not known in advance, the idea of Hill (1973) was to discover the 'underlying environmental gradient' by applying this averaging process both ways in an iterative fashion, starting from arbitrary initial values for sites or from arbitrary initial (indicator) values for species. It can be shown mathematically that this iteration process eventually converges to a set of values for sites and species that do not depend on the initial values. These values are the site and species scores of the first CA axis.

We illustrate now the process of reciprocal averaging. For abundance data, it is rather a process of two-way weighted averaging. Table 5.1a shows the Dune Meadow Data (Table 0.1), arranged in arbitrary order. We take as initial values for the sites the numbers 1 to 20, as printed vertically below Table 5.1a. As before, we shall use the word 'score', instead of 'value'. From the site scores, we derive species scores by calculating the weighted average of the site scores for each species. If we denote the abundance of species k at site i by y_{ki} , the score of site i by x_i and the score of species k by u_k , then the score of species k becomes the weighted average of site scores (Section 3.7)

 $u_k = \sum_{i=1}^n y_{ki} x_i / \sum_{i=1}^n y_{ki}$

Equation 5.1

For Achillea millefolium in Table 5.1a, we obtain $u_1 = (1 \times 1 + 3 \times 2 + 2 \times 5 + 2 \times 6 + 2 \times 7 + 4 \times 10 + 2 \times 17)/(1 + 3 + 2 + 2 + 2 + 4 + 2) = 117/16 = 7.31$. The species scores thus obtained are also shown in Table 5.1a. From these species scores, we derive new site scores by calculating for each site the weighted average of the species scores, i.e.

Tab	le 5.1a		
Spe k	cies	Sites (i) 00000000011111111112 12345678901234567890	<i>u</i> _k
12345678911123456789012234567890	Ach mil Agr sto Air pra Alto gen Anto gen Bet per Bro hor Che alto Ely rep Emp nid Jun art Leo per Pla lan Poa pra Ram fla Ram fla Ram pro Sal rep Tri pep Vic lat Bra rut Cal cus	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 7.31\\ 11.33\\ 18.20\\ 9.03\\ 11.24\\ 6.62\\ 5.60\\ 13.00\\ 14.84\\ 4.38\\ 13.33\\ 10.54\\ 16.78\\ 13.33\\ 10.54\\ 16.78\\ 13.33\\ 10.54\\ 15.14\\ 50.31\\ 15.14\\ 50.35\\ 18.10\\ 9.47\\ 12.50\\ 12.02\\ 16.40\\ \end{array}$
x_i		11111111112 12345678901234567890	

86

Table 5.1 Two-way weighted averaging algorithm of CA applied to the Dune Meadow Data presented in a preliminary section of this book. The site numbers and site scores are printed vertically. a: Original data table with at the bottom the initial site scores. b: Species and sites rearranged in order of their scores obtained after one cycle of two-way weighted averaging. c: Species and sites arranged in order of their final scores (CA scores). Note the minus signs in the site scores; for example, the score of Site 17 is -1.46.

Species Si B Cir arv 00 11 Ely rep 44 7 Bro hor 4 26 Tri pra 4 17 Lot per 75 6 Bel per 3 23 Rum ace 3 19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 2 27 Tri rep 5 24 Sag pro 15 Jun buf	tes (i) 200001001110111112 2534706931288764550 2 4444 = 6 42 = 324 2 = 5 = 5 = 5 = 5 = 5 = 5 = 5 = 5 = 5 =	u _k 4.00 4.38 5.60 6.00 6.31 6.62 6.83 7.25 7.25 7.25 7.31 9.03 9.27 9.47	Species k 3 Air pra 5 Ant odo 1 Ach mil 26 Tri pra 13 Hyp rad 18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	Sites (i) 10100011010001101121 75076191283492385406 2 3 44423 4 224221 3 2 25 2 52 25355 $3 3$ 2 242 4 3 5 36 22 1 $2 1$ 22 26667 752652 4 124434 443544 24	<i>u_k</i> -0.99 -0.91 -0.86 -0.84 -0.87 -0.65 -0.65 -0.65 -0.50 -0.50
8 00 11 Ely rep 44 7 Bro hor 4 28 Tri pra 17 17 Lot per 75 6 Bet per 5 23 Rum ace 13 23 Rum ace 13 19 Poa pra 44 20 Poa tri 27 1 Ach mit 13 4 Alo gen 2 18 Pta tan 2 27 Tri rep 5 24 Sag pro 15	2000001001110111112 2534706931288764550 2 24444 6 42 324 2 2 5 52656862 7 42 3222 2 2 5 3 62 2 2534443424 431 266554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	<i>u_k</i> 4.00 4.38 5.60 6.00 6.31 6.62 6.83 7.25 7.25 7.25 7.31 9.03 9.27 9.47	k 3 Air pra 5 Ant odo 1 Ach mil 26 Tri pra 13 Hyp rad 18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	<i>u_k</i> -0.99 -0.91 -0.84 -0.84 -0.84 -0.65 -0.65 -0.65 -0.65 -0.50 -0.50
K 12 S Cir arv 11 Ely rep 44 7 Bro hor 4 28 Tri pra 17 Lol per 75 6 Bel per 75 6 Bel per 75 23 Rum ace 1 19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 2 27 Tri rep 5 24 Sag pro 1 5 Jun buf	2 4444 6 42 324 2 2 2 5 526566652 7 42 3222 2 2 5 3 62 2 2254443424 431 266554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	"k 4.00 4.38 5.60 6.00 6.31 6.62 6.89 7.25 7.25 7.25 7.31 9.03 9.27 9.47	3 Air pra S Ant odo 1 Ach mil 26 Tri pra 13 Hyp rad 18 Pla lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic lat 6 Bel per 17 Lol per 19 Poa pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	"% -0.99 -0.91 -0.88 -0.84 -0.87 -0.69 -0.69 -0.69 -0.69 -0.50 -0.50
9 Cir arv 11 Ely rep 44 7 Bro hor 4 28 Tri pra 17 Lol per 75 6 Bel per 3 23 Rum ace 18 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 23 18 Pla lan 27 Tri rep 5 24 Sag pro	2 4444 6 42 324 2 2 5 52656862 7 42 3222 2 2 5 3 62 2 2254443424 431 766554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	4.00 4.38 5.60 6.00 6.31 6.62 6.89 7.25 7.25 7.31 9.03 9.27 9.47	3 Air pra 5 Ant odo 1 Ach mil 26 Tri pra 13 Hyp rad 18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-0.99 -0.91 -0.86 -0.84 -0.84 -0.87 -0.69 -0.69 -0.69 -0.69 -0.69 -0.50
11 Ely rep 44 7 Bro hor 4 28 Tri pra 4 17 Lot per 75 6 Bet per 3 23 Rum ace 1 18 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 24 Slag pro 5 24 Sag pro 5 15 Jun buf 5	444 6 12 324 2 2 52656862 7 3222 2 5 362 2254443424 431 766554459 44 2 22 2 242 2 25 325 35 344 32 35 35 35 32 3221265323322 612	4.38 5.60 6.00 6.31 6.62 6.83 7.25 7.25 7.31 9.03 9.27 9.47	S Ant odo 1 Ach mil 26 Tri pra 13 Hyp rad 18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-0.98 -0.91 -0.88 -0.84 -0.87 -0.69 -0.69 -0.69 -0.69 -0.69 -0.50
7 Bro hor 4 26 Tri pra 7 12 Lol per 75 6 Bel per 3 23 Rum ace 3 18 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 5 24 Sag pro 5 15 Jun buf 3	A2 324 2 25 32656862 7 42 3222 2 2 5 3 62 2 4254443424 431 666554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 3221265323322 612	5.60 6.00 6.31 6.62 6.89 7.25 7.25 7.25 7.31 5.03 9.27 9.47	1 Ach mil 26 Tri pra 13 Hyp rad 18 Pla lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic lat 6 Bel per 17 Lol per 19 Poa pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-0.97 -0.88 -0.84 -0.87 -0.87 -0.65 -0.65 -0.65 -0.50 -0.50
26 Tri pra 17 Lol per 75 6 Bel per 3 23 Rum ace 3 19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 5 24 Sag pro 5 15 Jun buf 3	2 2 5 22556852 7 42 3222 2 2 25 3 62 2 4254443424 431 766554459 44 2 22 242 2 72 35 85 4 5 535 3 32 5221265323322 612	6.00 6.31 6.62 6.89 7.25 7.25 7.31 9.03 9.27 9.47	26 Tri pra 13 Hyp rad 18 Pla lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic lat 6 Bel per 17 Lol per 19 Poa pra	2 25 2 52 25355 3 3 2 242 4 3 5 36 22 1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.88 -0.84 -0.84 -0.87 -0.85 -0.65 -0.65 -0.50 -0.50
17 Lol per 75 6 Bel per 3 23 Rum ace 3 19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 3 27 Tri rep 5 24 Sag pro 5 15 Jun buf	52656862 7 42 522 2 2 2 1254443424 431 766554459 44 2 2 72 35 85 4 5 535 3 32 5221265323322 612	6.31 6.62 6.89 7.25 7.25 7.31 9.03 9.27 9.47	13 Hyp rad 18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	2 52 25355 3 3 2 2 242 4 3 5 36 22 1 2 1 22 3222 26867 752652 4 124434 443544 24	-0.84 -0.87 -0.87 -0.68 -0.68 -0.68 -0.68 -0.50 -0.50
6 Bel per 3 23 Rum ace 3 19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 3 27 Tri rep 5 24 Sag pro 5 15 Jun buf	3222 2 2 2 5 3 62 2 2254443424 431 266554453 44 2 32 242 2 2 72 35 85 4 5 535 3 32 3221265323322 612	6.62 6.89 7.25 7.25 7.31 8.03 9.27 9.47	18 PLa Lan 12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	25355 3 3 2 242 4 3 5 36 22 1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.84 -0.65 -0.65 -0.65 -0.62 -0.50 -0.50
23 Rum ace 18 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 27 Tri rep 5 24 Sag pro 15 Jun buf	5 3 62 2 125443424 431 766554459 44 2 32 242 2 72 35 85 4 5 535 3 32 5221265323322 612	6.89 7.25 7.25 7.31 9.03 9.27 9.47	12 Emp nig 7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	2 242 4 3 5 36 22 1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.67 -0.68 -0.65 -0.62 -0.50 -0.50
19 Poa pra 44 20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 2 27 Tri rep 5 24 Sag pro 15 Jun buf	254443424 431 766554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	7.25 7.25 7.31 8.03 9.27 9.47	7 Bro hor 23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	242 4 3 5 36 22 1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.68 -0.69 -0.62 -0.50 -0.50
20 Poa tri 27 1 Ach mil 13 4 Alo gen 2 18 Pla lan 27 Tri rep 5 24 Sag pro 15 Jun buf	766554459 44 2 32 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	7.25 7.31 9.03 9.27 9.47	23 Rum ace 28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	5 36 22 1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.65 -0.62 -0.50 -0.50
1 Ach mil 13 4 Alo gen 2 18 Pla lan 27 Tri rep 5 24 Sag pro 15 Jun buf	92 242 2 2 72 35 85 4 5 535 3 32 5221265323322 612	7.31 9.03 9.27 9.47	28 Vic Lat 6 Bel per 17 Lol per 19 Poa pra	1 2 1 22 3222 26667 752652 4 124434 443544 24	-0.52
4 Alo gen 2 18 Pla lan 27 Tri rep 5 24 Sag pro 15 Jun buf	2 72 35 85 4 5 535 3 32 5221265323322 612	9.03 9.27 9.47	6 Bel per 17 Lol per 19 Poa pra	22 3222 26667 752652 4 124434 443544 24	-0.50
18 Pla Ian 27 Tri rep 5 24 Sag pro 15 Jun buf	5 535 3 32 221265323322 612	9.27 9.47	17 Lol per 19 Poa pra	26667 752652 4 124434 443544 24	-0.50
27 Tri rep 5 24 Sag pro 15 Jun buf	221265323322 612	9.47	19 Poa oca	124434 443544 24	-0 39
24 Sag pro 15 Jun buf					-0.00
15 Jun buf	5 22242 3	10.35	11 Ely rep	4 4 4 4 4 4 6	-0.37
	2 43 4	10.54	16 Leo aut	23333 6555222223222	-0.15
16 Leo aut 5	3223332252352 2262	10.94	20 Poa tri	64542 7 655494 2	-0.18
5 Ant odo	4 243 4 4	11.24	27 Tri rep	2625 235221332216	-0.08
2 Agr sto	48 35 44 744 5	11.33	9 Cir arv	2	-0.OE
29 Bra rut	2222262 4426 4 434	12.02	24 Sag pro	32 52422	0.00
28 Vic Lat	1 2 1	12.50	15 Jun buf	2 443	0.08
8 Che alb	1	13.00	29 Bra rut	2226 34 62224 24 44	0.18
14 Jun art	4 4 3 3 4	13.39	4 Alo gen	2 723855 4	0.40
21 Pot pal	22	14.50	8 Che alb	1	0.42
10 Ele pal	4 845 4	14.84	25 Sat rep	33 5	0.62
22 Ran fla	2 2 222 4	15.14	2 Agr sto	4834544457	0.93
30 Cal cus	343	16.40	14 Jun art	4 43 43	1.28
13 Hyp rad	2 2 5	16.78	22 Ran fla	222242	1.56
3 Hr bra	23	18.20	10 Ele pal	45448	1.77
12 Emp nig	2	19.00	21 Pot pal	22	1.92
25 Sat rep	3 35	19.18	30 Cal cus	433	1.96
	11111111				
r _i 67	7888888899900122234		x_i	10000000000000001112	
•••	8011748877778689997			40000000000000400000	
24	3/00707/3/200303 3/00707/38/019/786			45666666531002475550	

For Site 1 in Table 5.1a, we obtain $x_1 = (1 \times 7.31 + 4 \times 4.38 + 7 \times 6.31 + 4 \times 7.25 + 2 \times 7.25)/(1 + 4 + 7 + 4 + 2) = 112.5/18 = 6.25$. In Table 5.1b, the species and sites are arranged in order of the scores obtained so far. The new site scores are also printed vertically underneath. There is already some diagonal structure, i.e. the occurrences of each species tend to come together along the rows. We can improve upon this structure by calculating new species scores from the site scores that we have just calculated, and so on.

A practical numerical problem with this technique is that, by taking averages, the range of the scores gets smaller and smaller. For example, we started off with a range of 19 (site scores from 1 to 20) and after one cycle the site scores have a range of 14.36 - 6.25 = 8.11 (Table 5.1b). To avoid this, either the site scores or the species scores must be rescaled. Here the site scores have been rescaled. There are several ways of doing so. A simple way is to rescale to a range from 0 to 100 by giving the site with the lowest score the value 0 and the site with the highest score the value 100 and by calculating values for the remaining sites in proportion to their scores; in the example, the rescaled scores would be obtained with the formula $(x_i - 6.25)/0.0811$.

We shall use another way in which the site scores are standardized to (weighted)

Table 5.2 Two-way weighted averaging algorithm of CA.

a: Iteration process

Step 1. Take arbitrary, but unequal, initial site scores (x_i) .

Step 2. Calculate new species scores (u_k) by weighted averaging of the site scores (Equation 5.1).

Step 3. Calculate new site scores (x_i) by weighted averaging of the species scores (Equation 5.2).

Step 4. For the first axis, go to Step 5. For second and higher axes, make the site scores (x_i) uncorrelated with the previous axes by the orthogonalization procedure described below.

Step 5. Standardize the site scores (x_i) . See below for the standardization procedure.

Step 6. Stop on convergence, i.e. when the new site scores are sufficiently close to the site scores of the previous cycle of the iteration; ELSE go to Step 2.

b: Orthogonalization procedure

Step 4.1. Denote the site scores of the previous axis by f_i and the trial scores of the present axis by x_i .

Step 4.2. Calculate $v = \sum_{i=1}^{n} y_{+i} x_i f_i / y_{++}$ where $y_{+i} = \sum_{k=1}^{n} y_{ki}$ and $y_{+i} = \sum_{i=1}^{n} y_{+i}$. Step 4.3 Calculate $x_{i,new} = x_{i,old} - v f_i$. Step 4.4 Repeat Steps 4.1-4.3 for all previous axes.

c: Standardization procedure

Step 5.1 Calculate the centroid, z, of site scores $(x_i) z = \sum_{i=1}^{n} y_{+i} x_i / y_{++}$. Step 5.2 Calculate the dispersion of the site scores $s^2 = \sum_{i=1}^{n} y_{+i} (x_i - z)^2 / y_{++}$. Step 5.3 Calculate $x_{i,\text{new}} = (x_{i,\text{od}} - z) / s$. Note that, upon convergence, s equals the eigenvalue. mean 0 and variance 1 as described in Table 5.2c. If the site scores are so standardized, the dispersion of the species scores can be written as

$$\delta = \sum_{k=1}^m y_{k+} u_k^2 / y_{++}$$

Equation 5.3

where

 y_{k+} is the total abundance of species k y_{++} the overall total.

The dispersion will steadily increase in each iteration cycle until, after about 10 cycles, the dispersion approaches its maximum value. At the same time, the site and species scores stabilize. The resulting scores have maximum dispersion and thus constitute the first CA axis.

If we had started from a different set of initial site scores or from a set of arbitrary species scores, the iteration process would still have resulted in the same ordination axis. In Table 5.1c, the species and sites are rearranged in order of their scores on the first CA axis and show a clear diagonal structure.

A second ordination axis can also be extracted from the species data. The need for a second axis may be illustrated in Table 5.1c; Site 1 and Site 19 lie close together along the first axis and yet differ a great deal in species composition. This difference can be expressed on a second axis. The second axis is extracted by the same iteration process, with one extra step in which the trial scores for the second axis are made uncorrelated with the scores of the first axis. This can be done by plotting in each cycle the trial site scores for the second axis against the site scores of the first axis and fitting a straight line by a (weighted) leastsquares regression (the weights are y_{+1}/y_{++}). The residuals from this regression (i.e. the vertical deviations from the fitted line: Figure 3.1) are the new trial scores. They can be obtained more quickly by the orthogonalization procedure described in Table 5.2b. The iteration process would lead to the first axis again without the extra step. The intention is thus to extract information from the species data in addition to the information extracted by the first axis. In Figure 5.4, the final site scores of the second axis are plotted against those of the first axis. Site 1 and Site 19 lie far apart on the second axis, which reflects their difference in species composition. A third axis can be derived in the same way by making the scores uncorrelated with the scores of the first two axes, and so on. Table 5.2a summarizes the algorithm of two-way weighted averaging. A worked example is given in Exercise 5.1 and its solution.

In mathematics, the ordination axes of CA are termed eigenvectors (a vector is a set of values, commonly denoting a point in a multidimensional space and 'eigen' is German for 'self'). If we carry out an extra iteration cycle, the scores (values) remain the same, so the vector is transformed into itself, hence, the term eigenvector. Each eigenvector has a corresponding eigenvalue, often denoted by λ (the term is explained in Exercise 5.1.3). The eigenvalue is actually equal to the (maximized) dispersion of the species scores on the ordination axis, and is thus a measure of importance of the ordination axis. The first ordination axis has the largest eigenvalue (λ_1), the second axis the second largest eigenvalue (λ_2).



Figure 5.4 CA ordination diagram of the Dune Meadow Data in Hill's scaling. In this and the following ordination diagrams, the first axis is horizontal and the second axis vertical; the sites are represented by crosses and labelled by their number in Table 5.1; species names are abbreviated as in Table 0.1.

and so on. The eigenvalues of CA all lie between 0 and 1. Values over 0.5 often denote a good separation of the species along the axis. For the Dune Meadow Data, $\lambda_1 = 0.53$; $\lambda_2 = 0.40$; $\lambda_3 = 0.26$; $\lambda_4 = 0.17$. As λ_3 is small compared to λ_1 and λ_2 , we ignore the third and higher numbered ordination axes, and expect the first two ordination axes to display the biologically relevant information (Figure 5.4).

When preparing an ordination diagram, we plot the site scores and the species scores of one ordination axis against those of another. Because ordination axes differ in importance, one would wish the scores to be spread out most along the most important axis. But our site scores do not do so, because we standardized them to variance 1 for convenience in the algorithm (Table 5.2). An attractive standardization is obtained by requiring that the average width of the species curves is the same for each axis. As is clear from Figure 5.3b, the width of the curve for Species D is reflected in the spread among its presences along the axis. Therefore, the average curve width along an axis can be estimated from the data. For example, Hill (1979) proposed to calculate, for each species, the variance of the scores of the sites containing the species and to take the (weighted) average of the variances so obtained, i.e. Hill proposed to calculate

$\Sigma_k y_{k+} [\Sigma_i y_{ki} (x_i - u_k)^2 / y_{k+}] / y_{++}.$

To equalize the average curve width among different axes, we must therefore divide all scores of an axis by its average curve width (i.e. by the square root of the value obtained above). This method of standardization is used in the computer program DECORANA (Hill 1979a). Other than in Table 5.2, the program further uses the convention that site scores are weighted averages of species scores; so we must iterate Step 3 of our algorithm once more, before applying the standardization procedure just described. This scaling has already been used in preparing Figure 5.4 and we shall refer to it as Hill's scaling. A short cut to obtain Hill's scaling from the scores obtained from our algorithm is to divide the site scores after convergence by $\sqrt{(1 - \lambda)}/\lambda$ and the species scores by $\sqrt{\lambda(1 - \lambda)}$. The scores so obtained are expressed in multiples of one standard deviation (s.d.) and have the interpretation that sites that differ by 4 s.d. in score tend to have few species in common (Figure 5.3b). This use of s.d. will be discussed further in Subsection 5.2.4.

CA cannot be applied on data that contain negative values. So the data should not be centred or standardized (Subsection 2.4.4). If the abundance data of each species have a highly skew distribution with many small values and a few extremely large values, we recommend transforming them by taking logarithms:

 $\log_e (y_{ki} + 1)$, as in Subsection 3.3.1. By doing so, we prevent a few high values from unduly influencing the analysis. In CA, a species is implicitly weighted by its relative total abundance y_{k+}/y_{++} and, similarly, a site is weighted by y_{+i}/y_{++} . If we want to give a particular species, for example, triple its weight, we must multiply all its abundance values by 3. Sites can also be given greater or smaller weight by multiplying their abundance values by constants (ter Braak 1987b).

5.2.3 Diagonal structures: properties and faults of correspondence analysis

Table 5.3a shows artificial data in which the occurrences of species across sites appear rather chaotic and Table 5.3b shows the same data after arranging the species and sites in order of their score on the first CA axis. The data are rearranged into a perfectly diagonal table, also termed a two-way Petrie matrix. (A Petrie matrix is an incidence matrix that has a block of consecutive ones in every row; the matrix is two-way Petrie if the matrix also has a block of consecutive ones in every column, the block in the first column starting in the first row and the block of the last column ending in the last row.) For any table that permits such a rearrangement, we can discover the correct order of species and sites from the scores of the first axis of CA. This property of CA can be generalized to quantitative data (Gifi 1981) and to (one-way) Petrie matrices (Heiser 1981; 1986). For two-way Petrie matrices with many species and sites and with about equal numbers of occurrences per species and per site, the first eigenvalue is close to I; e.g. for Table 5.3, $\lambda_1 = 0.87$.

Note that CA does not reveal the diagonal structure if the ones and zeros are interchanged. Their role is asymmetrical, as is clear from the reciprocal averaging algorithm. The ones are important; the zeros are disregarded. Many ecologists feel the same sort of asymmetry between presences and absences of species.

The ordination of Table 5.3 illustrates two 'faults' of CA (Figure 5.5). First, the change in species composition between consecutive sites in Table 5.3. Column b is constant (one species appears; one disappears) and one would therefore wish that this constant change were reflected in equal distances between scores of neighbouring sites along the first axis. But the site scores at the ends of the first axis are closer together than those in the middle of the axis (Figure 5.5b). Secondly, the species composition is explained perfectly by the ordering of the sites and species along the first axis (Table 5.3, Column b) and the importance of the second axis should therefore be zero. However $\lambda_2 = 0.57$ and the site scores on the second axis show a quadratic relation with those on the first axis (Figure 5.5a). This fault is termed the arch effect. The term 'horseshoe' is also in use but is less appropriate, as the ends do not fold inwards in CA.

Table 5.3 CA applied to artificial data (~ denotes absence). Column a: The table looks chaotic. Column b: After rearrangement of species and sites in order of their scores on the first CA axis (u_k and x_i), a two-way Petrie matrix appears: $\lambda_1 = 0.87$.

Column a		Column b		u_k
Species	Sites 1 2 3 4 5 6 7	Species	Sites 1 7 2 4 6 5 3	
A B C D E F G H I	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	A B C E F D G H I I	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-1.40 -1.24 -1.03 -0.56 0.00 0.56 1.03 1.24 1.40
		x,	1 1 0 0 0 1 1 4 0 6 0 6 0 4 0 8 0 0 0 8 0	



Figure 5.5 Ordination by CA of the two-way Petrie matrix of Table 5.3. a: Arch effect in the ordination diagram (Hill's scaling; sites labelled as in Table 5.3; species not shown). b: One-dimensional CA ordination (the first axis scores of Figure a, showing that sites at the ends of the axis are closer together than sites near the middle of the axis. c: Onedimensional DCA ordination, obtained by nonlinearly rescaling the first CA axis. The sites would not show variation on the second axis of DCA.

Let us now give a qualitative explanation of the arch effect. Recall that the first CA axis maximally separates the species curves by maximizing the dispersion (Equation 5.3) and that the second CA axis also tries to do so but subject to the constraint of being uncorrelated with the first axis (Subsection 5.2.1). If the first axis fully explains the species data in the way of Figure 5.3b, then a possible second axis is obtained by folding the first axis in the middle and bringing the ends together (Figure 5.3c). This folded axis has no linear correlation with the first axis. The axis so obtained separates the species curves, at least Species C from Species B and D, and these from Species A and E, and is thus a strong candidate for the second axis of CA. Commonly CA will modify this folded axis somewhat, to maximize its dispersion, but the order of the site and species scores on the second CA axis will essentially be the same as that of the folded axis. Even if there is a true second underlying gradient, CA will not take it to be the second axis if its dispersion is less than that of the modified folded first axis. The intention in constructing the second CA axis is to express new information, but CA does not succeed in doing so if the arch effect appears.

5.2.4 Detrended correspondence analysis (DCA)

Hill & Gauch (1980) developed detrended correspondence analysis (DCA) as a heuristic modification of CA, designed to correct its two major 'faults': (1) that the ends of the axes are often compressed relative to the axes middle; (2) that the second axis frequently shows a systematic, often quadratic relation with the first axis (Figure 5.5). The major of these is the arch effect.

The arch effect is 'a mathematical artifact, corresponding to no real structure in the data' (Hill & Gauch 1980). They eliminate it by 'detrending'. Detrending is intended to ensure that, at any point along the first axis, the mean value of the site scores on the subsequent axes is about zero. To this end, the first axis is divided into a number of segments and within each segment the site scores on Axis 2 are adjusted by subtracting their mean (Figure 5.6). In the computer program DECORANA (Hill 1979a), running segments are used for this purpose. This process of detrending is built into the two-way weighted averaging algorithm, and replaces the usual orthogonalization procedure (Table 5.2). Subsequent axes are derived similarly by detrending with respect to each of the existing axes. Detrending applied to Table 5.3 gives a second eigenvalue of 0, as required.

The other fault of CA is that the site scores at the end of the first axis are often closer together than those in the middle of the axis (Figure 5.5b). Through this fault, the species curves tend to be narrower near the ends of the axis than in the middle. Hill & Gauch (1980) remedied this fault by nonlinearly rescaling the axis in such a way that the curve widths were practically equal. Hill & Gauch (1980) based their method on the tolerances of Gaussian response curves for the species, using the term standard deviation (s.d.) instead of tolerance. They noted that the variance of the optima of species present at a site (the 'within-site variance') is an estimate of the average squared tolerance of those species. Rescaling must therefore equalize the within-site variances as nearly as possible. For rescaling, the ordination axis is divided into small segments; the species ordination is expanded in segments with sites with small within-site variance and contracted in segments with sites with high within-site variance. Subsequently, the site scores are calculated by taking weighted averages of the species scores and the scores of sites and species are standardized such that the within-site variance equals 1. The tolerances of the curves of species will therefore approach 1. Hill & Gauch (1980) further define the length of the ordination axis to be the range of the site scores. This length is expressed in multiples of the standard deviation, abbreviated as s.d.



Figure 5.6 Method of detrending by segments (simplified). The crosses indicate site scores before detrending; the dots are site scores after detrending. The dots are obtained by subtracting, within each of the five segments, the mean of the trial scores of the second axis (after Hill & Gauch 1980).

The use of s.d. is attractive: a Gaussian response curve with tolerance 1 rises and falls over an interval of about 4 s.d. (Figure 3.6). Because of the rescaling, most species will about have this tolerance. Sites that differ 4 s.d. in scores can therefore be expected to have no species in common. Rescaling of the CA axis of Table 5.3 results in the desired equal spacing of the site scores (Figure 5.5c); the length of the axis is 6 s.d.

DCA applied to the Dune Meadow Data gives, as always, the same first eigenvalue (0.53) as CA and a lower second eigenvalue (0.29 compared to 0.40 in CA). The lengths of the first two axes are estimated as 3.7 and 3.1 s.d., respectively. Because the first axis length is close to 4 s.d., we predict that sites at opposite ends of the first axis have hardly any species in common. This prediction can be verified in Table 5.1c (the order of DCA scores on the first axis is identical to that of CA); Site 17 and Site 16 have no species in common, but closer sites have one or more species in common. The DCA ordination diagram (Figure 5.7) shows the same overall pattern as the CA diagram of Figure 5.4. There are, however,



Figure 5.7 DCA ordination diagram of the Dune Meadow Data. The scale marks are in multiples of the standard deviation (s.d.).

differences in details. The arch seen in Figure 5.4 is less conspicuous, the position of Sites 17 and 19 is less aberrant. Further, *Achillea millefolium* is moved from a position close to Sites 2, 5, 6, 7 and 10 to the bottom left of Figure 5.7 and is then closest to Site 1; this move is unwanted, as this species is most abundant in the former group of sites (Table 5.1).

In an extentive simulation study, Minchin (1987) found that DCA, as available in the program DECORANA, can flatten out some of the variation associated with one of the underlying gradients. He ascribed this loss of information to an instability in either, or both, detrending and rescaling. Pielou (1984, p. 197) warned that DCA is 'overzealous' in correcting the 'defects' in CA and that it 'may sometimes lead to the unwitting destruction of ecologically meaningful information'.

DCA is popular among practical field ecologists; presumably because it provides an effective approximate solution to the ordination problem for a unimodal response model in two or more dimensions - given that the data are reasonably representative of sections of the major underlying environmental gradients. Two modifications might increase its robustness with respect to the problems identified by Minchin (1987). First, nonlinear rescaling aggravates these problems; since the edge effect is not too serious, we advise against the routine use of nonlinear rescaling. Second, the arch effect needs to be removed, but this can be done by a more stable, less 'zealous' method of detrending, which was also briefly mentioned by Hill & Gauch (1980): detrending-by-polynomials. The arch is caused by the folding of the first axis (Figure 5.3c), so that the second CA axis is about a quadratic function of the first axis, the third CA axis a cubic function of the first axis, and so on (Hill 1974). The arch is therefore most simply removed by requiring that the second axis is not only uncorrelated with the first axis (x_i) , but also uncorrelated with its square (x_i^2) and, to prevent more folding, its cube (x_i^3) . In contrast with 'detrending-by-segments', the method of detrending-bypolynomials removes only specific defects of CA that are now theoretically understood. Detrending by polynomials can be incorporated into the two-way weighted averaging algorithm (Table 5.2) by extending Step 4 such that the trial scores are not only made uncorrelated with the previous axes, but also with polynomials of previous axes. The computer program CANOCO (ter Braak 1987b) allows detrending by up to fourth-order polynomials.

5.2.5 Joint plot of species and sites

An ordination diagram mirrors the species data (although often with some distortion), so we can make inferences about the species data from the diagram. With Hill's scaling (Subsection 5.2.2), site scores are weighted averages of the species scores. Site points then lie in the ordination diagram at the centroid of the points of species that occur in them. Sites that lie close to the point of a species are therefore likely to have a high abundance of that species or, for presence-absence data, are likely to contain that species. Also, in so far as CA and DCA are a good approximation to fitting bell-shaped response surfaces to the species data (Subsection 5.2.1 and Section 5.7), the species points are close

108

to the optima of these surfaces; hence, the expected abundance or probability of occurrence of a species decreases with distance from its position in the plot (Figure 3.14).

Using these rules to interpret DCA diagrams, we predict as an example the rank order of species abundance for three species from Figure 5.7 and compare the order with the data in Table 5.1. The predicted rank order for *Juncus bufonius* is Sites 12, 8, 13, 9, 18 and 4; in the data *Juncus bufonius* is present at four sites, in order of abundance Sites 9, 12, 13 and 7. The predicted rank order for *Rumex acetosa* is Sites 5, 7, 6, 10, 2 and 11; in the data *R. acetosa* occurs in five sites, in order of abundance Sites 6, 5, 7, 9 and 12. *Ranunculus flammula* is predicted to be most abundant at Sites 20, 14, 15, 16 and less abundant, if present at all, at Sites 8, 12 and 13; in the data, *R. flammula* is present in six sites, in order of abundance Sites 20, 14, 15, 16, 8 and 13. We see some agreement between observations and predictions but also some disagreement. What is called for is a measure of goodness of fit of the ordination diagram. Such a measure is, however, not normally available in CA and DCA.

In interpreting ordination diagrams of CA and DCA, one should be aware of the following aspects. Species points on the edge of the diagram are often rare species, lying there either because they prefer extreme (environmental) conditions or because their few occurrences by chance happen to be at sites with extreme conditions. One can only decide between these two possibilities by additional external knowledge. Such species have little influence on the analysis; if one wants to enlarge the remainder of the diagram, it may be convenient not to display them at all. Further, because of the shortcomings of the method of weighted averaging, species at the centre of the diagram may either be unimodal with optima at the centre, or bimodal, or unrelated to the ordination axes. Which possibility is most likely can be decided upon by table rearrangement as in Table 5.1c or by plotting the abundance of a species against the axes. Species that lie between the centre and the outer edge are most likely to show a clear relation with the axes.

5.2.6 Block structures and sensitivity to rare species

CA has attractive properties in the search for block structures. A table is said to have block structure if its sites and species can be divided into clusters, with each cluster of species occurring in a single cluster of sites (Table 5.4). For any table that allows such a clustering, CA will discover it without fail. With the four blocks in Table 5.4, the first three eigenvalues of CA equal 1 and sites from the same cluster have equal scores on the three corresponding axes. An eigenvalue close to 1 can therefore point to an almost perfect block structure or to a diagonal structure in the data (Subsection 5.2.3). The search for block structures or 'nearblock structures' by CA forms the basis of the cluster-analysis program TWINSPAN (Chapter 6).

This property of CA is, however, a disadvantage in ordination. If a table contains two disjoint blocks, one of which consists of a single species and a single site, then the first axis of CA finds this questionably uninteresting block. For a similar Table 5.4 Data table with block structure. Outside the Sub-tables A_1 , A_2 , A_3 and A_4 , there are no presences, so that there are four clusters of sites that have no species in common ($\lambda_1 = 1$, $\lambda_2 = 1$, $\lambda_3 = 1$).



reason, CA is sensitive to species that occur only in a few species-poor sites. In the 'down-weighting' option of the program DECORANA (Hill 1979a), species that occur in a few sites are given a low weight, so minimizing their influence, but this does not fully cure CA's sensitivity to rare species at species-poor sites.

5.2:7 Gaussian ordination and its relation with CA and DCA

In the introduction to CA (Subsection 5.2.1), we assumed that species show unimodal response curves to environmental variables, intuitively took the dispersion of the species scores as a plausible measure of how well an environmental variable explains the species data, and subsequently defined CA to be the technique that constructs a theoretical variable that explains the species data best in the sense of maximizing the dispersion. Because of the shortcomings of CA noted in the subsequent sections, the dispersion of the species scores is not ideal to measure the fit to the species data. We now take a similar approach but with a better measure of fit and assume particular unimodal response curves. We will introduce ordination techniques that are based on the maximum likelihood principle (Subsections 3.3.2 and 4.2.1), in particular Gaussian ordination, which is a theoretically sound but computationally demanding technique of ordination. We also show that the simpler techniques of CA and DCA give about the same result if particular additional conditions hold true. This subsection may now be skipped at first reading; it requires a working knowledge of Chapters 3 and 4. In maximum likelihood ordination, a particular response model (Subsection 3.1.2) is fitted to the species data by using the maximum likelihood principle. In this approach, the fit is measured by the deviance (Subsection 3.3.2) between the data and the fitted curves. Recall that the deviance is inversely related to the likelihood, namely deviance $= -2 \log_{e}$ (likelihood). If we fit Gaussian (logit) curves (Figure 3.9) to the data, we obtain Gaussian ordination. In Subsection 3.3.3, we fitted a Gaussian logit curve of pH to the presence-absence data of a particular species (Figure 3.10). In principle, we can fit a separate curve for each species under consideration. A measure of how badly pH explains the species data is then the deviance (Table 3.6) summed over all species. Gaussian ordination of presence-absence data is then the technique that constructs the theoretical variable that best explains the species data by Gaussian logit curves, i.e. that minimizes the deviance between the data and the fitted curves.

A similar approach can be used for abundance data by fitting Gaussian curves to the data, as in Section 3.4, with the assumption that the abundance data follow a Poisson distribution. A Gaussian curve for a particular species has three parameters: optimum, tolerance and maximum (Figure 3.6), for species k denoted by u_k , t_k and c_k , respectively. In line with Equation 3.8, the Gaussian curves can now be written as

$$Ey_{ki} = c_k \exp \left[-0.5(x_i - u_k)^2 / t_k^2 \right]$$
 Equation 5.4

where x_i is the score of site *i* on the ordination axis (the value of the theoretical variable at site *i*).

To fit this response model to data we can use an algorithm akin to that to obtain the ordination axis in CA (Table 5.2).

- Step 1: Start from initial site scores x_i .
- Step 2: Calculate new species scores by (log-linear) regression of the species data on the site scores (Section 3.4). For each species, we so obtain new values for u_k , t_k and c_k .
- Step 3: Calculate new site scores by maximum likelihood calibration (Subsection 4.2.1).
- Step 4: Standardize the site scores and check whether they have changed and, if so, go back to Step 2, otherwise stop.

In this algorithm, the ordination problem is solved by solving the regression problem (Chapter 3) and the calibration problem (Chapter 4) in an iterative fashion so as to maximize the likelihood. In contrast to the algorithm for CA, this algorithm may give different results for different initial site scores because of local maxima in the likelihood function for Equation 5.4. It is therefore not guaranteed that the algorithm actually leads to the (overall) maximum likelihood estimates; hence, we must supply 'good' initial scores, which are also needed to reduce the computational burden. Even for modern computers, the algorithm requires heavy computation. In the following, we show that a good choice for initial scores are the scores obtained by CA.

The CA algorithm can be thought of a simplification of the maximum likelihood algorithm. In CA, the regression and calibration problems are both solved by weighted averaging. Recall that in CA the species score (u_k) is a position on the ordination axis x indicating the value most preferred by that particular species (its optimum) and that the site score (x_i) is the position of that particular site on the axis.

We saw in Section 3.7 that the optimum or score of a species (u_k) can be estimated efficiently by weighted averaging of site scores provided that (Figure 3.18b):

A1. the site scores are homogeneously distributed over the whole range of occurrence of the species along the axis x.

In Section 4.3, we saw that the score (x_i) of a site is estimated efficiently by weighted averaging of species optima provided the species packing model holds, i.e. provided (Figure 4.1):

A2. the species' optima (scores) are homogeneously distributed over a large interval around x_{i} .

A3. the tolerances of species t_k are equal (or at least independent of the optima; ter Braak 1985).

A4. the maxima of species c_k are equal (or at least independent of the optima; ter Braak 1985).

Under these four conditions the scores obtained by CA approximate the maximum likelihood estimates of the optima of species and the site values in Gaussian ordination (ter Braak 1985). For presence-absence data, CA approximates similarly the maximum likelihood estimates of the Gaussian logit model (Subsection 3.3.3). CA does not, however, provide estimates for the maximum and tolerance of a species.

A problem is that assumptions A1 and A2 cannot be satisfied simultaneously for all sites and species: the first assumption requires that the range of the species optima is amply contained in the range of the site scores whereas the second assumption requires the reverse. So CA scores show the edge effect of compression of the end of the first axis relative to the axis middle (Subsection 5.2.3). In practice, the ranges may coincide or may only partly overlap. CA does not give any clue about which possibility is likely to be true. The algorithm in Table 5.2 results in species scores that are weighted averages of the site scores and, consequently, the range of the species scores is contained in the range of the site scores. But it is equally valid mathematically to stop at Step 3 of the algorithm, so that the site scores are weighted averages of the species scores and thus all lie within the range of the species scores; this is done in the computer program DECORANA (Hill 1979). The choice between these alternatives is arbitrary. It may help interpretation of CA results to go one step further in the direction of the maximum likelihood estimates by one regression step in which the data of each species are regressed on the site scores of CA by using the Gaussian response model. This can be done by methods discussed in Chapter 3. The result is new species scores (optima) as well as estimates for the tolerances and maxima. As an example, Figure 5.8 shows Gaussian response curves along the first CA axis fitted to the



Figure 5.8 Gaussian response curves for some Dune Meadow species, fitted by log-linear regression of the abundances of species (Table 5.1) on the first CA axis. The sites are shown as small vertical lines below the horizontal axis.

Dune Meadow Data in Table 5.1. The curve of a particular species was obtained by a log-linear regression (Section 3.4) of the data of the species on the site scores of the first CA axis by using $b_0 + b_1 x + b_2 x^2$ in the linear predictor (Equation 3.18).

Two dimensions

In two dimensions, Gaussian ordination means fitting the bivariate Gaussian surfaces (Figure 3.14)

$$Ey_{ki} = c_k \exp\left(-0.5[(x_{i1} - u_{k1})^2 + (x_{i2} - u_{k2})^2]/t_k^2\right)$$
 Equation 5.5

where

 (u_{kl}, u_{k2}) are the coordinates of the optimum of species k in the ordination diagram c_k is the maximum of the surface

 t_k is the tolerance

 (x_{i1}, x_{i2}) are the coordinates of site *i* in the diagram.

These Gaussian surfaces look like that of Figure 3.14, but have circular contours because the tolerances are taken to be the same in both dimensions.

One cannot hope for more than that the two-axis solution of CA provides an approximation to the fitting of Equation 5.5 if the sampling distribution of the abundance data is Poisson and if:

A1. site points are homogeneously distributed over a rectangular region in the ordination diagram with sides that are long compared to the tolerances of the species,

A2. optima of species are homogeneously distributed over the same region,

A3. the tolerances of species are equal (or at least independent of the optima), A4. the maxima of species are equal (or at least independent of the optima).

However as soon as the sides of the rectangular region differ in length, the arch effect (Subsection 5.2.3) crops up and the approximation is bad. Figure 5.9b shows the site ordination diagram obtained by applying CA to artificial species data (40 species and 50 sites) simulated from Equation 5.5 with $c_k = 5$ and $t_k = 1$ for each k. The true site points were completely randomly distributed over



Figure 5.9 CA applied to simulated species data. a: True configuration of sites (•). b: Configuration of sites obtained by CA, showing the arch effect. The data were obtained from the Gaussian model of Equation 5.5 with Poisson error, $c_k = 5$, $t_k = 1$ and optima that were randomly distributed in the rectangle $[-1,9] \times [-0.5,4.5]$. The vertical lines in Figures a and b connect identical sites.

a rectangular region with sides of 8 and 4 s.d. (Figure 5.9a). The CA ordination diagram is dominated by the arch effect, although the actual position of sites within the arch still reflects their position on the second axis in Figure 5.9a. The configuration of site scores obtained by DCA was much closer to the true configuration. DCA forcibly imposes Conditions A1, A2 and A3 upon the solution, the first one by detrending and the second and third one by rescaling of the axes.

We also may improve the ordination diagram of DCA by going one step further in the direction of maximum likelihood ordination by one extra regression step. We did so for the DCA ordination (Figure 5.7) of Dune Meadow Data in Table 0.1. For each species with more than 4 presences, we carried out a log-linear regression of the data of the species on the first two DCA axes using the response model

$$\log_{e} Ey_{ki} = b_0 + b_{1k} x_{i1} + b_{3k} x_{i2} + b_{4k} (x_{i1}^2 + x_{i2}^2)$$
 Equation 5.6

where x_{i1} and x_{i2} are the scores of site *i* on the DCA axes 1 and 2, respectively.

If $b_{4k} < 0$, this model is equivalent to Equation 5.5 (as in Subsection 3.3.3). The new species scores are then obtained from the estimated parameters in Equation 5.6 by $u_{k1} = -b_{1k}/(2b_{4k})$, $u_{k2} = -b_2/(2b_{4k})$ and $t_k = 1/\sqrt{(-2b_{4k})}$.

If $b_{4k} > 0$, the fitted surface shows a minimum and we have just plotted the DCA scores of the species. Figure 5.10a shows how the species points obtained by DCA change by applying this regression method to the 20 species with four or more presences. A notable feature is that *Achillea millefolium* moves towards its position in the CA diagram (Figure 5.4). In Figure 5.10b, circles are drawn with centres at the estimated species points and with radius t_k . The circles are contours where the expected abundance is 60% of the maximum expected abundance c_k . Note that exp (-0.5) = 0.60.

From Figure 5.10b, we see, for example, that *Trifolium repens* has a high tolerance (a large circle, thus a wide ecological amplitude) whereas *Bromus hordaceus* has a low tolerance (a small circle, thus a narrow ecological amplitude). With regression, the joint plot of DCA can be interpreted with more confidence. This approach also leads to a measure of goodness of fit. A convenient measure of goodness of fit is here

$$V = 1 - (\Sigma_k D_{k1}) / (\Sigma_k D_{k0})$$
 Equation 5.7

where D_{k0} and D_{k1} are the residual deviances of the kth species for the null model (the model without explanatory variables) and the model depicted in the diagram (Equation 5.6), respectively. These deviances are obtained from the regressions (as in Table 3.7). We propose to term V the fraction of deviance accounted for by the diagram. For the two-axis ordination (only partially displayed in Figure 5.10b) V = (1 - 360/987) = 0.64. For comparison, V = 0.51 for the one-axis ordination (partially displayed in Figure 5.8).



Figure 5.10 Gaussian response surfaces for several Dune Meadow species fitted by loglinear regression of the abundances of species on the site scores of the first two DCA axes (Figure 5.7). a: Arrows for species running from their DCA scores (Figure 5.7) to their fitted optimum. b: Optima and contours for some of the species. The contour indicates where the abundance of a species is 60% of the abundance at its optimum.

The regression approach can of course be extended to more complicated surfaces (e.g. Equation 3.24), but this will often be impractical, because these surfaces are more difficult to represent graphically.

5.3 Principal components analysis (PCA)

5.3.1 From least-squares regression to principal components analysis

.

Principal components analysis (PCA) can be considered to be an extension of fitting straight lines and planes by least-squares regression. We will introduce



Figure 5.10b

٠.

PCA, assuming the species data to be quantitative abundance values.

Suppose we want to explain the abundance values of several species by a particular environmental variable, say moisture, and suppose we attempt to do so by fitting straight lines to the data. Then, for each species, we have to carry out a least-squares regression of its abundance values on the moisture values and obtain, among other things, the residual sum of squares, i.e. the sum of squared vertical distances between the observed abundance values and the fitted line (Figure 3.1; Subsection 3.2.2). This is a measure of how badly moisture explains the data of a single species. To measure how badly moisture explains the data of all species, we now use the total of the separate residual sums of squares over all species, abbreviated the total residual sum of squares. If the total residual sum of squares is small, moisture can explain the species data well.

Now, suppose that, among a set of environmental variables, moisture is the variable that best explains the species data in the sense of giving the least total residual sum of squares. As in all ordination techniques, we now wish to construct a theoretical variable that explains the species data still better. PCA is the ordination technique that constructs the theoretical variable that minimizes the total residual sum of squares after fitting straight lines to the species data. PCA does so by

choosing best values for the sites, the site scores. This is illustrated in Figure 5.11 for the Dune Meadow Data. The site scores are indicated by ticks below the horizontal axis. The fitted lines are shown for six of the 30 species and the observed abundance values and residuals for one of them. Any other choice of site scores would result in a larger sum of squared residuals. Note that Figure 5.11 shows only 20 out of all $20 \times 30 = 600$ residuals involved. The horizontal axis in Figure 5.11 is the first PCA axis, or first principal component. The score of a species in PCA is actually the slope of the line fitted for the species against the PCA axis. A positive species score thus means that the abundance increases along the axis (e.g. Agrostis stolonifera in Figure 5.11); a negative score means that the abundance decreases along the axis (e.g. Lolium perenne in Figure 5.11) and a score near 0 that the abundance is poorly (linearly) related to the axis (e.g. Sagina procumbens in Figure 5.11).

If a single variable cannot explain the species data sufficiently well, we may attempt to explain the data with two variables by fitting planes (Subsection 3.5.2). Then, for each species we have to carry out a least-squares regression of its abundance values on two explanatory variables (Figure 3.11), obtain its residual sum of squares and, by addition over species, the total residual sum of squares. The first two axes of PCA are now the theoretical variables minimizing the total residual sum of squares among all possible choices of two explanatory variables. Analogously, the first three PCA axes minimize the total residual sum of squares by fitting the data to hyperplanes, and so on. PCA is thus a multi-species extension of multiple (least-squares) regression. The difference is that in multiple regression



Figure 5.11 Straight lines for several Dune Meadow species, fitted by PCA to the species abundances of Table 5.1. Also shown are the abundances of *Lolium perenne* and their deviations from the fitted straight line. The horizontal axis is the first principal component. Fitting straight lines by least-squares regression of the abundances of species on the site scores of the first PCA axis gives the same results. The slope equals the species score of the first axis. The site scores are shown by small vertical lines below the horizontal axis.

the explanatory variables are supplied environmental variables whereas in PCA the explanatory variable are theoretical variables estimated from the species data alone. It can be shown (e.g. Rao 1973) that the same result as above is obtained by defining the PCA axes sequentially as follows. The first PCA axis is the variable that explains the species data best, and second and later axes also explain the species data best but subject to the constraint of being uncorrelated with previous PCA axes. In practice, we ignore higher numbered PCA axes that explain only a small proportion of variance in the species data.

5.3.2 Two-way weighted summation algorithm

We now describe an algorithm that has much in common with that of CA and that gives the ordination axes of PCA. The algorithm also shows PCA to be a natural extension of straight-line regression.

If the relation between the abundance of a species and an environmental variable is rectilinear, we can summarize the relation by the intercept and slope of a straight line. The error part of the model is taken to consist of independent and normally distributed errors with a constant variance. The parameters (intercept and slope) are then estimated by least-squares regression of the species abundances on the values of the environmental variable (Subsection 3.2.2). Conversely, when the intercepts and slopes are known, we can estimate the value of the environmental variable from the species abundances at a site by calibration (Subsection 4.2.3). If it is not known in advance which environmental variable determines the abundances of the species, the idea is as in CA (Subsection 5.2.2) to discover the 'underlying environmental gradient' by applying straight-line regression and calibration alternately in an iterative fashion, starting from arbitrary initial values for sites or from arbitrary initial values for the intercepts and slopes of species. As in CA, the iteration process eventually converges to a set of values for species and sites that does not depend on the initial values.

The iteration process reduces to simple calculations when we first centre the abundances of each species to mean 0 and standardize the site scores to $\bar{x} = 0$ and $\Sigma_i (x_i - \bar{x})^2 = 1$. Then, the equations to estimate the intercept and the slope of a straight line (Equations 3.6a,b) reduce to $b_0 = 0$ and $b_1 = \Sigma_i y_i x_i$, because in the notation of Subsection 3.2.2 $\bar{y} = 0$, $\bar{x} = 0$ and $\Sigma_i (x_i - \bar{x})^2 = 1$. Hence we ignore the intercepts and concentrate on the slope parameters. From now on, b_k will denote the slope parameter for species k and y_{ki} the centred abundance of species k at site i (i.e. $y_{k+} = 0$). In this notation, the slope parameter of species k is calculated by

$$\sum_{i=1}^{n} y_{ki} x_i$$

 $b_k =$

Equation 5.8

As an example, Table 5.5a shows the Dune Meadow Data used before with an extra column of species means and, as arbitrary initial scores for the sites, values obtained by standardizing the numbers 1 to 20 (bottom row). For Achillea millefolium, the mean abundance is 0.80 and we obtain

Table 5.5a			
Species k	Sites (1) 00000000011111111112 12345678901234567890	mean	b _k
1 Ach mil 2 Agr sto 3 Air pra 4 Alo gen 5 Ant odo 6 Bel pre 7 Bro hor 8 Che alb 9 Cir arv 10 Ele pal 11 Ely rep 12 Emp nig 13 Hyp rad 14 Jun art 15 Jun buf 16 Leo aut 17 Lol per 18 Pla lan 19 Poa pra 20 Poa tri 21 Pot pal 22 Rum ace 24 Sag pro 25 Sal rep 26 Tri pra 27 Tri rep 28 Vic lat 29 Cat cus	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0.80\\ 2.40\\ 0.25\\ 1.65\\ 0.75\\ 0.05\\ 1.30\\ 0.45\\ 0.30\\ 0.45\\ 0.30\\ 0.45\\ 0.90\\ 1.30\\ 0.45\\ 0.290\\ 1.30\\ 0.50\\ 1.00\\ 0.55\\ 0.20\\ 0.55\\ 0.20\\ 0.55\\ 0.20\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50\\ 0.55\\ 0.20\\ 0.50$	$\begin{array}{c} -1.98\\ 1.55\\ 1.49\\ -2.06\\ 0.696\\ -1.88\\ 0.150\\ -2.850\\ 0.501\\ -2.850\\ -0.521\\ -0.521\\ -0.59\\ 2.022\\ 0.033\\ -9.42\\ -1.24\\ -7.93\\ 0.52\\ 2.552\\ -0.70\\ -1.88\\ 0.31\\ 2.89\\ 2.25\\ 2.52\\ -0.31\\ 2.89\\ 2.29\\ -1.57\\ -1.88\\ 0.31\\ 2.89\\ 2.29\\ -0.52\\ -0.$
x _i	000000000000000000000000000000000000000		
	33222111000011122233 73951740622604715937		

Table 5.5 Two-way weighted summation algorithm of PCA applied to the Dune Meadow Data. a: The original data table with at the bottom the initial site scores. b: The species and sites rearranged in order of their scores obtained after one cycle of two-way weighted summation. c: The species arranged in order of their final scores (PCA scores).

. . -

Table 5.5b			Table 5.5c		
Species	Sites (i) 00010000011011111112 222041558218282748550	b_k	Species	Sites (i) 01000100100111011121 60725113894793824506	b_k
k 17 Lol per 20 Poa tri 11 Ely rep 19 Poa pra 7 Bro hor 23 Rum ace 4 Alo gen 1 Ach mil 6 Bel per 27 Tri rep 26 Tri pra 18 Pla Lan 9 Cir arv 24 Sag pro 15 Jun buf 8 Che alb 28 Vic Lat 5 Ant odo 21 Pot pal 12 For pig	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-9.42 -7.93 -6.17 -6.05 -2.52 -1.98 -1.98 -1.88 -1.57 -0.12 0.02 0.31 0.60 0.65	k 17 Lol per 18 Pla lan 19 Poa pra 20 Poa tri 1 Ach mil 23 Rum ace 27 Tri rep 5 Ant odo 7 Bro hor 16 Leo aut 11 Ely rep 26 Tri pra 6 Bel per 28 Vic lat 13 Hyp rad 9 Cir arv 12 Emp nig 8 Che alb 3 Air pra	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-9.21 -5.77 -5.69 -4.80 -3.68 -3.67 -3.531 -2.86 -2.61 -2.86 -2.611 -0.08 0.01 0.09 0.115 0.40
12 Emp nig 16 Leo aut 3 Air pra 2 Agr sto 14 Jun art 13 Hyp rad 30 Cal cus 22 Ran fla 29 Bra rut 25 Sal rep 10 Ele pal	52332 33225325226 22 2 3 4 8 35 44 4 745 4 4 334 2 2 5 4 3 3 2 2 2 224 2222 262 4246 3444 3 3 5 4 4 854 000000000000000000000000 32221111100001122334 36109987454174869220	0.93 1.49 1.55 2.02 2.19 2.29 2.52 2.89 3.70 4.21	29 Bra rut 24 Sag pro 21 Pot pal 25 Sal rep 4 Alo gen 30 Cal cus 22 Ran fla 14 Jun art 10 Ele pal 2 Rgr sto	622 24 2622 3 24 444 2 25 3224 22 3 3 5 2 7 32 558 4 4 33 22 242 4 4 343 4 4548 4 38 5444457 000000000000000000000000000000000000	0.94 0.98 1.07 1.85 3.33 3.40 3.95 4.29 8.08 8.67

•

120

 $b_1 = (1 - 0.80) \times (-0.37) + (3 - 0.80) \times (-0.33) + (0 - 0.80) \times (-0.29) + ... + (0 - 0.80) \times (0.37) = -1.98.$

From the slopes thus obtained (Table 5.5a, last column), we derive new site scores by least-squares calibration (Equation 4.2 with $a_k = 0$). The site scores so obtained are proportional to

 $x_i = \sum_{k=1}^m y_{ki} b_k$

Equation 5.9

because the denominator of Equation 4.1 has the same value for each site. This denominator is unimportant in PCA, because the next step in the algorithm is to standardize the site scores, as shown in Table 5.6c. For Site 1 in Table 5.5a. we get from Equation 5.9 the site score $x_1 = (1 - 0.80) \times (-1.98) + (0 - 2.40)$ \times (1.55) + (0 - 0.25) \times (1.49) + ... + (0 - 0.50) \times (2.29) = -0.19. Note that the species mean abundance is subtracted each time from the abundance value. In Table 5.5b, the species and sites are arranged in order of the scores obtained so far, in which the slopes (b_k) form the species scores. The abundance of the species in the top row (Lolium perenne) has the tendency to decrease along the row, whereas the abundance of the species in the bottom row (Eleocharis palustris) has the tendency to increase across the row. The next cycle of the iteration is to calculate new species scores (b_{ν}) , then new site scores, and so on. As in CA, the scores stabilize after several iterations and the resulting scores (Table 5.5c) constitute the first ordination axis of PCA. In Table 5.5c, the species and sites are arranged in order of their scores on the first axis. Going from top row to bottom row, we see first a decreasing trend in abundance across the columns (e.g. for Lolium perenne), then hardly any trend (e.g. for Sagina procumbens) and finally an increasing trend (e.g. for Agrostis stolonifera). A graphical display of the trends has already been shown in Figure 5.11. The order of species in Table 5.5c is quite different from the order in the table arranged by CA (Table 5.1c), but the difference in ordering of the sites is more subtle.

In the above iteration algorithm of PCA (Table 5.6), weighted sums (Equations 5.8 and 5.9) replace the weighted averages in CA (Table 5.2; Equations 5.1 and 5.2). For this analogy to hold, let us consider the data y_{ki} as weights (which can be negative in PCA), so that the species scores are a weighted sum of the site scores and, conversely, the site scores are a weighted sum of the species scores (Table 5.6). The standard terminology used in mathematics is that x_i is a linear combination of the variables (species) and that b_k is the loading of species k.

After the first axis, a second axis can be extracted as in CA, and so on. (There is a subtle difference in the orthogonalization procedure, which need not concern us here.) The axes are also eigenvectors to which correspond eigenvalues as in CA (Subsection 5.2.2). The meaning of the eigenvalues in PCA is given below. The axes are also termed principal components.

So PCA decomposes the observed values into fitted values and residuals (Equations 3.1 and 3.2). In one dimension, we have the decomposition

 $y_{ki} = b_k x_i + residual$

Equation 5.10

a: Iteration process

Step 1. Take arbitrary initial site scores (x_i) , not all equal to zero.

Step 2. Calculate new species scores (b_k) by weighted summation of the site scores (Equation 5.8).

Step 3. Calculate new site scores (x_i) by weighted summation of the species scores (Equation 5.9).

Step 4. For the first axis go to Step 5. For second and higher axes, make the site scores (x_i) uncorrelated with the previous axes by the orthogonalization procedure described below.

Step 5. Standardize the site scores (x_i) . See below for the standardization procedure.

Step 6. Stop on convergence, i.e. when the new site scores are sufficiently close to the site scores of the previous cycle of the iteration; ELSE go to Step 2.

b: Orthogonalization procedure

Step 4.1. Denote the site scores of the previous axis by f_i and the trial scores of the present axis by x_i .

Step 4.2, Calculate $v = \sum_{i=1}^{n} x_i f_i$. Step 4.3 Calculate $x_{i,new} = x_{i,old} - v f_i$. Step 4.4 Repeat Steps 4.1-4.3 for all previous axes.

c: Standardization procedure

Step 5.1 Calculate the sum of squares of the site scores $s^2 = \sum_{i=1}^{n} x_i^2$. Step 5.2 Calculate $x_{i,new} = x_{i,old}/s$. Note that, upon convergence, s equals the eigenvalue.

where y_{ki} is the (mean corrected) observed value and $b_k x_i$ the fitted value.

As an example, the values fitted by the first PCA axis (Table 5.5c) for the centred abundances of Agrostis stolonifera ($b_2 = 8.67$) at Site 6 ($x_6 = -0.31$) and Site 16 ($x_{16} = 0.45$) are: $8.67 \times (-0.31) = -2.75$ and $8.67 \times 0.45 = 3.99$, respectively. Adding the mean value of A. stolonifera (2.40), we obtain the values -0.35 and 6.39, respectively, which are close to the observed abundance values of 0 and 7 at Site 6 and Site 16. In PCA, the sum of squared residuals in Equation 5.10 is minimized (Subsection 5.3.1). Analogously, one can say that PCA maximizes the sum of squares of fitted values and the maximum is the eigenvalue of the first axis. In two dimensions (Figure 5.12), we have the decomposition

$$y_{ki} = (b_{k1} x_{i1} + b_{k2} x_{i2}) + \text{residual}$$

Equation 5.11

where

 b_{k1} and b_{k2} are the scores of species k

 x_{i1} and x_{i2} are the scores of site i on Axis 1 and Axis 2, respectively.

On the second axis, the score of A. stolonifera is 6.10 and the scores of Sites 6 and 16 are -0.17 and 0.033 (Figure 5.12), so that the fitted values become 8.67 \times (-0.31) + 6.10 \times (-0.17) = -3.72 and 8.67 \times 0.45 + 6.10 \times 0.033 = 4.10.



Figure 5.12 PCA-ordination diagram of the Dune Meadow Data in covariance biplot scaling with species represented by arrows. The b scale applies to species, the x scale to sites. Species not represented in the diagram lie close to the origin (0,0).

The first two PCA axes thus give approximate abundance values of -3.72 + 2.40 = -1.3 and 4.10 + 2.40 = 6.5, slightly worse values than those obtained from the first axis, but most of the remaining abundance values in the data table will be approximated better with two axes than with a single axis. The sum of squares of fitted values now equals $\lambda_1 + \lambda_2$. Further, the total sum of squares ($\Sigma_k \Sigma_i y_{ki}^2$) equals the sum of all eigenvalues. (This equality means that we can reconstruct the observed values exactly from the scores of species and sites on all eigenvectors and the mean abundance values.) The fraction of variance accounted for (explained) by the first two axes is therefore $(\lambda_1 + \lambda_2)/(\text{sum of all eigenvalues})$. This measure is the equivalent of R^2 in Section 3.2. For the Dune Meadow Data, $\lambda_1 = 471$, $\lambda_2 = 344$ and the total sum of squares = 1598. So the two-axes solution explains (471 + 344)/1598 = 51% of the variance. The first axis actually explains 471/1598 = 22%.

5.3.3 Best lines and planes in m-dimensional space

Here we present a geometric approach to PCA. In this approach, the aim of PCA is seen as being to summarize multivariate data in a graphical way. The approach is best illustrated with data on two species only. Figure 5.13a displays the abundances of Species A and B at 25 sites in the form of a scatter diagram, with axes labelled by the species names. The simplest summary of data is by the mean abundances of A (25) and B (15). Knowing the means, we may shift the axes to the centroid of the data points, i.e. to the point with the coordinates (25,15), provided we remember that the origin (0,0) of the new coordinate system is the point (25,15) in the old coordinate system. Next we draw a line through the new origin in the direction of maximum variance in the plot. This line is the first principal component (PC1), or first PCA axis, and perpendicularly we draw PC2. Next we rotate the plot, so that PC1 is horizontal (Figure 5.13b). Figure 5.13b is an ordination diagram with arrows representing the species. These arrows are the shifted and rotated axes of the species in the original diagram. PC2 shows so much less variation than PC1 that PC2 can possibly be neglected. This is done in Figure 5.13c showing a one-dimensional ordination; the points in Figure 5.13c were obtained from Figure 5.13b by drawing perpendicular lines from each point on the horizontal axis (projection onto PCI). In this way, the first coordinate of the points in Figure 5.13b is retained in Figure 5.13c; this coordinate is the site score on PC1. The first coordinate of the arrows in Figure 5.13b is the species loading on PC1, which is also represented by an arrow in Figure 5,13c. These arrows indicate the direction in which Species A and Species B increase in abundance; hence Figure 5.13c still shows which sites have high abundances of Species A and of Species B (those on the right side) and which sites have low abundance (those on the left side).

The example is, of course, artificial. Usually there are many species $(m \ge 3)$, so that we need an m-dimensional coordinate system, and we want to derive a two-dimensional or three-dimensional ordination diagram. Yet the principle remains the same: PCA searches for the direction of maximum variance: this is PC1, the best line through the data points. It is the best line in the sense that it minimizes the sum of squares of perpendicular distances between the data points and the line (as is illustrated in Figure 5.13a for m = 2). So the first component in Figure 5.13a is neither the regression line of Species B on Species A nor that of Species A on Species B, because regression minimizes the sum of squares of vertical distances (Figure 3.1). But, as we have seen in Subsection 5.3.1, PCA does give the best regression of Species A on PC1 and of Species B on PC1 (Figure 5.11). After the first component, PCA seeks the direction of maximum variance that is perpendicular onto the first axis; that is PC2, which with PCI forms the best plane through the data points, and so on. In general, the site scores are obtained by projecting each data point from the m-dimensional space onto the PCA axes and the species scores are obtained by projecting the unit vectors: for the first species (1,0,0,...); for the second species (0,1,0,0,...), etc., onto the PCA axes (Figure 5.13).



Figure 5.13 Artificial abundance data for two species A and B at 25 sites. a: First principal component, running through the centroid of the sites in the direction of the greatest variance (at 34° of the axis of Species A). b: Rotated version of Figure a with the first principal component horizontally. c: One-dimensional PCA ordination with species represented by arrows. The scores are simply those of the first axis of b.

5.3.4 Biplot of species and site scores

The scores obtained from a PCA for species and sites can be used to prepare a biplot (Gabriel 1971). The biplot serves the same function as the joint plot in CA (Subsection 5.2.5), but the rules to interpret the biplot are rather different. We limit the discussion to the two-dimensional biplot as it is more difficult to visualize three-dimensional or higher ones. The prefix 'bi' in biplot refers to the joint representation of sites and species, and not to the dimension of the plot; for example, Figure 5.13c shows a one-dimensional biplot.

The ranges of the scores for sites and for species (scores and loadings) in PCA are often of a different order of magnitude. For example in Table 5.3c, the range of the species scores is 17.9 whereas the range of the site scores is 0.8. A biplot is therefore constructed most easily by drawing separate plots of sites and of species on transparent paper, each one with its own scaling. In each of the plots, the scale unit along the vertical axis must have the same physical length as the scale unit along the horizontal axis, as in CA. A biplot is obtained by superimposing the plots with the axes aligned. A biplot may therefore have different scale units for the sites (x scale) and species (b scale). Figures 5.12 and 5.15 provide examples for the Dune Meadow Data.

In Subsection 5.3.1, we showed that for each species PCA fits a straight line in one dimension to the (centred) abundances of the species (Figure 5.11; Equation 5.10) and in two dimensions a plane with respect to the PCA axes (Figure 3.11; Equation 5.11). The abundance of a species as fitted by PCA thus changes linearly across the biplot. We represent the fitted planes in a biplot by arrows as shown in Figure 5.12. The direction of the arrow indicates the direction of steepest ascent of the plane, i.e. the direction in which the abundance of the corresponding species increases most, and the length of the arrow equals the rate of change in that direction. In the perpendicular direction, the fitted abundance is constant. The arrows are obtained by drawing lines that join the species points to the origin, the point with coordinates (0,0).

The fitted abundances of a species can be read from the biplot in very much the same way as from a scatter diagram, i.e. by projecting each site onto the axis of the species. (This is clear from Figure 5.13a.) The axis of a species in a biplot is in general, however, not the horizontal axis or the vertical axis, as in Figure 5.13a, but an oblique axis, the direction of which is given by the arrow of the species. As an example of how to interpret Figure 5.12, some of the site points are projected onto the axis of Agrostis stolonifera in Figure 5.14. Without doing any calculations, we can see the ranking of the fitted abundances of A. stolonifera among the sites from the order of the projection points of the sites along the axis of that species. From Figure 5.14, we thus infer that the abundance of A. stolonifera is highest at Site 16, second highest at Site 13, and so on to Site 6, which has the lowest inferred abundance. The inferred ranking is not perfect when compared with the observed ranking, but not bad either.

Another useful rule to interpret a biplot is that the fitted value is positive if the projection point of a site lies, along the species' axis, on the same side of the origin as the species point does, and negative if the origin lies between the



Figure 5.14 Biplot interpretation of Figure 5.12 for Agrostis stolonifera. For explanation, see text.

projection point and the species point. As we have centred the abundance data, the fitted abundance is higher than the species mean in the former case and lower than the species mean in the latter case. For example, Site 3 and Site 20 are inferred to have a higher than average abundance of *A. stolonifera*, whereas Sites 2 and 19 are inferred to have a lower than average abundance of this species. These inferences are correct, as can be seen from Table 5.5. One can also obtain quantitative values for the abundances as represented in the biplot, either algebraically with Equation 5.11 or geometrically as follows (ter Braak 1983). For this, we need the distance of the species point from the origin. In Figure 5.12, we see from the *b* scale that *A. stolonifera* lies at a distance of about 10 from the origin. We need further the projection points of sites onto the species' axis (Figure 5.14). From the *x* scale, we see that, for example, the projection point of Site 20 lies a distance of about 0.2 from the origin. The fitted value is now about $10 \times 0.2 = 2$. Adding the mean of *A. stolonifera* (2.4), we obtain 4.4 as the fitted abundance for A. stolonifera at Site 20; the observed value is 5. This biplot accounts in this way for 51% of the variance in abundance values of all species. This value was computed at the end of Subsection 5.3.2. Note, however, that the fraction of variance accounted for usually differs among species. In general, the abundances of species that are far from the origin are better represented in the biplot than the abundances of species near the origin. For example, the fractions accounted for are 80% for Agrossis stolonifera, 78% for Poa trivialis, 25% for Bromus hordaceus, 4% for Brachythecium rutabulum and 3% for Empetrum nigrum.

The scaling of the species and site scores in the biplot requires attention. From Equation 5.11, we deduce that scaling is rather arbitrary; for example, the fitted values remain the same if we jointly plot the species points $(3b_{k1}, 5b_{k2})$ and the site points $(x_{i1}/3, x_{i2}/5)$. Yet, there are two types of scaling that have special appeal.

In the first type of scaling, the site scores are standardized to unit sum of squares and the species scores are weighted sums of the site scores (Table 5.6). The sum of squared scores of species is then equal to the eigenvalue of the axis. In this scaling, the angle between arrows of each pair of species (Figure 5.12) provides an approximation of their pair-wise correlation, i.e.

$r \approx \cos \theta$

with r the correlation coefficient and θ the angle.

Consequently, arrows that point in the same direction indicate positively correlated species, perpendicular arrows indicate lack of correlation and arrows pointing in the opposite direction indicate negatively correlated species. This biplot is termed the covariance biplot and is considered in detail by Corsten & Gabriel (1976).

In the second type of scaling, the species scores are standardized to unit sum of squares and the site scores are standardized, so that their sum of squares equals the eigenvalue of each axis. Then, the site scores are the weighted sum of the species scores. This scaling was used implicitly in Subsection 5.3.3 and is intended to preserve Euclidean Distances between sites (Equation 5.16), i.e. the length of the line segment joining two sites in the biplot then approximates the length of the line segment joining the sites in m-dimensional space, the axes of which are formed by the species. When scaled in this way, the biplot is termed a Euclidean Distance biplot (ter Braak 1983). Figure 5.15 shows this biplot for the Dune Meadow Data.

The Euclidean Distance biplot is obtained from the covariance biplot by simple rescaling of species and site scores. Species k with coordinates (b_{k1}, b_{k2}) in the covariance biplot gets the coordinates $(b_{k1}/\sqrt{\lambda_1, b_{k2}}/\sqrt{\lambda_2})$ in the Euclidean Distance biplot, and site i with coordinates (x_{i1}, x_{i2}) gets coordinates $(x_{i1}\sqrt{\lambda_1, x_{i2}}\sqrt{\lambda_2})$ in the Euclidean Distance biplot. Figure 5.15 does not look very different from Figure 5.12, because the ratio of $\sqrt{\lambda_1}$ and $\sqrt{\lambda_2}$ is close to 1.



Figure 5.15 Euclidean Distance biplot of Dune Meadow Data.

5.3.5 Data transformation

We have so far described the standard form of PCA as treated in statistical textbooks (e.g. Morrison 1967). In ecology, this form is known as 'species-centred PCA'. In a variant of this, 'standardized PCA', abundances of each species are also divided by its standard deviation. In species-centred PCA, each species is implicitly weighted by the variance of its abundance values. Species with high variance, often the abundant ones, therefore dominate the PCA solution, whereas species with low variance, often the rare ones, have only minor influence on the solution. This may be reason to apply standardized PCA, in which all species receive equal weight. However the rare species then unduly influence the analysis if there are a lot of them, and chance can dominate the results. We therefore recommend species-centred PCA, unless there is strong reason to use standardized PCA. Standardization is necessary if we are analysing variables that are measured in different units, for example quantitative environmental variables such as pH, mass fraction of organic matter or ion concentrations. Noy Meir et al. (1975) fully discuss the virtues and vices of various data transformations in PCA.

The fraction of variance accounted for by the first few axes is not a measure

of the appropriateness of a particular data transformation. By multiplying the abundances of a single species by a million, the first axis of a species-centred PCA will in general account for nearly all the variance, just because nearly all the variance after this transformation is due to this species and the first axis almost perfectly represents its abundances.

If some environmental variables are known to influence the species data strongly, the axes of a PCA will probably show what is already known. To detect unknown variation, one can for each species first apply a regression on the known environmental variables, collect the residuals from these regressions in a twoway table and apply PCA to this table of residuals. This analysis is called partial PCA and is standardly available in the computer program CANOCO (ter Braak 1987b). The analysis is particularly simple if, before sampling, groups of sites are recognized. Then, the deviations of the group means should be analysed instead of the deviations from the general mean. An example is the analysis of vegetation change in permanent plots by Swaine & Greig-Smith (1980).

5.3.6 R-mode and Q-mode algorithms

The iteration algorithm in Table 5.6 is a general-purpose algorithm to extract eigenvectors and eigenvalues from an $m \times n$ matrix Y with elements y_{ki} . The algorithm is used in the computer program CANOCO (ter Braak 1987b) to obtain the solution to species-centred PCA if the rows are centred and to standardized PCA if the rows are standardized, but also to non-centred PCA (Noy Meir 1973) if the data are neither centred nor standardized. However many computer programs for PCA use other algorithms, most of which implicitly transform the data. Centring by variables is done implicitly when PCA is carried out on the matrix of covariances between the variables. Also, standardization by variables is implicit in an analysis of the correlation matrix. The role of species in our discussion therefore corresponds to the role of variables in a general-purpose computer program for PCA. The rest of Subsection 5.3.6 may be skipped at a first reading.

Algorithms that are based on the covariance matrix or correlation matrix are termed R-mode algorithms. More generally, R-mode algorithms extract eigenvectors from the species-by-species cross-product matrix A with elements

$$a_{kl} = \sum_{i} y_{ki} y_{li} \ (k, l = 1, ..., m)$$

where, as before, y_{ki} is the data after transformation.

By contrast, Q-mode algorithms extract eigenvectors from the site-by-site crossproduct matrix C with elements

$$c_{ij} = \Sigma_k y_{ki} y_{kj} (i, j = 1, ..., n).$$

A particular Q-mode algorithm is obtained from Table 5.6 by inserting Equation 5.8 in Equation 5.9. In this way, Steps 2 and 3 are combined into a single step, in which

new $x_i = \sum_{j=1}^n c_{ij} x_j$.

It can be shown that the eigenvalues of the Matrix A equal those of the Matrix C, and further that the eigenvectors of C can be obtained from those of A by applying Equation 5.9 to each eigenvector and, conversely, that the eigenvectors of A can be obtained from those of C by applying Equation 5.8 to each eigenvector of C. The terms R-mode and Q-mode therefore refer to different algorithms and not to different methods. If the number of species is smaller than the number of sites, R-mode algorithms are more efficient than Q-mode algorithms, and conversely.

h

5.4 Interpretation of ordination with external data

Once data on species composition have been summarized in an ordination diagram, the diagram is typically interpreted with help of external knowledge on sites and species. Here we discuss methods that facilitate interpretation when data on environmental variables are collected at different sites. Analogous methods exist when there is external data on the species, for example growth form of plant species or indicator values for environmental variables from previous studies or from the literature (Table 5.7).

Simple interpretative aids include:

- writing the values of an environmental variable in the order of site scores of an ordination axis below the arranged species data table (Table 5.7)
- writing the values of an environmental variable near the site points in the ordination diagram (Figure 5.16)
- plotting the site scores of an ordination axis against the values of an environmental variable (Figure 5.17)
- calculating (rank) correlation coefficients between each of the quantitative environmental variables and each of the ordination axes (Table 5.8)
- calculating mean values and standard deviations of ordination scores for each class of a nominal environmental variable (ANOVA, Subsection 3.2.1) and plotting these in the ordination diagram (Figure 5.16).

An ordination technique that is suited for the species composition data extracts theoretical environmental gradients from these data. We therefore expect straight line (or at least monotonic) relations between ordination axes and quantitative environmental variables that influence species. Correlation coefficients are therefore often adequate summaries of scatter plots of environmental variables against ordination axes.

Three of these simple interpretative aids are directed to the interpretation of axes instead of to the interpretation of the diagram as a whole. But the ordination axes do not have a special meaning. Interpretation of other directions in the diagram is equally valid. A useful idea is to determine the direction in the diagram that has maximum correlation with a particular environmental variable (Dargie 1984). For the *j*th environmental variable, z_j , that direction can be found by multiple (least-squares) regression of z_j on the site scores of the first ordination axis (x_1) and the second ordination axis (x_2) , i.e. by estimating the parameters b_1 and

Table 5.7 Values of environmental variables and Ellenberg's indicator values of species written alongside the ordered data table of the Dune Meadow Data, in which species and sites are arranged in order of their scores on the second DCA axis. A1: thickness of A1 horizon (cm), 9 meaning 9 cm or more; moisture: moistness in five classes from I = dry to 5 = wet; use: type of agricultural use, I = hayfield, 2 = a mixture of pasture and hayfield, 3 = pasture; manure; amount of manure applied in five classes from 0 = no manure to 5 = heavy use of manure. The meadows are classified by type of management; SF, standard farming; BF, biological farming; HF, hobby farming; NM, nature management; F, R, N refer to Ellenberg's indicator values for moisture, acidity and nutrients, respectively.

Species	Site (i)	FRN
k	0000100801011111121 12350749638162485709	
1 Ach mil 11 Ely rep 7 Bro hor 17 Lol per 6 Bel per 18 Poa pra 20 Poa tri 26 Tri pra 23 Rum ace 21 Pot pal 18 Pla Lan 4 Alo gen 8 Che alb 10 Ele pal 27 Tri rep 2 Agr sto 15 Jun buf 30 Cal cus 14 Jun art 9 Cir arv 22 Ran fla 28 Vic Lat 29 Bra rut 16 Leo aut 5 Ant odo 24 Sag pro 25 Sal rep 13 Hun red	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	455570394067892567 37575752277752722567
3 Hyp rad 3 Rir pra 12 Emp nig	2 25 23 2	543 231 64
/ 2 moisture 3 use 4 manure 5 SF 6 BF 7 HF 8 NM	34453344454465353444 11212124155154515255 22211321223332312111 4242134123313200000 101000100100110000000 010010000001000000	·

Table 5.8 Correlation coefficients $(100 \times r)$ of environmental variables with the first four DCA axes for the Dune Meadow Data.

Variable		Axes			
		1	2	3	4
1	Al	58	24	7	9
2	moisture	76	57	7	-7
3	use	35	-21	-3	-5
4	manure	6	-68	-7	-64
5	SF	22	-29	5	-60
6	BF	-28	-24	39	22
7	HF	-22	-26	-55	-14
8	NM	21	73	17	56
Ei	genvalue	0.54	0.29	0.08	0.05



Figure 5.16 The amount of manure written on the DCA ordination of Figure 5.7. The trend in the amount across the diagram is shown by an arrow, obtained by a multiple regression of manure on the site scores of the DCA axes. Also shown are the mean scores for the four types of management, which indicate, for example, that the nature reserves (NM) tend to lie at the top of the diagram.



Figure 5.17 Site scores of the second DCA axis plotted against the amount of manure.

 b_2 of the regression equation (as in Subsection 3.5.2)

$$E_{z_i} = b_0 + b_1 x_1 + b_2 x_2$$
 Equation 5.12

The direction of maximum correlation makes an angle of θ with the first axis where $\theta = \arctan(b_2/b_1)$ and the maximum correlation equals the multiple correlation coefficient (Subsection 3.2.1). This direction can be indicated in the ordination diagram by an arrow running from the centroid of the plot, for instance with coordinates (0,0), to the point with coordinates (b_1, b_2) , as illustrated for manure in Figure 5.16. This is an application of the biplot idea; the environmental variable is represented in the diagram by an arrow that points in the direction of maximum change (Subsection 5.3.4). Several environmental variables can be accommodated in this way in a single ordination diagram.

In Chapter 3, presence and abundance of a single species represented the response variable to be explained by the environmental variables. By applying an ordination technique to the abundances of many species, we have reduced many response variables to a few ordination axes. It is therefore natural to consider the ordination axes as new derived response variables and to attempt to explain each of them by use of multiple regression analysis. For example, we can fit for the first axis (x_1) the response model

$$Ex_1 = c_0 + c_1 z_1 + c_2 z_2 + \dots + c_q z_q$$
 Equation 5.13

where z_j is the *j*th (out of *q*) environmental variables and c_j is the corresponding regression coefficient. The multiple correlation coefficient and the fraction of variance accounted for by the regression (Subsection 3.2.1) indicate whether the environmental variables are sufficient to predict the variation in species composition that is represented by the first ordination axis. Table 5.9 shows an example.

Table 5.9 Multiple regression of the first CA axis on four environmental variables of the dune meadow data, which shows that moisture contributes significantly to the explanation of the first axis, whereas the other variables do not.

Term	Parameter		Estimate	s.e.	t
constant	c _o		-2.32	0.50	-4.62
AI	c ₁		0.14	0.08	1.71
moisture	c_2		0.38	0.09	4.08
use	C3		0.31	0.22	1.37
manure	¢4		0.00	0.12	-0.01
ANOVA table					
	d.f.	s.s.	m.s.	F	
Regression	4	17.0	4.25	10.6	
Residual	15	6.2	0.41		
Total	19	23.2	1.22		
$R^2 = 0.75$					
$R_{\rm adj}^2 = 0.66$					

There are good reasons not to include the environmental variables in the ordination analysis itself, nor to reverse the procedure by applying ordination to the environmental data first and by adding the species data afterwards: the main variation in the environmental data is then sought, and this may well not be the major variation in species composition. For example, if a single environmental variable is important for the species and many more variables are included in the analysis, the first few axes of the environmental ordination mainly represent the relations among the unimportant variables and the relation of the important variable with the species' data would not be discovered. It is therefore better to search for the largest variation in the species' data first and to find out afterwards which of the environmental variables is influential.

5.5 Canonical ordination

5.5.1 Introduction

Suppose we are interested in the effect on species composition of a particular set of environmental variables. What can then be inferred from an indirect gradient analysis (ordination followed by environmental gradient interpretation)? If the ordination of the species data can be readily interpreted with these variables, the environmental variables are apparently sufficient to explain the main variation in the species' composition. But, if the environmental variables cannot explain the main variation, they may still explain some of the remaining variation, which can be substantial, especially in large data sets. For example, a strong relation of the environmental variables with the fifth ordination axis will go unnoticed, when only four ordination axes are extracted, as in some of the computer programs in common use. This limitation can only be overcome by canonical ordination.

Canonical ordination techniques are designed to detect the patterns of variation in the species data that can be explained 'best' by the observed environmental variables. The resulting ordination diagram expresses not only a pattern of variation in species composition but also the main relations between the species and each of the environmental variables. Canonical ordination thus combines aspects of regular ordination with aspects of regression.

We introduce, consecutively, the canonical form of CA, the canonical form of PCA (redundancy analysis) and two other linear canonical techniques, namely canonical correlation analysis and canonical variate analysis. After introducing these particular techniques, we discuss how to interpret canonical ordination axes and the possible effect of data transformations.

5.5.2 Canonical correspondence analysis (CCA)

To introduce canonical correspondence analysis (CCA), we consider again the artificial example by which we have introduced CA (Subsection 5.2.1). In this example (reproduced in Figure 5.18a), five species each preferred a slightly different moisture value. The species score was defined to be the value most preferred and was calculated by averaging the moisture values of the sites in which the species is present. Environmental variables were standardized to mean 0 and variance 1 (Table 5.2c) and the dispersion of the species scores after standardization was taken to express how well a variable explains the species data.

Now suppose, as before, that moisture is the best single variable among the environmental variables measured. In Subsection 5.2.1, we proceeded by constructing the theoretical variable that best explains the species data and, in Section 5.4, we attempted to explain the variable so obtained by a combination of measured environmental variables (Equation 5.13). But, as discussed in Subsection 5.5.1, such attempts may fail, even if we measure environmental variables influencing the species. So why not consider combinations of environmental variables from the beginning? In the example, someone might suggest considering a combination of moisture and phosphate, and Figure 5.18b actually shows that, after standardization, the combination ($3 \times \text{moisture} + 2 \times \text{phosphate}$) gives a larger dispersion than moisture alone. So it can be worthwhile to consider not only the environmental variables singly but also all possible linear combinations of them, i.e. all weighted sums of the form

$$x_i = c_0 + c_1 z_{1i} + c_2 z_{2i} + \dots + c_n z_n$$

where

 z_i is the value of environmental variable j at site i

 c_i is the weight (not necessary positive) belonging to that variable

 x_i is the value of the resulting compound environmental variable at site *i*.



Figure 5.18 Artificial example of unimodal response curves of five species (A-E) with respect to standardized environmental variables showing different degrees of separation of the species curves. a: Moisture. b: Linear combination of moisture and phosphate, chosen a priori. c: Best linear combination of environmental variables, chosen by CCA. Sites are shown as dots, at y = 1 if Species D is present and at y = 0 if Species D is absent.

CCA is now the technique that selects the linear combination of environmental variables that maximizes the dispersion of the species scores (Figure 5.18c; ter Braak 1987a). In other words, CCA chooses the best weights (c_i) for the environmental variables. This gives the first CCA axis.

The second and further CCA axes also select linear combinations of environmental variables that maximize the dispersion of the species scores, but subject to the constraint of being uncorrelated with previous CCA axes (Subsection 5.2.1). As many axes can be extracted as there are environmental variables.

CA also maximizes the dispersion of the species scores, though irrespective of any environmental variable; that is, CA assigns scores (x_i) to sites such that the dispersion is absolutely maximum (Subsection 5.2.1). CCA is therefore 'restricted correspondence analysis' in the sense that the site scores are restricted to be a linear combination of measured environmental variables (Equation 5.14). By incorporating this restriction in the two-way weighted averaging algorithm of CA (Table 5.2), we obtain an algorithm for CCA. More precisely, in each iteration cycle, a multiple regression must be carried out of the site scores obtained in Step 3 on the environmental variables (for technical reasons with y_{+i}/y_{++} as site weights). The fitted values of this regression are by definition a linear combination of the environmental variables (Equation 5.14) and are thus the new site scores to continue with in Step 4 of Table 5.2a. As in CA, the scores stabilize after several iterations and the resulting scores constitute an ordination axis of CCA. The corresponding eigenvalue actually equals the (maximized) dispersion of the species scores along the axis. The eigenvalues in CCA are usually smaller than those in CA because of the restrictions imposed on the site scores in CCA.

The parameters of the final regression in the iteration process are the best weights, also called canonical coefficients, and the multiple correlation of this regression is called the species-environment correlation. This is the correlation between the site scores that are weighted averages of the species scores and the site scores that are a linear combination of the environmental variables. The species-environment correlation is a measure of the association between species and environment, but not an ideal one; axes with small eigenvalues may have misleadingly high species-environment correlations. The importance of the association is expressed better by the eigenvalue because the eigenvalue measures how much variation in the species data is explained by the axis and, hence, by the environmental variables.

CCA is restricted correspondence analysis but the restrictions become less strict the more environmental variables are included in the analysis. If $q \ge n - 1$, then there are actually no restrictions any more; CCA is then simply CA. The arch effect may therefore crop up in CCA, as it does in CA (Gauch 1982). The method of detrending (Hill & Gauch 1980) can be used to remove the arch and is available in the computer program CANOCO (ter Braak 1987b). But in CCA, the arch can be removed more elegantly by dropping superfluous environmental variables. Variables that are highly correlated with the arched axis (often the second axis) are most likely to be superfluous. So a CCA with the superfluous variables excluded does not need detrending.

In Subsection 5.2.7, we saw that CA approximated the maximum likelihood solution of Gaussian ordination when Conditions A1-A4 hold true. If we change the Gaussian ordination model by stating that the site scores must be a linear combination of the environmental variables, the maximum likelihood solution of the model so obtained is again approximated by CCA when these conditions hold true (ter Braak 1986a). The data on species composition are thus explained by CCA through a Gaussian response model in which the explanatory variable is a linear combination of the environmental variables. Furthermore, tests of real data showed that CCA is extremely robust when these assumptions do not hold. The vital assumption is that the response model is unimodal. For a simpler model where relations are monotonic, the results can still be expected to be adequate in a qualitative sense, but for more complex models the method breaks down.

As an example, we use the Dune Meadow Data, which concerns the impact of agricultural use on vegetation in dune meadows on the Island of Terschelling (the Netherlands). The data set consists of 20 relevés, 30 plant species (Table

0.1) and 5 environmental variables (Table 0.2), one of which is the nominal variable 'type of management' consisting of four classes. CCA can accommodate nominal explanatory variables by defining dummy variables as in multiple regression (Subsection 3.5.5). For instance, the dummy variable 'nature management' (Table 5.7) indicates that meadows received that type of management. The first eigenvalue of CCA is somewhat lower than that of CA (0.46 compared to 0.54). Multiple regression of the site scores of the first CA axis on the environmental variables, as we proposed in Section 5.4, resulted in a multiple correlation of 0.87. If the multiple regression is carried out within the iteration algorithm, as in CCA, the multiple correlation increases to 0.96, which is the species-environment correlation. The CCA scores for species and sites look similar to those of CA: not surprisingly, since the multiple correlation obtained with CA is already high. We conclude that, in this example, the measured environmental variables account for the main variation in the species composition. This is true for the second axis also. The second eigenvalue of CCA is 0.29, compared to 0.40 for CA and the second species-environment correlation is 0.89, compared to a multiple correlation of 0.83 in CA. Table 5.10 shows the canonical coefficients that define the first two axes and the correlations of the environmental variables with these axes. These correlations are termed intra-set correlations to distinguish them from the interset correlations, which are the correlations between the environmental variables and the site scores that are derived from the species scores. (The inter-set correlation is R times the intra-set correlation; R is the species-environment correlation of the axis). From the correlations in Table 5.10, we infer that the first axis is a moisture gradient and that the second axis is a manuring axis, separating the meadows managed as a nature reserve from the standardly farmed meadows. This can be seen also from the CCA ordination diagram (Figure 5.19a).

Table 5.10 Canonical correspondence analysis: canonical coefficients $(100 \times c)$ and intra-set correlations $(100 \times r)$ of environmental variables with the first two axes of CCA for the Dune Meadow Data. The environmental variables were standardized first to make the canonical coefficients of different environmental variables comparable. The class SF of the nominal variable 'type of management' was used as reference class in the analysis (Subsection 3.5.5).

Variable	Coeffici	ents	Correlations		
	Axis I	Axis 2	Axis 1	Axis2	
AI	9	-37	57	-17	
moisture	71	-29	93	-14	
use	25	5	21	-41	
manure	-7	-27	-30	-79	
SF	_	-	16	-70	
BF	-9	16	-37	15	
HF	18	19	-36	-12	
NM	20	92	56	76	
		1		<u></u>	
140		7			

The species and sites are positioned as points in the CCA diagram as in CA and their joint interpretation is also as in CA; sites with a high value of a species tend to be close to the point for that species (Subsection 5.2.5). The environmental variables are represented by arrows and can be interpreted in conjunction with the species points as follows. Each arrow determines an axis in the diagram and the species points must be projected onto this axis. As an example, the points of a few species are projected on to the axis for manuring in Figure 5.19b. The order of the projection points now corresponds approximately to the ranking of the weighted averages of the species with respect to amount of manure. The weighted average indicates the 'position' of a species curve along an environmental variable (Figure 5.18a) and thus the projection point of a species also indicates this position, though approximately. Thus Cirsium arvense, Alopecurus geniculatus, Elymus repens and Poa trivialis mainly occur in these data in the highly manured meadows, Agrostis stolonifera and Trifolium repens in moderately manured meadows and Ranunculus flammula and Anthoxanthum odoratum in meadows with little manuring. One can interpret the other arrows in a similar way. From Figure 5.19a, one can see at a glance which species occur mainly in wetter conditions (those on the right of the diagram) and which prefer drier conditions (those on the left of the diagram).

The joint plot of species points and environmental arrows is actually a biplot that approximates the weighted averages of each of the species with respect to each of the environmental variables. The rules for quantitative interpretation of the CCA biplot are the same as for the PCA biplot described in Subsection 5.3.4. In the diagram, the weighted averages are approximated as deviations from the grand mean of each environmental variable; the grand mean is represented by the origin (centroid) of the plot. A second useful rule to interpret the diagram is therefore that the inferred weighted average is higher than average if the projection point lies on the same side of the origin as the head of an arrow and is lower than average if the origin lies between the projection point and the head of an arrow. As in Subsection 5.3.2, a measure of goodness of fit is $(\lambda_1 + \lambda_2)/(sum$ of all eigenvalues), which expresses the fraction of variance of the weighted averages accounted for by the diagram. In the example, Figure 5.19a accounts for 65% of the variance of the weighted averages. (The sum of all canonical eigenvalues is 1.177.)

The positions of the heads of the arrows depend on the eigenvalues and on the intra-set correlations. In Hill's scaling (Subsection 5.2.2), the coordinate of the head of the arrow for an environmental variable on axis s is r_{js} $\sqrt{\lambda_s (1 - \lambda_s)}$, with r_{js} the intra-set correlation of environmental variable j with axis s and λ_s is the eigenvalue of axis s. The construction of biplots for detrended canonical correspondence analysis is described by ter Braak (1986a). Environmental variables with long arrows are more strongly correlated with the ordination axes than those with short arrows, and therefore more closely related to the pattern of variation in species composition shown in the ordination diagram.

Classes of nominal environmental variables can also be represented by arrows (ter Braak 1986a). The projection of a species on such an arrow approximates the fraction of the total abundance of that species that is achieved at sites of



Figure 5.19 CCA of the Dune Meadow Data. a: Ordination diagram with environmental variables represented by arrows. The c scale applies to environmental variables, the u scale to species and sites. The types of management are also shown by closed squares at the centroids of the meadows of the corresponding types of management. b: Inferred ranking of the species along the variable amount of manure, based on the biplot interpretation of Part a of this figure.

that class. However it is sometimes more natural to represent each class of a nominal variable by a point at the centroid (the weighted average) of the sites belonging to that class (Figure 5.19a). Classes consisting of sites with high values for a species are then positioned close to the point of that species. In Figure 5.19a, the meadows managed as a nature reserve are seen to lie at the top-right of the diagram; the meadows of standard farms lie at the bottom.

A second example (from ter Braak 1986a) concerns the presence or absence of 133 macrophytic species in 125 freshwater ditches in the Netherlands. The first four axes of detrended correspondence analyses (DCA) were poorly related (multiple correlation R < 0.60) to the measured environmental variables, which were: electrical conductivity (κ), orthophosphate concentration (PHOSPHATE), both transformed to logarithms, chloride ratio (CHLORIDE, the share of chloride ions in κ) and soil type (clay, peaty soil, sand). By choosing the axes in the





light of these environmental variables by means of CCA, the multiple correlations increased considerably, R being 0.82 and 0.81 for the first two axes. The eigenvalues dropped somewhat - for the first two axes, from 0.34 and 0.25 in DCA to 0.20 and 0.13 in CCA. Apparently, the environmental variables are not sufficient to predict the main variation in species composition extracted by DCA, but they do predict a substantial part of the remaining variation. From the CCA ordination diagram (Figure 5.20), it can be seen that κ and PHOSPHATE are strongly correlated (> 0.8) with the first CCA axis. Species with a high positive score on that axis are therefore almost restricted to ditches with high κ and PHOS-PHATE, and species with a large negative score to ditches with low κ and PHOSPHATE. Species with intermediate scores are either unaffected by κ and PHOSPHATE or restricted to intermediate values of κ and PHOSPHATE. The second CCA axis is strongly correlated (r = 0.9) with CHLORIDE. The arrow for PEAT shows that species whose distribution is the most restricted to peaty soils lie in the top-left corner of the diagram. The arrows for SAND and CLAY are to be interpreted analogously.



Figure 5.20 CCA ordination diagram of the ditch vegetation data (sites are not shown).

5.5.3 Redundancy analysis (RDA)

Redundancy analysis (RDA) is the canonical form of PCA and was invented by Rao (1964). RDA has so far been neglected by ecologists, but appears attractive when used in combination with PCA.

As in PCA (Subsection 5.3.1), we attempt to explain the data of all species by fitting a separate straight line to the data of each species. As a measure of how badly a particular environmental variable explains the species data, we take the total residual sum of squares, as in PCA (Figure 5.11). The best environmental variable is then the one that gives the smallest total residual sum of squares. From this, we can derive a canonical ordination technique, as in Subsection 5.5.2, by considering also linear combinations of environmental variables. RDA is the technique selecting the linear combination of environmental variables that gives the smallest total residual sum of squares.

PCA also minimizes the total residual sum of squares, but it does so without looking at the environmental variables (Subsection 5.3.1). We can obtain the RDA axes by extending the algorithm of PCA (Table 5.6) in a similar fashion to how

we modified the CA algorithm in Subsection 5.5.2; in each iteration cycle, the site scores calculated in Step 3 are regressed on the environmental variables with Equation 5.13 and the fitted values of the regression are taken as the new site scores to continue in Step 4 of the algorithm. (In contrast to CCA, we must now use equal site weights in the regression.) So the site scores are restricted to a linear combination of the environmental variables and RDA is simply PCA with a restriction on the site scores. The species-environment correlation is obtained in the same way as for CCA; but, in RDA, this correlation equals the correlation between the site scores that are weighted sums of the species scores and the site scores that are a linear combination of the environmental variables.

We illustrate RDA with the Dune Meadow Data, using the same environmental variables as in Subsection 5.5.2. The first two axes of PCA explained 29% and 21% of the total variance in the species data, respectively. RDA restricts the axes to linear combinations of the environmental variables and the RDA axes explain therefore less, namely 26% and 17% of the total variance. The first two species-environment correlations are 0.95 and 0.89, both a little higher than the multiple correlations resulting from regressing the first two PCA axes on the environmental variables. We conclude, as with CCA, that the environmental variables account for the main variation in the species composition. From the canonical coefficients and intra-set correlations (Table 5.11), we draw the same conclusions as with CCA, namely that the first axis is mainly a moisture gradient and the second axis a manuring gradient.

The RDA ordination diagram (Figure 5.21) can be interpreted as a biplot (Subsection 5.3.4). The species points and site points jointly approximate the species abundance data as in PCA, and the species points and environmental arrows

Table 5.11 Redundancy analysis: canonical coefficients (100 \times c) and intra-set correlations (100 \times r) of environmental variables with the first two axes of RDA for the Dune Meadow Data. The environmental variables were standardized first to make the canonical coefficients of different environmental variables comparable. The class SF of the nominal variable 'type of management' was used as reference class in the analysis (as in Table 5.10).

Variable	Coeffici	Correlations			
	Axis 1	Axis 2	Axis I	Axis 2	
AI	-1	-5	54	-6	
moisture	15	9	92	12	
use	5	-6	15	29	
manure	-8	16	-26	86	
SF	-	-	25	7 6	
BF	-10	0	-48	-11	
HF	-10	-2	-40	13	
NM	-4	13	51	-79	



Figure 5.21 RDA ordination diagram of the Dune Meadow Data with environmental variables represented by arrows. The scale of the diagram is: 1 unit in the plot corresponds to 1 unit for the sites, to 0.067 units for the species and to 0.4 units for the environmental variables.

jointly approximate the covariances between species and environmental variables. If species are represented by arrows as well (a natural representation in a PCA biplot), the cosine of the angle between the arrows of a species and an environmental variable is an approximation of the correlation coefficient between the species and the environmental variable. One gets a qualitative idea of such correlations from the plot by noting that arrows pointing in roughly the same direction indicate a high positive correlation, that arrows crossing at right angles indicate near-zero correlation, and that arrows pointing in roughly opposite directions indicate a high negative correlation. If arrows are drawn for *Poa trivialis, Elymus repens* and *Cirsium arvense* in Figure 5.21, they make sharp angles with the arrow for manuring; hence, the abundances of these species are inferred to be positively correlated with the amount of manure. We can be more confident about this inference for *Poa trivialis* than for *Cirsium arvense* because the former species lies much further from the centre of the diagram than the latter species. As in

PCA, species at the centre of the diagram are often not very well represented and inferences from the diagram about their abundances and correlations are imprecise. From Figure 5.21, we infer also that, for instance, *Salix repens*, *Hypochaeris radicata* and *Aira praecox* are negatively correlated with the amount of manure.

A measure of goodness of fit of the biplot of species and environmental variables is $(\lambda_1 + \lambda_2)/(\text{sum of all eigenvalues})$, which expresses the fraction of variance of all covariances between species and environment accounted for by the diagram. For the example, Figure 5.21 accounts for 71% of this variance.

The scaling of Figure 5.21 conforms to that of the Euclidean distance biplot (Subsection 5.3.4): the sum of squares of the species scores is unity and site points are obtained by weighted summation of species scores. The positions of the heads of arrows of the environmental variables depend on the intra-set correlations (Table 5.11) and the eigenvalues. With this scaling, the coordinate of the head of the arrow for an environmental variable on axis s must be $r_{js} \sqrt{(\lambda_s/n)}$ where r_{js} is the intra-set correlation of environmental variable j with axis s, n is the number of sites, and λ_s the eigenvalue of axis s. The diagram scaled in this way gives not only a least-squares approximation of the covariances between species and environment, but also approximations of the (centred) abundances values, of the Euclidean Distances among the sites as based on the species data (Equation 5.15), and of covariances among the environmental variables, though the latter two approximations are not least-squares approximations. Other types of scaling are possible (ter Braak 1987b).

5.5.4 Canonical correlation analysis (COR)

The species-environment correlation was a by-product in CCA and RDA, but is central in canonical correlation analysis (COR). The idea of COR is to choose coefficients (scores) for species and coefficients for environmental variables so as to maximize the species-environment correlation. In COR, the species-environment correlation is defined as in RDA, as the correlation between site scores (x_i^*) that are weighted sums of species scores: $(x_i^* = \sum_k b_k y_{ki})$ and site scores (x_i) that are a linear combination of the environmental variables $(x_i = c_0 +$ $\Sigma_i c_i z_i$). An algorithm to obtain the COR axes is given in Table 5.12. The resulting species-environment correlation is termed the canonical correlation, and is actually the squareroot of the first eigenvalue of COR. Step 2 of the algorithm makes the difference from RDA: in RDA, the species scores are simply a weighted sum of the site scores, whereas in COR the species scores are parameters estimated by a multiple regression of the site scores on the species variables. This regression has the practical consequence that, in COR, the number of species must be smaller than the number of sites. It can be shown that the restriction on the number of species is even stronger than that: the number of species plus the number of environmental variables must be smaller than the number of sites. This requirement is not met in our Dune Meadow Data and is generally a nuisance in ecological research. By contrast, RDA and CCA set no upper limit to the number of species that can be analysed. Examples of COR can be found in Gittins Table 5.12 An iteration algorithm for canonical correlation analysis (COR).

Step 1. Start with arbitrary initial site scores (x_i) , not all equal to zero.

- Step 2. Calculate species scores by multiple regression of the site scores on the species variables. The species scores (b_t) are the parameter estimates of this regression.
- Step 3. Calculate new site scores (x_i^*) by weighted summation of the species scores (Equation 5.9). The site scores in fact equal the fitted values of the multiple regression of Step 2.
- Step 4. Calculate coefficients for the environmental variables by multiple regression of the site scores (x_i) on the environmental variables. The coefficients (c_j) are the parameter estimates of this regression.
- Step 5. Calculate new site scores (x_i) by weighted summation of the coefficients of the environmental variables, i.e. by $x_i = \sum_{j=1}^{q} c_j z_{ji}$. The site scores in fact equal the fitted values of the multiple regression of Step 4.
- Step 6. For second and higher axes, orthogonalize the site scores (x_i) as in Table 5.6.
- Step 7. Standardize the site scores (x_i) as in Table 5.6.
- Step 8. Stop on convergence, i.e. when the new site scores are sufficiently close to the site scores of the previous cycle of the iteration process; ELSE go to Step 2.

(1985). COR allows a biplot to be made, from which the approximate covariances between species and environmental variables can be derived in the same way as in RDA (Subsection 5.5.3). The construction of the COR biplot is given in Subsection 5.9.3.

In our introduction to COR, species and environmental variables enter the analysis in a symmetric way (Table 5.12). Tso (1981) presented an asymmetric approach in which the environmental variables explain the species data. In this approach COR is very similar to RDA, but differs from it in the assumptions about the error part of the model (Equations 5.10 and 5.14): uncorrelated errors with equal variance in RDA and correlated normal errors in COR. The residual correlations between errors are therefore additional parameters in COR. When the number of species is large, there are so many of them that they cannot be estimated reliably from data from few sites. This causes practical problems with COR that are absent in RDA and CCA.

5.5.5 Canonical variate analysis (CVA)

Canonical variate analysis (CVA) belongs to the classical linear multivariate techniques along with PCA and COR. CVA is also termed linear discriminant analysis.

If sites are classified into classes or clusters, we may wish to know how the species composition differs among sites of different classes. If we have recorded the abundance values of a single species only, this question reduces to how much the abundance of the species differs between classes, a question studied in Subsection 3.2.1 by analysis of variance. If there are more species, we may wish to combine the abundance values of the species to make the differences between classes clearer than is possible on the basis of the abundance values of a single species. CVA does so by seeking a weighted sum of the species abundances; however not one

148

that maximizes the total variance along the first ordination axis, as PCA does, but one that maximizes the ratio of the between-class sum of squares and the within-class sum of squares of the site scores along the first ordination axis. (These sums of squares are the regression sum of squares and residual sum of squares, respectively, in an ANOVA of the site scores, cf. Subsection 3.2.1.).

Formally, CVA is a special case of COR in which the set of environmental variables consists of a single nominal variable defining the classes. So from Subsection 3.5.5, the algorithm of Table 5.12 can be used to obtain the CVA axes. We deduce that use of CVA makes sense only if the number of sites is much greater than the number of species and the number of classes (Schaafsma & van Vark 1979; Varmuza 1980). Consequently, many ecological data sets cannot be analysed by CVA without dropping many species. Examples of CVA can be found in Green (1979), Pielou (1984) and Gittins (1985).

In contrast to CVA, CCA and RDA can be used to display differences in species composition between classes without having to drop species from the analysis.



Figure 5.22 CCA ordination diagram of the Dune Meadow Data optimally displaying differences in species composition among different types of management.

For this, we must code classes as dummy environmental variables, as in Subsection 3.5.5. Such an analysis by CCA is equivalent to the analysis of concentration proposed by Feoli & Orlóci (1979). As an example, Figure 5.22 displays the differences in vegetation composition between the meadows receiving different types of management in our Dune Meadow Data. The first axis ($\lambda_1 = 0.32$) is seen to separate the meadows receiving nature management from the remaining meadows and second axis ($\lambda_2 = 0.18$) separates the meadows managed by standard farming from those managed by hobby farming and biological farming, although the separations are not perfect. The species displayed on the right side of the diagram occur mainly in the meadows receiving nature management and those on the upper left in the meadows managed by standard farming, and so on. Figure 5.22 displays almost the same information as Figure 5.19a, as can be seen by joining Site Points 16 and 18 in both diagrams. Moisture and manuring are presumably the major factors bringing about vegetation differences between types of management.

5.5.6 Interpreting canonical axes

To interpret the ordination axes, one can use the canonical coefficients and the intra-set correlations. The canonical coefficients define the ordination axes as linear combinations of the environmental variables by means of Equation 5.14 and the intra-set correlations are the correlation coefficients between the environmental variables and these ordination axes. As before, we assume that the environmental variables have been standardized to a mean of 0 and a variance of 1 before the analysis. This standardization removes arbitrariness in the units of measurement of the environmental variables and makes the canonical coefficients comparable among each other, but does not influence other aspects of the analysis.

By looking at the signs and relative magnitudes of the intra-set correlations and of the canonical coefficients standardized in this way, we may infer the relative importance of each environmental variable for prediction of species composition. The canonical coefficients give the same information as the intra-set correlations, if the environmental variables are mutually uncorrelated, but may provide rather different information if the environmental variables are correlated among one another, as they usually are in field data. Both a canonical coefficient and an intra-set correlation relate to the rate of change in species composition by changing the corresponding environmental variable. However it is assumed that other environmental variables are being held constant in the former case, whereas the other environmental variables are assumed to covary with that one environmental variable in the particular way they do in the data set in the latter case. If the environmental variables are strongly correlated with one another, for example simply because the number of environmental variables approaches the number of sites, the effects of different environmental variables on the species composition cannot be singled out and, consequently, the canonical coefficients will be unstable. This is the multicollinearity problem discussed in the context of multiple regression in Subsection 3.5.3. The algorithms to obtain the canonical axes show that canonical coefficients are actually coefficients of a multiple regression (Subsection 5.5.2), so both suffer identical problems. If the multicollinearity problem arises (the program CANOCO (ter Braak 1987b) provides statistics to help detecting it), one should abstain from attempts to interpret the canonical coefficients. But the intra-set correlations do not suffer from this problem. They are more like simple correlation coefficients. They can still be interpreted. One can also remove environmental variables from the analysis, keeping at least one variable per set of strongly correlated environmental variables. Then, the eigenvalues and species-environment correlations will usually only decrease slightly. If the eigenvalues and species-environment correlations drop considerably, one has removed too many (or the wrong) variables.

Their algorithms indicate that COR and CVA are hampered also by strong correlations among species, whereas CCA and RDA are not. So in CCA and RDA, the number of species is allowed to exceed the number of sites.

5.5.7 Data transformation

As in CA and PCA, any kind of transformation of the species abundances may affect the results of CCA and RDA. We refer to Subsections 5.2.2 and 5.3.5 for recommendations about this. The results of COR and CVA are affected by non-linear transformations of the species data, but not by linear transformations. Canonical ordination techniques are not influenced by linear transformations of the environmental variables, but non-linear transformation of environmental variables can be considered if there is some reason to do so. Prior knowledge about the possible impact of the environmental variables on species composition may suggest particular non-linear transformations and particular non-linear combinations, i.e. environmental scalars in the sense of Loucks (1962) and Austin et al. (1984). The use of environmental scalars can also circumvent the multicollinearity problem described in Subsection 5.5.6.

5.6 Multidimensional scaling

In Section 5.1, ordination was defined as a method that arranges site points in the best possible way in a continuum such that points that are close together correspond to sites that are similar in species composition, and points which are far apart correspond to sites that are dissimilar. A particular ordination technique is obtained by further specifying what 'similar' means and what 'best' is. The definition suggests that we choose a measure of (dis)similarity between sites (Subsection 6.2.2), replace the original species composition data by a matrix of dissimilarity values between sites and work further from the dissimilarity matrix to obtain an ordination diagram. This final step is termed multidimensional scaling.

In general, it is not possible to arrange sites such that the mutual distances between the sites in the ordination diagram are equal to the calculated dissimilarity values. Therefore a measure is needed that expresses in a single number how well or how badly the distances in the ordination diagram correspond to the dissimilarity values. Such a measure is termed a loss function or a stress function. In metric ordination techniques such as CA and PCA, the loss function depends on the actual numerical values of the dissimilarities, whereas, in non-metric techniques, the loss function depends only on the rank order of the dissimilarities.

In CA and PCA, one need not calculate a matrix of dissimilarity values first, yet those techniques use particular measures of dissimilarity. In CA, the implied measure of dissimilarity is the chi-squared distance and, in PCA, the Euclidean Distance, as follows immediately from Subsection 5.3.3. The chi-square distance δ_{ii} between site *i* and site *j* is defined as

$$\delta_{ij}^2 = y_{++} \sum_{k=1}^{m} (y_{ki} | y_{+i} - y_{kj} | y_{+j})^2 | y_{k+}$$
 Equation 5.15

and the Euclidean Distance δ_{ij} between these sites is

 $\delta_{ii}^2 = \sum_{k=1}^m (y_{ki} - y_{ki})^2$ Equation 5.16

The chi-squared distance involves proportional differences in abundances of species between sites, whereas the Euclidean Distance involves absolute differences. Differences in site and species totals are therefore less influential in CA than in PCA, unless a data transformation is used in PCA to correct for this effect.

A simple metric technique for multidimensional scaling is principal coordinate analysis (PCO), also called classical scaling (Gower 1966; Pielou 1977, p.290-395). PCO is based on PCA, but is more general than PCA, in that other measures of dissimilarity may be used than Euclidean Distance. In PCO, the dissimilarity values δ_{ii} are transformed into similarity values by the equation

$$c_{ij} = -0.5 \left(\delta_{ij}^2 - \delta_{i+1}^2 / n - \delta_{+j}^2 / n + \delta_{++1}^2 / n^2 \right)$$
 Equation 5.17

where the index + denotes a sum of squared dissimilarities. The matrix with elements c_{ij} is then subjected to the Q-mode algorithm of PCA (Subsection 5.3.6). If the original dissimilarities were computed as Euclidean Distances, PCO is identical to species-centred PCA calculated by the Q-mode algorithm.

In most techniques for (non-metric) multidimensional scaling, we must specify a priori the number of ordination axes and supply an initial ordination of sites. The technique then attempts to modify the ordination iteratively to minimize the stress. In contrast to the iterative algorithms for CA, PCA and PCO, different initial ordinations may lead to different results, because of local minima in the stress function (Subsection 5.2.7); hence, we must supply a 'good' initial ordination or try a series of initial ordinations. From such trials, we then select the ordination with minimum stress.

The best known technique for non-metric multidimensional scaling is ascribed to Shepard (1962) and Kruskal (1964). The stress function, which is minimized in their technique, is based on the Shepard diagram. This is a scatter diagram of the dissimilarities (δ_{ij}) calculated from the species data against the distances d_{ii} between the sites in the ordination diagram.

"The ordination fits perfectly (stress = 0), if the dissimilarities are monotonic with the distances, i.e. if the points in the Shepard plot lie on a monotonically increasing curve. If they do not, we can fit a monotonic curve through the points

by least-squares. This is called monotonic or isotonic regression (Barlow et al. 1972). We then use as stress, a function of the residual sum of squares (for example, Kruskal's stress formula 1, which is the residual sum of squares divided by the total sum of squared distances). The algorithm to seek the ordination that minimizes the stress proceeds further as described above. Note that the method can work equally well with similarities, the only modifications being that a monotonically decreasing curve is fitted in the Shepard diagram. There are two methods to deal with equal dissimilarity values (ties). In the primary approach to ties, the corresponding fitted distances need not be equal, whereas they must be equal in the secondary approach. The primary approach to ties is recommended, because equal dissimilarity values do not necessarily imply equal habitat differences, in part-cular if the equalities arise between pairs of sites that have no species in common (Prentice 1977).

The Shepard-Kruskal method is based on the rank order of all dissimilarities. But calculated dissimilarities may not be comparable in different parts of a gradient, for example if there is a trend in species richness. This potential problem can be overcome by making a separate Shepard diagram for each site, in which we plot the dissimilarities and distances between the particular site and all remaining sites. Each Shepard diagram the distances leads to a stress value and the total stress is taken to be a combination of the separate stress values. This is the local non-metric technique proposed by Sibson (1972). Prentice (1977; 1980) advocated a particular similarity coefficient for use in Sibson's technique. This coefficient is

$$s_{ij} = \Sigma_k \min(y_{ki}, y_{kj})$$

Equation 5.18

Kendall (1971) proved that this coefficient contains all the information required to reconstruct the order of sites when abundances of species follow arbitrary unimodal response curves.

5.7 Evaluation of direct gradient and indirect gradient analysis techniques

Table 5.13 summarizes the techniques described in Chapters 3, 4 and 5 by type of response model and types of variables. We can classify response models as linear and non-linear. Each linear technique (from multiple regression to COR) has non-linear counterparts. A non-linear model that has special relevance in community ecology is the unimodal model. In principle, unimodal models can be fitted to data by the general methods used for non-linear models (in particular by maximum likelihood methods). For regression analysis, these methods are available (GLM, Chapter 3) but, in ordination, they are not so readily available and tend to require excessive computing. Therefore we have also introduced much simpler methods for analysing data for which unimodal models are appropriate. These simple methods start from the idea that the optima of species response curves can be estimated roughly by weighted averaging and we have shown (Section 3.7) that under particular conditions the estimates are actually quite good. This idea resulted in CA, DCA and CCA.

Multidimensional scaling is left out of Table 5.13, because it is unclear what

Table 5.13 Summary of gradient analysis techniques classified by type of response model and types of variables involved. MR, normal multiple regression; IR, inverse regression; PCA, principal components analysis; RDA, redundancy analysis; COR, canonical correlation analysis; WAE, weighted averaging of environmental values; GLM, generalized linear modelling; ML, maximum likelihood; WAI, weighted averaging of indicator values; CA, correspondence analysis; DCA, detrended correspondence analysis; CCA, canonical correspondence analysis; DCA, detrended canonical correspondence analysis; env.vars, environmental variables; comp. gradients, composite gradients of environmental variables, either measured or theoretical.

	Response model		Number of variables				
	linear	unimodal	response (species)	explanatory (env. vars)	composite (comp.gradients)		
Regression Calibration Ordination Canonical ordination	MR IR PCA RDA	WAE, GLM, ML WAI, ML CA, DCA, ML CCA, DCCA, ML	one at a time ≥1* many many	≥l* rarely >1 none many*	one per species none a few for all species a few for all species		
	COR	variants of CCA,ML	many*	many*	a few		

* less than number of sites, except for WAE, WAI and some applications of ML.

response models multidimensional scaling can cope with. Whether (non-metric) multidimensional scaling may detect a particular underlying data structure depends in an unknown way on the chosen dissimilarity coefficient and on the initial ordinations supplied. Non-metric multidimensional scaling could sometimes give better ordinations than DCA does, but the question is whether the improvements are worth the extra effort in computing power and manpower (Clymo 1980; Gauch et al. 1981).

Unimodal models are more general than monotonic ones (Figure 3.3), so it makes sense to start by using unimodal models and to decide afterwards whether one could simplify the model to a monotonic one. Statistical tests can help in this decision (Subsection 3.2.3). In ordination, we might therefore start by using CA, DCA or CCA. This initial analysis will provide a check on how unimodal the data are. If the lengths of the ordination axes are less than about 2 s.d., most of the response curves (or surfaces) will be monotonic, and we can consider using PCA or RDA. The advantage of using PCA and RDA is that in their biplot they provide more quantitative information than CA, DCA and (D)CCA in their joint plot, but this advantage would be outweighed by disadvantages when the data are strongly non-linear (ordination lengths greater than about 4 s.d.).

As illustrated by the Dune Meadow Data whose ordination lengths are about 3 s.d., DCA and PCA may result in similar configurations of site points (Figures 5.7 and 5.15). That they result in dissimilar configurations of species points, even if the ordination lengths are small, is simply due to the difference in meaning

of the species scores in DCA and PCA (Subsections 5.2.5 and 5.3.5).

Table 5.13 also shows the types of variables involved in regression, calibration, ordination and canonical ordination. We distinguish between response variables, explanatory variables and 'composite' variables, which in community ecology typically correspond to species presences or abundances, measured environmental variables and 'composite gradients', respectively. A composite gradient is either a linear combination of measured environmental variables or a theoretical variable. Which technique is the appropriate one to use largely depends on the research purpose and the type of data available. Ordination and cluster analysis (Chapter 6) are the only available techniques when one has no measured environmental data. Calibration must be considered if one wants to make inferences about values of a particular environmental variable from species data and existing knowledge of species-environment relations. Regression and canonical ordination are called for if one wants to build up and extend the knowledge of species-environment relations (Subsections 3.1.1 and 5.1.1).

Whether to use regression or to use canonical ordination depends on whether it is considered advantageous to analyse all species simultaneously or not. In a simultaneous analysis by canonical ordination, one implicitly assumes that all species are reacting to the same composite gradients of environmental variables according to a common response model. The assumption arises because canonical ordination constructs a few composite gradients for all species. By contrast in regression analysis, a separate composite gradient is constructed for each species. Regression may therefore result in more detailed descriptions and more accurate predictions of each particular species, at least if sufficient data are available. However ecological data that are collected over a large range of habitat variation require non-linear models; building good non-linear models by regression is not easy, because it requires construction of composite gradients that are non-linear combinations of environmental variables (Subsection 3.5.4). In CCA, the composite gradients are linear combinations of environmental variables, giving a much simpler analysis, and the non-linearity enters the model through a unimodal model for a few composite gradients, taken care of in CCA by weighted averaging. Canonical ordination is easier to apply and requires less data than regression. It provides a summary of the species-environment relations. The summary may lack the kind of detail that can in principle be provided by regression; on the other hand, the advantages of using regression, with its machinery of statistical tests, may be lost in practice, through the sheer complexity of non-linear model building and through lack of data. Because canonical ordination gives a more global picture than regression, it may be advantageous to apply canonical ordination in the early exploratory phase of the analysis of a particular data set and to apply regression in subsequent phases to selected species and environmental variables.

As already shown in the examples in Subsection 5.5.2, canonical ordination and ordination followed by environmental interpretation can be used fruitfully in combination. If the results do not differ much, then we know that no important environmental variables have been overlooked in the survey. But note that those included could merely be correlated with the functionally important ones. A further proviso is that the number of environmental variables (q) is small compared to the number of sites (n). If this proviso is not met, the species-environment correlation may yield values close to 1, even if none of the environmental variables affects the species. (Note the remarks about R^2 in Subsection 3.2.1.) In particular, canonical ordination and ordination give identical ordination axes if $q \ge n - 1$. If the results of ordination and canonical ordination do differ much, then we may have overlooked major environmental variables, or important non-linear combinations of environmental variables already included in the analysis. But note that the results will also differ if CA or DCA detect a few sites on their first axis that have an aberrant species composition and if these sites are not aberrant in the measured environmental variables. After deleting the aberrant sites, the ordinations provided by (D)CA and CCA may be much more alike.

The question whether we have overlooked major environmental variables can also be studied by combining ordination and canonical ordination in a single analysis. Suppose we believe that two environmental variables govern the species composition in a region. We may then choose two ordination axes as linear combinations of these variables by canonical ordination, and extract further (unrestricted) axes as in CA or PCA, i.e. by the usual iteration process, making the axes unrelated to the previous (canonical) axes in each cycle. The eigenvalues of the extra axes measure residual variation, i.e. variation that cannot be explained by linear combinations of the environmental variables already included in the analysis. Such combined analyses are called partial ordination. Partial PCA (Subsection 5.3.5) is a special case of this.

A further extension of the analytical power of ordination is partial canonical ordination. Suppose the effects of particular environmental variables are to be singled out from 'background' variation imposed by other variables. In an environmental impact study, for example, the effects of impact variables are to be separated from those of other sources of variation, represented by 'covariables'. One may then want to eliminate ('partial out') the effects of the covariables and to relate the residual variation to the impact variables. This is achieved in partial canonical ordination. Technically, partial canonical ordination can be carried out by any computer program for canonical ordination. The usual environmental variables are simply replaced by the residuals obtained by regressing each of the impact variables on the covariables. The theory of partial RDA and partial CCA is described by Davies & Tso (1982) and ter Braak (1988). Partial ordination and partial canonical ordination are available in the computer program CANOCO (ter Braak 1987b). The program also includes a Monte Carlo permutation procedure to investigate the statistical significance of the effects of the impact variables.

5.8 Bibliographic notes

A simple ordination technique of the early days was polar ordination (Bray & Curtis 1957; Gauch 1982), which has been recently reappraised by Beals (1985). PCA was developed early this century by K. Pearson and H. Hotelling (e.g. Mardia et al. 1979) and was introduced in ecology by Goodall (1954). PCA was popularized by Orlóci (1966). CA has been invented independently since 1935 by several authors working with different types of data and with different rationales. Mathematically

CA is the same as reciprocal averaging, canonical analysis of contingency tables, and optimal or dual scaling of nominal variables (Gifi 1981; Gittins 1985; Greenacre 1984; Nishisato 1980). Benzécri et al. (1973) developed CA in a geometric context. Neither of these different approaches to CA is particularly attractive in ecology. Hill (1973) developed an ecological rationale (Subsection 5.2.2). The dispersion of the species scores by which we introduced CA in Subsection 5.2.1 is formally identical to the 'squared correlation ratio' (η^2) used by Torgerson (1958, Section 12.7) and Nishisato (1980, p.23) and also follows from the reciprocal gravity problem in Heiser (1986). RDA is also known under several names (Israëls 1984): PCA of instrumental variables (Rao 1964), PCA of y with respect to x, reduced rank regression (Davies & Tso 1982). Ter Braak (1986a) proposed CCA, COR was derived by H. Hotelling in 1935 (Gittins 1985). Campbell & Atchley (1981) provide a good geometric and algebraic introduction to CVA and Williams (1983) discusses its use in ecology. Methods to obtain the maximum likelihood solutions for Gaussian ordination have been investigated, under the assumption of a normal distribution, a Poisson distribution and a Bernoulli distribution for the species data, by Gauch et al. (1974), Kooijman (1977) and Goodall & Johnson (1982), respectively. However the computational burden of these methods and, hence, the lack of reliable computer programs have so far prevented their use on a routine basis. Ihm & van Groenewoud (1984) and ter Braak (1985) compared Gaussian ordination and CA. Non-metric multidimensional scaling started with the work by Shepard (1962) and Kruskal (1964). Schiffman et al. (1981) provide a clear introduction. They refer to local non-metric scaling as (row) conditional scaling. Meulman & Heiser (1984) describe a canonical form of non-metric multidimensional scaling. Early applications of non-metric multidimensional scaling in ecology were Anderson (1971), Noy-Meir (1974), Austin (1976), Fasham (1977), Clymo (1980) and Prentice (1977; 1980). The simple unfolding model (response models with circular contours) can in principle be fitted by methods of multidimensional scaling (Kruskal & Carroll 1969; Dale 1975; de Sarbo & Rao 1984; Heiser 1987), but Schiffman et al. (1981) warn of practical numerical problems that may reduce the usefulness of this approach. Most of the problems have, however, been circumvented by Heiser (1987).

Many textbooks use matrix algebra to introduce multivariate analysis techniques, because it provides an elegant and concise notation (Gordon 1981; Mardia et al. 1979; Greenacre 1984; Rao 1973; Gittins 1985). For ecologists, the book of Pielou (1984) is particularly recommended. All techniques described in Chapter 5 can be derived from the singular-value decomposition of a matrix, leading to singular vectors and singular values (Section 5.9). The decomposition can be achieved by many numerical methods (e.g. Gourlay & Watson 1973), one of which is the power algorithm (Table 5.6). The power algorithm is used in Chapter 5 because it provides the insight that ordination is simultaneously regression and calibration, and because it does not require advanced mathematics. The power algorithm can easily be programmed on a computer, but is one of the slowest algorithms available to obtain a singular-value decomposition. Hill (1979a) and ter Braak (1987b) use the power algorithm with a device to accelerate the process. The iteration processes of Tables 5.2 and 5.6 are examples of alternating leastsquares methods (Gifi 1981) and are related to the EM algorithm (Everitt 1984). The power algorithm is also a major ingredient of partial least squares (Wold 1982).

Computer programs for PCA, COR and CVA are available in most statistical computer packages. CA and DCA are available in DECORANA (Hill 1979). The program CANOCO (ter Braak 1987b) is an extension of DECORANA and it also includes PCA, PCO, RDA, CCA, CVA and partial variants of these techniques. All these techniques can be specified in terms of matrix algebra (Section 5.9). With the facilities for matrix algebra operations in GENSTAT (Alvey et al. 1977) or SAS (SAS Institute Inc. 1982), one can therefore write one's own programs to analyse small to medium-sized data sets. Schiffman et al. (1981) describe various programs for multidimensional scaling.

Chapter 5 uses response models as a conceptual basis for ordination. Carroll (1972) defined a hierarchy of response models, from the linear model (Equation 5.11), through the model with circular contour lines (Equation 5.5) to the full quadratic model (Equation 3.24) with ellipsoidal contours of varying orientation. He terms these models the vector model, the (simple) unfolding model and the general unfolding model, respectively (also Davison 1983). By taking even more flexible response models, we can define even more general ordination techniques. However the more flexible the model, the greater the computational problems (Prentice 1980). Future research must point out how flexible the model can be to obtain useful practical solutions.

5.9 Ordination methods in terms of matrix algebra

What follows in this section is a short introduction to ordination methods in terms of matrix algebra:

- to facilitate communication between ecologists and the mathematicians they may happen to consult
- to bridge the gap between the approach followed in Chapter 5 and the mainstream of statistical literature on multivariate methods
- to suggest computational methods based on algorithms for singular-value decomposition of a matrix or to extract eigenvalues and eigenvectors from a symmetric matrix.

To start, please read Section 5.8 first.

5.9.1 Principal components analysis (PCA)

Let $Y = \{y_{ki}\}$ be an $m \times n$ matrix containing the data on m species (rows of the matrix) and n sites (columns of the matrix). In the most familiar form of PCA, species-centred PCA, the data are abundances with the species means already subtracted, so that $y_{k+} = 0$ as in Subsection 5.3.1. PCA is equivalent to the singular-value decomposition (SVD) of Y (e.g. Rao 1973; Mardia et al. 1979; Greenacre 1984)

 $\mathbf{Y} = \mathbf{P} \, \Lambda^{0.5} \, \mathbf{Q}'$

Equation 5.19

158

where **P** and **Q** are orthonormal matrices of dimensions $m \times r$ and $n \times r$, respectively, with $r = \min(m,n)$, i.e. **P'P = I** and **Q'Q = I**, and Λ is a diagonal matrix with diagonal elements λ_s ($s = 1 \dots r$), which are arranged in order of decreasing magnitude $\lambda_1 \ge \lambda_2 \ge \lambda_3 \ge \dots \ge 0$.

The columns of **P** and **Q** contain the singular vectors of **Y**, and $\lambda_s^{0.5}$ is the sth singular value of **Y**. If the sth column of **P** is denoted by \mathbf{p}_s , an *m* vector, and the sth column of **Q** by \mathbf{q}_s , an *n* vector, Equation 5.19 can be written as

 $\mathbf{Y} = \boldsymbol{\Sigma}_{s=1}^{\prime} \quad \boldsymbol{\lambda}_{s}^{0.5} \mathbf{p}_{s} \mathbf{q}_{s}^{\prime} \qquad \qquad \text{Equation 5.20}$

The least-squares approximation of Y in Equation 5.11 of Subsection 5.3.2 is obtained from Equation 5.20 by retaining only the first two terms of this summation, and by setting $\mathbf{b}_s = \lambda_s^{0.5} \mathbf{p}_s$ and $\mathbf{x}_s = \mathbf{q}_s$ (s = 1, 2).

The kth element of \mathbf{b}_1 then contains the species score b_{k1} , and the *i*th element of \mathbf{x}_1 contains the site score x_{i1} on the first axis of PCA. Similarly, \mathbf{b}_2 , and \mathbf{x}_2 contain the species and sites scores on the second axis of PCA. The species and sites scores on both axes form the coordinates of the points for species and sites in the biplot (Subsection 5.3.4). The interpretation of the PCA biplot follows from Equation 5.11: inner products between species points and site points provide a least-squares approximation of the elements of the matrix Y (Gabriel 1971; 1978). Equation 5.20 shows that the total sum of squares $\sum_{ki} y_{ki}^2$ equals $\lambda_1 + ... + \lambda_n$, the sum of all eigenvalues, and that the total residual sum of squares

 $\Sigma_{k,i} [y_{ki} - (b_{k1} x_{i1} + b_{k2} x_{i2})]^2 = \lambda_3 + \lambda_4 + \dots + \lambda_r.$

An appropriate measure of goodness of fit is therefore $(\lambda_1 + \lambda_2)/(\text{sum of all eigenvalues})$. From $\mathbf{P'P} = \mathbf{I}$, $\mathbf{Q'Q} = \mathbf{I}$ and Equation 5.20, we obtain

and

 $\mathbf{b}_{s} = \mathbf{Y}\mathbf{x}_{s}$

$$\lambda_s \mathbf{x}_s = \mathbf{Y}' \mathbf{b}_s.$$
 Equation 5.22

Hence, the species scores are a weighted sum of the site scores and the site scores are proportional to a weighted sum of the species scores (Table 5.6 and Subsection 5.3.2). Equation 5.21 and Equation 5.22 show that \mathbf{b}_s and \mathbf{x}_s are eigenvectors of YY' and Y'Y, respectively, and that λ_s is their common eigenvalue; whence, the R-mode and Q-mode algorithms of Subsection 5.3.6.

The SVD of the species-by-species cross-product matrix YY' is $P \wedge P'$, as follows from Equation 5.19 by noting that Q'Q = I. A least-squares approximation of the matrix YY' in two dimensions is therefore given by the matrix $b_1b'_1 + b_2b'_2$. Since YY'/(n-1) contains covariances between species, the biplot of x_s and b_s is termed the covariance biplot (Corsten & Gabriel 1976; ter Braak 1983).

The SVD of the site-by-site cross-product matrix Y'Y is $Q \wedge Q'$. A biplot of Y and Y'Y is therefore obtained by redefining \mathbf{b}_s and \mathbf{x}_s as $\mathbf{b}_s = \mathbf{p}_s$ and $\mathbf{x}_s = \lambda_s^{0.5}\mathbf{q}_s$. The inter-site distances in this biplot approximate the Euclidean Distances between sites as defined by Equation 5.16; hence the name Euclidean Distance biplot. The approximation is, however, indirect namely through Equation 5.17 with c_{ij} the (i, j)th element of Y'Y. A consequence of this is that the intersite distances are always smaller than the Euclidean Distances.

5.9.2 Correspondence analysis (CA)

In CA, the species-by-sites matrix Y contains the abundance values y_{ki} in which $y_{ki} \ge 0$. The data is not previously centred in CA. Let $\mathbf{M} = \text{diag}(y_{k+})$, an $m \times m$ diagonal matrix containing the row totals of Y, $\mathbf{N} = \text{diag}(y_{+i})$, an $n \times n$ diagonal matrix containing the column totals of Y.

As stated in Subsection 5.2.1, CA chooses standardized site scores x that maximize the dispersion of species scores, which are themselves weighted averages of the site scores (Equation 5.1). In matrix notation, the vector of species scores $\mathbf{u} = (u_k)[k = 1, ..., m]$ is

 $u = M^{-1}Yx$ Equation 5.23

and the dispersion is

 $\delta = \mathbf{u}'\mathbf{M}\mathbf{u}/\mathbf{x}'\mathbf{N}\mathbf{x} = \mathbf{x}'\mathbf{Y}'\mathbf{M}^{-1}\mathbf{Y}\mathbf{x}/\mathbf{x}'\mathbf{N}\mathbf{x}$ Equation 5.24

where the denominator takes account of the standardization of x (Table 5.2c), provided x is centred (1'Nx = 0).

The problem of maximizing δ with respect to x has as solution the second eigenvector of the eigenvalue equation

 $Y'M^{-1}Yx = \lambda N x$ Equation 5.25

with $\delta = \lambda$ (Rao 1973, Section 1f.2 and p.74; Mardia et al. 1979, Theorem A9.2).

This can be seen by noting that the first eigenvector is a trivial solution (x = 1; $\lambda = 1$); because the second eigenvector is orthogonal to the first eigenvector in the N metric, the second eigenvector maximizes δ subject to 1'Nx = 0. What is called the first eigenvector of CA in Section 5.2 is thus the second eigenvector of Equation 5.25, i.e. its first non-trivial eigenvector. The second non-trivial eigenvector of Equation 5.25 is similarly seen to maximize δ , subject to being centred and to being orthogonal to the first non-trivial eigenvector, and so on for subsequent axes. Equation 5.25 can be rewritten as

 $\lambda \mathbf{x} = \mathbf{N}^{-1}\mathbf{Y}'\mathbf{u}$

Equation 5.26

160

Equations 5.23 and 5.26 form the 'transition equations' of CA. In words: the species scores are weighted averages of the site scores and the site scores are proportional to weighted averages of the species scores (Table 5.2 and Exercise 5.1.3).

The eigenvectors of CA can also be obtained from the SVD

 $M^{-0.5} Y N^{-0.5} = P \Lambda^{0.5} Q'$ Equation 5.27

by setting $\mathbf{u}_s = \lambda_s^{0.5} \mathbf{M}^{-0.5} \mathbf{p}_s$ and $\mathbf{x}_s = \mathbf{N}^{-0.5} \mathbf{q}_s$, where \mathbf{p}_s and \mathbf{q}_s are the sth columns of **P** and **Q**, respectively (s = 1, ..., r).

This can be seen by inserting the equations for \mathbf{u}_s and \mathbf{x}_s in Equations 5.23 and 5.26, and rearranging terms. It is argued in Subsection 5.2.7 that it is equally valid to distribute λ_s in other ways among \mathbf{u}_s and \mathbf{x}_s , as is done, for example, in Hill's scaling (Subsection 5.2.2).

CA differs from PCA in the particular transformation of Y in Equation 5.27 and in the particular transformation of the singular vectors described just below that equation.

5.9.3 Canonical correlation analysis (COR)

As in species-centred PCA, let Y be an $m \times n$ matrix in which the kth row contains the centred abundance values of the kth species (i.e. $y_{k+} = 0$) and let Z be a $q \times n$ matrix in which the *j*th row contains the centred values of the *j*th environmental variable (i.e. $z_{i+} = 0$). Define

$$S_{12} = YZ', S_{11} = YY', S_{22} = ZZ' \text{ and } S_{21} = S'_{12}.$$
 Equation 5.28

The problem of COR is to determine coefficients for the species $\mathbf{b} = (b_k)[k = 1, ..., m]$ and for the environmental variables $\mathbf{c} = (c_j)[j = 1, ..., q]$ that maximize the correlation between $\mathbf{x}^* = \mathbf{Y}'\mathbf{b}$ and $\mathbf{x} = \mathbf{Z}'\mathbf{c}$. The solution for **b** and **c** is known to be the first eigenvector of the respective eigenvalue equations

 $\mathbf{S}_{12} \, \mathbf{S}_{22}^{-1} \, \mathbf{S}_{21} \, \mathbf{b} = \lambda \, \mathbf{S}_{11} \, \mathbf{b}$ Equation 5.29

 $S_{21} S_{11}^{-1} S_{12} c = \lambda S_{22} c$ Equation 5.30

The eigenvalue λ equals the squared canonical correlation (Rao 1973; Mardia et al. 1979; Gittins 1985).

Note that \mathbf{b} can be derived from a multiple regression of \mathbf{x} on the species, or from \mathbf{c} , by

$$\mathbf{b} = (\mathbf{Y}\mathbf{Y}')^{-1} \mathbf{Y}\mathbf{x} = \mathbf{S}_{11}^{-1} \mathbf{S}_{12} \mathbf{c}$$
 Equation 5.31

and, similarly, c can be derived from a multiple regression of x^* on the environmental variables, or from b, by

 $\lambda c = (ZZ')^{-1} Zx^* = S_{22}^{-1} S_{21} b$

It can be verified that **b** and **c** from Equations 5.31 and 5.32 satisfy Equations 5.29 and 5.30, by inserting **b** from Equation 5.31 into Equation 5.32 and by inserting **c** from Equation 5.32 into Equation 5.31; but note that we could have distributed λ in other ways among Equations 5.31 and 5.32. Equations 5.31 and 5.32 form the basis of the iteration algorithm of Table 5.12. Step 7 of Table 5.12 takes care of the eigenvalue: at convergence, **x** is divided by λ (Table 5.6c). Once convergence is attained, **c** should be divided by λ to ensure that the final site scores satisfy $\mathbf{x} = \mathbf{Z'c}$ (Step 5); hence the λ in Equation 5.32. The second and further axes obtained by Table 5.12 also maximize the correlation between **x** and **x**^{*}, but subject to being uncorrelated to the site scores of the axes already extracted.

COR can also be derived from the SVD of

$$S_{11}^{-0.5} S_{12} S_{22}^{-0.5} = P \Lambda^{0.5} Q'$$
 Equation 5.33

The equivalence of Equation 5.31 with Equation 5.33 can be verified by premultiplying Equation 5.33 on both sides with $S_{11}^{-0.5}$ and post-multiplying Equation 5.33 on both sides with Q and by defining

$$B = S_{11}^{-0.5} P \Lambda^{0.5}$$
 and $C = S_{22}^{-0.5} Q$. Equation 5.34

The sth column of **B** and of C contain the canonical coefficients on the sth axis of the species and environmental variables, respectively. The equivalence of Equation 5.32 with Equation 5.33 can be shown similarly.

COR allows a biplot to be made in which the correlations between species and environmental variables are approximated. The problem to which the canonical correlation biplot is the solution can be formulated as follows: determine points for species and environmental variables in *t*-dimensional space in such a way that their inner products give a weighted least-squares approximation to the elements of the covariance matrix S_{12} . In the approximation, the species and the environmental variables are weighted inversely with their covariance matrices S_{11} and S_{22} , respectively. Let the coordinates of the points for the species be collected in the $m \times t$ matrix G and those for the environmental variables in the $q \times t$ matrix H. The problem is then to minimize

$$\|\mathbf{S}_{11}^{-0.5} (\mathbf{S}_{12} - \mathbf{GH}') \mathbf{S}_{22}^{-0.5} \|^2 = \|\mathbf{S}_{11}^{-0.5} \mathbf{S}_{12} \mathbf{S}_{22}^{-0.5} - (\mathbf{S}_{11}^{-0.5} \mathbf{G})(\mathbf{S}_{22}^{-0.5} \mathbf{H})'\|^2$$

Equation 5.35

with respect to the matrices G and H, where $\|\cdot\|$ is the Euclidean matrix norm, e.g. $\|\mathbf{Y}\|^2 = \sum_{k,i} y_{ki}^2$.

From the properties of an SVD (Subsection 5.9.1), it follows that the minimum is attained when $S_{11}^{0.5}$ G and $S_{22}^{-0.5}$ H correspond to the first *t* columns of the matrices $P\Lambda^{0.5}$ and Q of Equation 5.33, respectively. The required least-squares

approximation is thus obtained by setting G and H equal to the first *t* columns of $S_{11}^{0.5} P \Lambda^{0.5}$ and $S_{22}^{0.5} Q$, respectively. Again, Λ can be distributed in other ways among P and Q. For computational purposes, note that

$$S_{11}^{0.5} P \Lambda^{0.5} = S_{11} S_{11}^{-0.5} P \Lambda^{0.5} = S_{11} B = YY'B = YX$$
 Equation 5.36

and

$$S_{22}^{0.5} Q = S_{22} S_{22}^{-0.5} Q = S_{22} C = ZX$$
 Equation 5.37

where X = Z'C. Because X'X = I, the biplot can thus be constructed from the inter-set correlations of the species and the intra-set correlations of the environmental variables (which are the correlations of the site scores x with the species variables and environmental variables, respectively). This construction rule requires the assumption that the species and environmental variables are standardized to unit variance, so that S_{12} is actually a correlation matrix. The angles between arrows in the biplot are, however, not affected by whether either covariances or correlations between species and environment are approximated in the canonical correlation biplot.

5.9.4 Redundancy analysis (RDA)

RDA is obtained by redefining S_{11} in subsection 5.9.3 to be the identity matrix (Rao 1973, p.594-595). In the RDA biplot, as described in Subsection 5.5.3, the coordinates of the point for the species and the variables are given in the matrices P and $S_{22}^{0.5}$ Q $\Lambda^{0.5}$, respectively.

5.9.5 Canonical correspondence analysis (CCA)

CCA maximizes Equation 5.24 subject to Equation 5.14, provided x is centred. If the matrix Z is extended with a row of ones, Equation 5.14 becomes x = Z'c, with $c = (c_j)[j = 0, 1, ..., q]$. By inserting x = Z'c in Equation 5.24 and (re)defining, with Y non-centred,

$$S_{12} = YZ', S_{11} = M = \text{diag}(y_{k+}) \text{ and } S_{22} = ZNZ'$$
 Equation 5.38

we obtain

$$\delta = c' S_{21} S_{11}^{-1} S_{12} c / c' S_{22} c$$
 Equation 5.39

The solutions of CCA can therefore be derived from the eigenvalue Equation 5.30 with S_{12} , S_{11} and S_{22} defined as in Equation 5.38. If defined in this way, CCA has a trivial solution c' = (1, 0, 0, ..., 0), $\lambda = 1$, x = 1 and the first non-trivial eigenvector maximizes δ subject to 1'Nx = 1'NZ'c = 0 and the maximum δ equals the eigenvalue. A convenient way to exclude the trivial solution is to subtract from each environmental variable its weighted mean $\bar{z}_i = \Sigma_i y_{+i} z_{ji}/y_{++}$

(and to remove the added row of ones in the matrix Z). Then, the matrix Z has weighted row means equal to 0: $\sum_i y_{+i} z_{ji} = 0$. The species scores and the canonical coefficients of the environmental variables can be obtained from Equation 5.33 and Equation 5.34, by using the definitions of Equation 5.38.

As described in Subsection 5.5.2, the solution of CCA can also be obtained by extending the iteration algorithm of Table 5.2. Steps 1, 4, 5 and 6 remain the same as in Table 5.2. In matrix notation, the other steps are

$\operatorname{Step 2} \mathbf{b} = \mathbf{M}^{-1} \mathbf{Y} \mathbf{x}$	Equation 5.40
Step 3a $x^* = N^{-1} Y b$	Equation 5.41
Step 3b c = $(ZNZ')^{\perp}ZNx^*$	Equation 5.42
Step 3c $x = Z'c$	Equation 5.43

with $\mathbf{b} = \mathbf{u}$, the *m* vector containing the species scores u_k (k = 1, ..., m).

Once convergence has been attained, to ensure that the final site scores satisfy x = Z'c, c should be divided by λ , as in COR (below Equation 5.32). This amounts to replacing c in Equation 5.42 by λc (as in Equation 5.32). To show that the algorithm gives a solution of Equation 5.30, we start with Equation 5.42, modified in this way, insert x^* of Equation 5.41 in Equation 5.42, next insert b by using Equation 5.40, next insert x by using Equation 5.43 and finally use the definitions of S_{11} , S_{12} and S_{22} in CCA.

CCA allows a biplot to be made, in which the inner products between points for species and points for environmental variables give a weighted least-squares approximation of the elements of the $m \times q$ matrix

 $\mathbf{W} = \mathbf{M}^{-1}\mathbf{Y}\mathbf{Z}',$

the (k_i) th element of which is the weighted average of species k with respect to the (centred) environmental variable j. In the approximation, the species are given weight proportional to their total abundance (v_{k+}) and the environmental variables are weighted inversely with their covariance matrix S_{22} . The possibility for such a biplot arises because

$$M^{0.5} W S_{22}^{-0.5} = S_{11}^{-0.5} S_{12} S_{22}^{-0.5}$$
 Equation 5.44

so that, from Equations 5.44 and 5.33, after rearranging terms,

 $W = (S_{11}^{-0.5} P) \Lambda^{0.5} (S_{22}^{0.5} Q)'$

Apart from particular considerations of scale (Subsection 5.2.2), the coordinates of the points for species and environmental variables in the CCA biplot are thus

given by the first t columns of $S_{11}^{-0.5} P \Lambda^{0.5}$ and $S_{22}^{0.5} Q$, respectively. The matrix $S_{11}^{-0.5} P \Lambda^{0.5}$ actually contains the species scores, as follows from Equation 5.34. The other matrix required for the biplot can be obtained by

$$S_{22}^{0.5}Q = S_{22} S_{22}^{0.5} Q = S_{22} C = ZNZ'C = ZNX$$
 Equation 5.46

5.10 Exercises

Exercise 5.1 Correspondence analysis: the algorithm

This exercise illustrates the two-way weighted-averaging algorithm of CA (Table 5.2) with the small table of artificial data given below.

Species	Site	es				
-	1	2	3	4	5	
A	1	0	0	1	0	
В	0	0	1	0	1	
С	0	2	0	1	0	
D	3	0	0	1	ì	

The data appear rather chaotic now, but they will show a clear structure after having extracted the first CA ordination axis. The first axis is dealt with in Exercises 5.1.1-3, and the second axis in Exercises 5.1.4-6.

Exercise 5.1.1 Take as site scores the values 1, 2, ..., 5 as shown above the data table. Now, standardize the site scores by using the standardization procedure described in Table 5.2c.

Exercise 5.1.2 Use the site scores so standardized as initial site scores in the iteration process (Table 5.2a). Carry out at least five iteration cycles and in each cycle calculate the dispersion of the species scores. (Use an accuracy of three decimal places in the calculations for the site and species scores and of four decimal places for s.) Note that the scores keep changing from iteration to iteration, but that the rank order of the site scores and of the species scores remains the same from Iteration 4 onwards. Rearrange the species and sites of the table according to their rank order. Note also that the dispersion increases during the iterations.

Exercise 5.1.3 After 19 iterations, the site scores obtained are 0.101, -1.527, 1.998, -0.524, 1.113. Verify these scores for the first CA axis (within an accuracy of two decimal places) by carrying out one extra iteration cycle. What is the eigenvalue of this axis? Verify that Equation 5.1 holds true for the species scores and site scores finally obtained, but that Equation 5.2 does not hold true. Modify Equation 5.2 so that it does hold true.

Exercise 5.1.4 We now derive the second CA axis by using the same initial site scores as in Exercise 5.1.2. Orthogonalize these scores first with respect to the first axis by using the orthogonalization procedure described in Table 5.2b, and next standardize them (round the site scores of the first axis to two decimals and use four decimals for v and s and three for the new scores).

Exercise 5.1.5 Use the site scores so obtained as initial site scores to derive the second axis. The scores stabilize in four iterations (within an accuracy of two decimal places).

Exercise 5.1.6 Construct an ordination diagram of the first two CA axes. The diagram shows one of the major 'faults' in CA. What is this fault?

Exercise 5.2 Adding extra sites and species to a CA ordination

Exercise 5.2.1 We may want to add extra species to an existing CA ordination. In the Dune Meadow Data, *Hippophae rhamnoides* is such a species, occurring at Sites 9, 18 and 19 with abundances 1, 2 and 1, respectively. Calculate from the site scores in Table 5.1c the score for this species on the first CA axis in the way this is done with CA. Plot the abundance of the species against the site score. What does the species score mean in this plot? At which place does the species appear in Table 5.1c? Answer the same questions for *Poa annua*, which occurs at Sites 1, 2, 3, 4, 7, 9, 10, 11, 13 and 18 with abundances 3, 3, 6, 4, 2, 2, 3, 2, 3 and 4, respectively, and for *Ranunculus acris*, which occurs at Sites 5, 6, 7, 9, 14 and 15 with abundances 2, 3, 2, 2, 1 and 1, respectively.

Exercise 5.2.2 Similarly, we may want to add an extra site to an existing CA ordination. Calculate the score of the site where the species *Bellis perennis*, *Poa pratensis* and *Rumex acetosa* are present with abundances 5, 4 and 3, respectively (imaginary data). (Hint: recall how the site scores were obtained from the species scores in Exercise 5.1.3.) Species and sites so added to an ordination are called passive, to distinguish them from the active species and sites of Table 5.1. The scores on higher-order axes are obtained in the same way.

Exercise 5.2.3 Rescale the scores of Table 5.1c to Hill's scaling and verify that the resulting scores were used in Figure 5.4.

Exercise 5.3 Principal components analysis

Add the extra species and the extra site of Exercise 5.2 to the PCA ordination of Table 5.5c. Plot the abundance of the extra species against the site scores. What does the species score mean in this plot? At which places do the species appear in Table 5.5c?

Exercise 5.4 Length of gradient in DCA

Suppose DCA is applied to a table of abundances of species at sites and that the length of the first axis is 1.5 s.d. If, for each species, we made a plot of its abundance against the site scores of the first axis, would most plots suggest monotonic curves or unimodal curves? And what would the plots suggest if the length of the axis was 10 s.d.?

Exercise 5.5 Interpretation of joint plot and biplot

Exercise 5.5.1 Rank the sites in order of abundance of *Juncus bufonius* as inferred from Figure 5.7, as inferred from Figure 5.15 and as observed in Table 5.1a. Do the same for *Eleocharis palustris*.

Exercise 5.5.2 If Figure 5.15 is interpreted erroneously as a joint plot of DCA, one gets different inferred rank orders and, when Figure 5.7 is interpreted erroneously as a biplot, one also gets different rank orders. Is the difference in interpretation greatest for species that lie near the centre of an ordination diagram or for species that lie on the edge of an ordination diagram?

Exercise 5.6 Detrended canonical correspondence analysis

Cramer (1986) studied vegetational succession on the rising sea-shore of an island in the Stockholm Archipelago. In 1978 and 1984, the field layer was sampled on 135 plots of 1 m² along 4 transects. The transects ran from water level into mature forest. One of the questions was whether the vegetational succession keeps track with the land uplift (about 0.5 cm per year) or whether it lags behind. In both cases, the vegetation zones 'run down the shore', but in the latter case too slowly. Because succession in the forest plots was not expected to be due to land uplift, only the 63 plots up to the forest edge were used. These plots contained 68 species with a total of about 1000 occurrences on the two sampling occasions. An attempt was made to answer the question by using detrended canonical correspondence analysis (DCCA) with two explanatory variables, namely altitude above water level in 1984 (not corrected for land uplift; so each plot received the same value in 1978 as in 1984) and time (0 for 1978, 6 for 1984). The altitude ranged from -14 to 56 cm. The first two axes gave eigenvalues 0.56 and 0.10, lengths 4.4 and 0.9 s.d. and species-environment correlations 0.95 and 0.74, respectively. Table 5.14 shows that the first axis is strongly correlated with altitude and almost uncorrelated with time, whereas the second axis is strongly correlated with time and almost uncorrelated with altitude. However the canonical coefficients tell a more interesting story.

Exercise 5.6.1 With Table 5.14, show that the linear combination of altitude and time best separating the species in the sense of Section 5.5.2 is

 $x = 0.054 \ z_1 + 0.041 \ z_2$

Equation 5.47

Table 5.14 Detrended canonical correspondence analysis of rising shore vegetation data: canonical coefficients $(100 \times c)$ and intra-set correlations $(100 \times r)$ for standardized environmental variables. In brackets, the approximate standard errors of the canonical coefficients. Also given are the mean and standard deviation (s.d.) of the variables.

Variable	Coefficie	ents	Correla	tions	mean	s.d.
	Axis 1	Axis 2	Axis I	Axis 2		
Altitude (cm)	100 (3)	4 (4)	99	19	22	18.5
Time (years)	12 (3)	-34 (3)	7	-99	3	2.9

where z_1 is the numeric value of altitude (cm) and z_2 is the numeric value of time (years) and where the intercept is, arbitrary, set to zero.

Hint: note that Table 5.14 shows standardized canonical coefficients, i.e. canonical coefficients corresponding to the standardized variables $z_1^* = (z_1 - 22)/18.5$ and $z_2^* = (z_2 - 3)/2.9$.

Similarly, show that the standard errors of estimate of $c_1 = 0.054$ is 0.0016 and of $c_2 = 0.041$ is 0.010.

Exercise 5.6.2 Each value of x in Equation 5.47 stands for a particular species composition (Figures 5.8 and 5.18) and changes in the value of x express species turnover along the altitude gradient in multiples of s.d. With Equation 5.47, calculate the species turnover between two plots that were 15 cm and 25 cm above water level in 1984, respectively. Does the answer depend on the particular altitudes of these plots or only on the difference in altitude? What is, according to Equation 5.47, the species turnover between these plots in 1978?

Exercise 5.6.3 With Equation 5.47, calculate the species turnover between 1978 and 1984 for a plot with an altitude of 15 cm in 1984? Does the answer depend on altitude?

Exercise 5.6.4 With Equation 5.47, calculate the altitude that gives the same species turnover as one year of succession.

Exercise 5.6.5 Is there evidence that the vegetational succession lags behind uplift?

Exercise 5.6.6 Roughly how long would it take to turn the species composition of the plot closest to the sea into that of the plot that is on the edge of the forest? Hint: use the length of the first axis. Is there evidence from the analysis that there might also be changes in species composition that are unrelated to land uplift? Hint: consider the length of the second axis.

5.11 Solutions to exercises

Exercise 5.1 Correspondence analysis: the algorithm

Exercise 5.1.1 The centroid of the site scores is $z = (4 \times 1 + 2 \times 2 + 1 \times 3 + 3 \times 4 + 2 \times 5)/12 = 2.750$ and their dispersion is $s^2 = [4 \times (1 - 2.750)^2 + ... + 2 \times (5 - 2.750)^2]/12 = 2.353$, thus s = 1.5343. The standardized initial score for the first site is thus $x_1 = (1 - 2.750)/1.5343 = -1.141$. The other scores are listed on the second line of Table 5.15.

Exercise 5.1.2 In the first iteration cycle at Step 2, we obtain for Species C, for example, the score $[2 \times (-0.489) + 1 \times 0.815)]/(2 + 1) = -0.054$, and for Site 5 at Step 3, the score (0.815 - 0.228)/(1 + 1) = 0.294. The dispersion of the species scores in the first iteration cycle is $\delta = (2 \times 0.163^2 + 2 \times 0.815^2 + 3 \times 0.054^2 + 5 \times 0.228^2)/12 = 0.138$. See further Table 5.15. In the iterations shown z = 0.000, apart from Iteration 3, where z = -0.001 (Step 5). The rearranged data table shows a Petrie matrix (Subsection 5.2.3).

Exercise 5.1.3 The standardized site scores obtained in the 19th and 20th iteration are equal within the accuracy of two decimal places; so the iteration process has converged (Table 5.15). The eigenvalue of the first axis is $\lambda_1 = 0.7799$, the value of s calculated last. Equation 5.2 does not hold true for the final site and species scores. But the site scores calculated in Step 3 are weighted averages of the species scores and are divided in the 20th iteration by s = 0.7799 to obtain the final site scores. On convergence, s equals the eigenvalue λ ; thus the final site and species scores satisfy the relation $\lambda x_i = \sum_{k=1}^{m} y_{ki} u_k / \sum_{k=1}^{m} y_{ki}$. Applying Steps 3, (4) and 5 to the eigenvector (the scores x_i) thus transforms the eigenvector into a multiple of itself. The multiple is the 'eigenvalue' of the eigenvector. Note that δ equals λ within arithmetic accuracy.

Exercise 5.1.4 In Step 4.2, we obtain $v = [4 \times (-1.141) \times 0.10 + 2 \times (-0.489) \times (-0.53) + 1 \times 0.163 \times 2.00 + 3 \times 0.815 \times (-0.53) + 2 \times 1.466 \times 1.11]/12 = 0.2771$ and for Site 1 at Step 4.3, the score $-1.141 - 0.277 \times 0.10 = -1.169$. See further the first four lines of Table 5.16.

.

Exercise 5.1.5 See Table 5.16.

Exercise 5.1.6 The configuration of the site points looks like the letter V, with Site 1 at the bottom and Sites 2 and 3 at the two extremities. This is the arch effect of CA (Section 5.2.3).

Exercise 5.2 Adding extra sites and species to a CA ordination

Exercise 5.2.1 In CA, Equation 5.1 is used to obtain species scores from site scores. Thus the score for *Hippophae rhamnoides* is $[1 \times 0.09 + 2 \times (-0.31)$

Table 5.15 Two-way weighted averaging algorithm applied to the data of Exercise 5.1 to obtain the first ordination axis of CA. The initial site scores (Line 1) are first standardized (Line 2). The values in brackets are rank numbers of the scores of the line above. Column 1, iteration number; Column 2, step number in Table 5.2; Column 3, x is site score and u is species score; Column 4, dispersion of the species scores (δ) when preceded by u, or otherwise the square root of the dispersion of the site scores of the line above (s).

Col	นกา	n		Sites					Species			
1	2	3	4	1	2	3	4	5	A	B	С	D
0	1	x		1.000	2.000	3.000	4.000	5.000				
0	5	x	1.5343	-1.141	-0.489	0.163	0.815	1.466				
1	2	u	0.1375						-0.163	0.815	-0.054	-0.228
I	3	x		-0.212	-0.054	0.815	-0.148	0.294	(2)	(4)	(3)	(1)
ł	5	x	0.3012	-0.704	-0.179	2.706	-0.491	0.976				
2	2	u	0.6885						-0.598	1,841	-0.283	-0.325
2	3	x		-0.393	-0.283	1.84 I	-0.402	0.758	(1)	(4)	(3)	(2)
2	5	x	0.6953	-0.567	0.408	2.646	-0.580	1.089				
3	2	u	0.7171						-0.574	1.868	-0.465	-0.238
3	3	х		-0.322	0.465	1.868	-0.426	0.815	(1)	(4)	(2)	(3)
3	5	x	0.7193	-0.448	0.646	2.597	-0.592	1.133				
4	2	u	0.7342						-0.520	1.865	6 -0.628	-0.161
4	3	x		-0.251	-0.628	1.865	-0.436	0.852	(2)	(4)	(1)	(3)
4	5	x	0.7383	-0.340	-0.851	2.526	-0.591	1.154				
5	2	u	0.7498						-0.466	1,840) -0.764	-0.091
5	3	х		-0.185	-0.764	1.840	-0.440	0.875	(2)	(4)	(1)	(3)
5	5	х	0.7529	-0.246	-1.015	2.444	-0.584	1.162				
	•	•	0.7606	(3)	(1)	(5)	(2)	(4)				
	•	,										
	•											
20	2	u	0.7800						-0.211	1.556	5 – 1. 193	0.178
20	3	x		0.081	-1.193	1.556	-0.409	0.867	(2)	(4)	(1)	(3)
20	5	x	0.7799	0.104 (3)	-1.530 (1)	1.995 (5)	-0.524 (2)	1.112 (4)				

 $+ 1 \times (-0.68)]/(1 + 2 + 1) = -0.30$, for *Poa annua* -0.33 and for *Ranunculus acris* -0.19. All three species come in Table 5.1c between *Elymus repens* and *Leontodon autumnalis*. The plots asked for suggest unimodal response curves for *Hippophae rhamnoides* and *Poa annua*, but a bimodal curve for *Ranunculus acris*. The species score is the centroid (centre of gravity) of the site scores in which they occur. The score gives an indication of the optimum of the response curve for the former two species, but has no clear meaning for the latter species. In general, species with a score close to the centre of the ordination may either be unimodal, bimodal or unrelated to the axes (Subsection 5.2.5).

Exercise 5.2.2 The weighted average for the site is $[3 \times (-0.65) + 5 \times (-0.50) + 4 \times (-0.39)]/(3 + 5 + 4) = -0.50$, which must be divided as in Exercise 5.1.3 by λ (= 0.536) to obtain the site score -0.93. If we calculated the score for the

Table 5.16 Two-way weighted averaging algorithm applied to the data of Exercise 5.1 to obtain the second ordination axis of CA. The first line shows the site scores of the first ordination axis (f). The scores on the second line are used as the initial scores after orthogonalizing them with respect to the first axis (Line 3) and standardizing them (Line 4). Column 5 is v, defined in Table 5.2; the other columns are defined in Table 5.15.

Column					Sites					Species			
12		3	4	5	1	2	3	4	5	A	В	с	D
04.	.1	ſ			0,10	-1.53	2.00	-0.53	1.11				
0 4.	1.	x			-1.141	-0.489	0.163	0.815	1.466				
0 4	.3	х		0.2771	-1,169	-0.065	-0.391	0.962	1.158				
0 5	.3	x	0.9612		-1.216	-0.068	-0.407	1.001	1.205				
12		u	0.0837							-0.107	Ø.399	0.288	-0.288
13		x			-0.243	0.288	0.399	~0.036	0.056				
14		x		0.0001	-0.243	0.288	0.399	-0.036	0.056				
15		x	0.2182		-1.[14	1.320	1.829	-0.165	0.257				
22		u	0.5956							-0.639	1.043	0.825	-0.650
23		х			-0.647	0.825	1.043	-0.155	0.197				
24	ŀ	х		-0.0011	-0.647	0.823	1.045	-0.156	0.198				
25	i	х	0.5967		-1.084	1.379	1,751	-0.261	0.332				
32	!	u	0.5980							-0.673	1.042	0.832	-0.636
33	;	x			-0.645	0.832	1.042	-0.159	0.203				
34	ŀ	x		0.0014	-0.645	0.830	1.045	-0.160	0.205				
35	;	х	0.5982		-1.078	1.387	1.747	-0.267	0.343				
4 2	2	u	0.5984							-0.672	1.045	0.836	-0.632
43	3	x			-0.642	0.836	1.045	-0.156	0.206				
44	ţ	х		-0.0016	-0.642	0.834	1.048	-0,157	0.208				
4 5	5	х	0.5985		-1.073	1.393	1.751	-0.262	2 0.348				

second axis by the same method, the extra site would come somewhat below Site 5 in the ordination diagram (Figure 5.4).

Exercise 5.2.3 The site scores of Table 5.1c must be divided by $\sqrt{(1-\lambda)/\lambda} = \sqrt{(0.464/0.536)} = 0.93$ and the species scores by $\sqrt{\lambda(1-\lambda)} = \sqrt{(0.536 \times 0.464)} = 0.50$ (Subsection 5.2.2). For Site 20, for example, we obtain the score 1.95/0.93 = 2.10 and for *Juncus articulatus* 1.28/0.50 = 2.56. In Hill's scaling, the scores satisfy Equation 5.2 whereas Equation 5.1 must be modified analogously to the modification of Equation 5.2 in Exercise 5.1.3

Exercise 5.3 Principal components analysis

The mean abundance of *Hippophae rhamnoides* is 0.2. With Equation 5.8, we obtain the score $(0-0.2) \times (-0.31) + (0 - 0.2) \times (-0.30) + ... + (2-0.2) \times (-0.04) + (1-0.2) \times 0.00 + ... + (0-0.2) \times 0.45 = -0.03$. Similarly we obtain the scores -3.22 for *Poa annua* and -1.48 for *Ranunculus acris*. The plots suggest monotonic decreasing relations for the latter two species, and a unimodal relation

(if any) for the first species. If straight lines are fitted in these plots, the slope of regression turns out to equal the species score (Subsection 5.3.1). The species come at different places in Table 5.5c. For example, *Poa annua* comes just after *Bromus hordaceus*. The score for the extra site is calculated by dividing the weighted sum (Equation 5.9) by the eigenvalue: 3.90/471 = 0.008.

Exercise 5.4 Length of gradient in DCA

In DCA, axes are scaled such that the standard deviation (tolerance) of the response curve of each species is close to one and is on average equal to one. Each response curve will therefore rise and decline over an interval of about 4 s.d. (Figure 3.6; Figure 5.3b). If the length of the first axis equals 1.5 s.d., the length of the axis covers only a small part of the response curve of each species. Most plots will therefore suggest monotonic curves, although the true response curves may be unimodal (Figure 3.3). If the length of the first axis is 10 s.d., the response curves of many species are contained within the length of the axis, so that many of the plots will suggest unimodal response curves.

5.5 Interpretation of joint plot and biplot

Exercise 5.5.1 Inferred rank orders of abundance are for Juncus bufonius

from Figure 5.7 (DCA) sites $12 > 8 > 13 > 9 > 4 \approx 18$ from Figure 5.15 (PCA) sites $13 \approx 3 > 4 > 9 \approx 12$ from Table 5.1a (data) sites 9 = 12 > 13 > 7 - -

and for Eleocharis palustris

from Figure 5.7 (DCA) sites $16 > 14 \approx 15 > 20 > 8$ from Figure 5.15 (PCA) sites 16 > 20 > 15 > 14 > 19from Table 5.1a (data) Sites 16 > 15 > 8 = 14 = 20.

Exercise 5.5.2 The difference in interpretation is greatest for species that lie at the centre of the ordination diagram. In a DCA diagram, the inferred abundance drops with distance from the species point in any direction, whereas in a PCA diagram the inferred abundance decreases or increases with distance from the species point, depending on the direction. This difference is rather unimportant for species that lie on the edge of the diagram, because the site points all lie on one side of the species point. One comes to the same conclusion by noting that a species point in a DCA diagram is its inferred optimum; if the optimum lies far outside the region of the sites the inferred abundance changes monotonically across the region of site points (*Eleocharis palustris* in Figure 5.7).

Exercise 5.6 Detrended canonical correspondence analysis

Exercise 5.6.1 From Table 5.14, we see that the best linear combination is

 $x = 1.00 z_1^{+} + 0.12 z_2^{+}$. In terms of unstandardized variables, we obtain $x = 1.00 \times (z_1 - 22)/18.5 + 0.12 \times (z_2 - 3)/2.9 = (1.00/18.5)z_1 + (0.12/2.9)z_2 - 22/18.5 - 0.12 \times 3/2.9 = 0.054 z_1 + 0.041 z_2 - 1.31$. The standard error of c_1 is 0.03/18.5 = 0.00162 and of c_2 is 0.03/2.9 = 0.010.

Exercise 5.6.2 The value of x for the plot that was 15 cm above water level in 1984 is $x = 0.054 \times 15 + 0.041 \times 6 = 1.056$ s.d. For the plot 25 cm above water level, we obtain $x = 0.054 \times 25 + 0.041 \times 6 = 1.596$ s.d. Hence, the species turnover is 1.596 - 1.056 = 0.54 s.d. According to Equation 5.47, turnover depends only on the difference in altitude between the plots: $0.054 \times (25 - 15) = 0.54$, and does not depend on the particular altitudes of the plots nor on the year of sampling. The species turnover between plots differing 10 cm in altitude is therefore 0.54 s.d. on both occasions of sampling.

Exercise 5.6.3 The value of x for a plot with an altitude of 15 cm in 1984 was 1.056 s.d. in 1984 (Exercise 5.6.2) and was $0.054 \times 15 + 0.041 \times 0 = 0.81$ s.d. in 1978. (Note that in the model altitude was not corrected for uplift; hence $z_1 = 15$ in 1984 and in 1978.) The species turnover is 1.056 - 0.81 = 0.246 s.d., which equals 0.041×6 s.d. and which is independent of altitude. Hence, each plot changes about a quarter standard deviation in 6 years.

Exercise 5.6.4 The species turnover rate is 0.041 s.d. per year, whereas the species turnover due to altitude is 0.054 s.d. per centimetre. The change in altitude that results in 0.041 s.d. species turnover is therefore 0.041/0.054 = 0.76 cm. An approximate 95% confidence interval can be obtained for this ratio from the standard error of c_1 and c_2 and their covariance by using Fieller's theorem (Finney 1964). Here the covariance is about zero. In this way, we so obtained the interval (0.4 cm, 1.1 cm).

Exercise 5.6.5 From Exercise 5.6.4, we would expect each particular species composition to occur next year 0.76 cm lower than its present position. Uplift (about 0.5 cm per year) is less; hence, there is no evidence that the vegetational succession lags behind the land uplift. The known uplift falls within the confidence interval given above. Further, because the value 0 cm lies outside the confidence interval, the effect of uplift on species composition is demonstrated. Uplift apparently drives the vegetational succession without lag.

Exercise 5.6.6 The length of the first axis is 4.4 s.d. From Exercise 5.6.3, we know that each plot changes about 0.25 s.d. in 6 years. The change from vegetation near the sea to vegetation at the edge of the forest therefore takes roughly $(4.4/0.25) \times 6$ years ≈ 100 years. The second axis is 0.9 s.d. and mainly represents the differences in species composition between the two sampling occasions that are unrelated to altitude and land uplift. More precisely, the canonical coefficient of time on the second axis is -0.34/2.9 = -0.117. It therefore accounts for 0.117 $\times 6$ s.d. = 0.70 s.d. of the length of the second axis, whereas time accounted for 0.25 s.d. of the length of the first axis. There are apparently more changes going on than can be accounted for by uplift.

compared in pairs for each criterion on a nine-point scale expressing relative priority. From each matrix of these paired comparisons, weights were extracted for the projects in very much the same way as scores are extracted in ordination. The sub-criteria were also compared in pairs to obtain their relative importance. By applying the same ordination method to the matrices of comparisons, we were also able to derive weights for the sub-critera. Subsequently, the weights of each sub-criterion for a project were combined with the weights of just the sub-criteria to give overall weights for the projects; the weights express the priority of the projects judged (Table 8.4). In this method, a concordance measure is also given (Ancot & van de Nes 1981): a low concordance value indicates that the weight can change if priorities among criteria are changed.

Based on this list of priorities, several brook systems were restored: three of the first group, three of the second group and one of the third group. This was implemented in part by the Polder Board and in part by private initiative. One brook system has been restored, although it was relegated to the third group, indicating that serious problems had to be solved before restoration could be completed. Because diminished extraction of ground water by a paper-mill caused the water-table in this area to rise, the water supply improved so much that it made restoration possible.

The results of this research were also used to help in the decision-making for physical planning on a local scale and to issue permits for water extraction. It is also used as a point of reference to evaluate the management of the brooks.

è

- Agterberg, F.D., 1982. Recent developments in Geomathematics. Geo-Processing 2: 1-32.
- Aitchison. J. & J.A.C. Brown, 1969. The lognormal distribution. Cambridge University Press, Cambridge, 176 pp.
- Alderdice, D.F., 1972. Factor combinations: responses of marine poikilotherms to environmental factors acting in concert. In: O. Kinne (Editor): Marine ecology, vol 1, part 3. Wiley, New York, p. 1659-1722.
- Alvey, N.G. et al., 1977. GENSTAT: a general statistical program. Rothamsted Experimental Station, Harpenden.
- Ancot, J.P. & T. J. van de Nes, 1981. Integral evaluations of alternative water management scenarios in East Gelderland. In: Water resources management on a regional scale. Committee for hydrological research TNO, Proceedings and Information no 27. The Hague, p. 129-164.
- Anderberg, M.R., 1973. Cluster analysis for applications. Academic Press, London, 373 pp.
- Anderson, A.J.B., 1971. Ordination methods in ecology. Journal of Ecology 59:. 713-726.
- Anderson, S., A. Auquier, W.W. Hunck, D. Oakes, W. Vandaele & H.I. Weisberg, 1980. Statistical methods for comparative studies. Wiley, New York, 302 pp.
- Austin, M.P., 1971. Role of regression analysis in plant ecology. Proceedings of the Ecological Society of Australia 6: 63-75.
- Austin, M.P., 1976. Performance of four ordination techniques assuming three different non-linear species response models. Vegetatio 33: 43-49.
- Austin, M.P., 1980. Searching for a model for use in vegetation analysis. Vegetatio 42: 11-21.
- Austin, M.P., R.B. Cunningham & P.M. Fleming, 1984. New approaches to direct gradient analysis using environmental scalars and statistical curve-fitting procedures. Vegetatio 55: 11-27.
- Baker, R.J. & J.A. Nelder, 1978. The GLIM system, Release 3. Numerical Algorithms Group, Oxford.
- Ball, G.H., 1966. A comparison of some cluster seeking techniques. Stanford Research Institute, California.
- Barkman, J.J., H. Doing & S. Segal, 1964. Kritische Bemerkungen und Vorschläge zur quantitativen Vegetationsanalyse. Acta Botanica Neerlandica 13: 394-419.
- Barlow, R.E., D.J. Bartholomew, J.M. Bremner & H.D. Brunk, 1972. Statistical inference under order restrictions. Wiley, New York, 388 pp.

Bartlein, P.J. & T. III Webb, 1985. Mean July temperature at 6000 yr B.T. in

274

Eastern North America: regression equations from fossil-pollen data. Syllogeus 55: 301-342.

- Bartlein, P.J., I.C. Prentice & T. III Webb, 1986. Climatic response surfaces from pollen data for some eastern North American taxa. Journal of Biogeography 13: 35-57.
- Battarbee, R.W., 1984. Diatom analysis and the acidification of lakes. Philosophical Transactions of the Royal Society of London 305: 451-477.
- Batterink M. & G. Wijffels, 1983. Een vergelijkend vegetatiekundig onderzoek naar de typologie en invloeden van het beheer van 1973 tot 1982 in de Duinweilanden op Terschelling. Report Agricultural University, Department of Vegetation Science, Plant Ecology and Weed Science, Wageningen. pp 101.
- Beadle, N.C.W., 1954. Soil phosphate and the delimitation of plant communities in eastern Australia. Ecology 35: 370-375.
- Beals, E.W., 1985. Bray-Curtis ordination: an effective strategy for analysis of multivariate ecological data. Advances in Ecological Research 14: 1-55.

1

- Beckett, P.H.T. & P.A. Burrough, 1971. The relation between cost and utility in soil survey. IV. Comparison of the utilities of soil maps produced by different survey procedures and to different scales. Journal of Soil Science 22: 466-480.
- Becking, R.W. 1957. The Zürich-Montpellier school of Phytosociology. Botanical Review 23: 411-488.
- Belsley, D.A., E. Kuh & R.E. Welsch, 1980. Regression diagnostics. Wiley, New York, 292 pp. ____
- Benzécri, J-P. et al., 1973. L'analyse des données. II: L'analyse des correspondances. Dunod, Paris, 619 pp.
- Bie, S.W. & P.H.T. Beckett, 1973. Comparison of four independent soil surveys by air photo interpretation, Paphos Area (Cyprus). Photogrammetrica 29 (6): 189-202.
- Böcker, R., I. Kowarik & R. Bornkamm, 1983. Untersuchungen zur Anwendung der Zeigerwerte nach Ellenberg. Verhandlungen der Gesellschaft für Ökologie 11: 35-56.
- Box, G.E.P., 1976. Science and statistics. Journal of the American Statistical Association 71: 791-799.
- Box, G.E.P., W.G. Hunter & J.S. Hunter, 1978. Statistics for experimenters. Wiley, New York, 671 pp.
- Braun-Blanquet, J.J., 1964. Pflanzensoziologie, Grundzüge der Vegetationskunde. 3rd Edition. Springer, Vienna, New York, 865 pp.
- Bray, J.R. & J.T. Curtis, 1957. An ordination of the upland forest communities of southern Wisconsin. Ecological Monographs 27: 325-349.
- Brown, G.H., 1979. An optimization criterion for linear inverse estimation. Technometrics 21: 575-579.
- Brown, P.J., 1982. Multivariate calibration. Journal of the Royal Statistical Society, Series B 44: 287-321.
- Buckland, S.T., 1982. Statistics in ornithology. Ibis 124: 61-66.
- Bullock, A. & O. Stallybrass, 1977. Fontana dictionary of modern thought. Fontana Books, London, 694 pp.

- Bunce, R.G.H., C.J. Barr & H. Whittaker, 1983. A stratification system for ecological sampling. In: R.M. Fuller (Editor): Ecological mapping from ground, air and space, I.T.E. symposium 10: 39-46.
- Burrough, P.A., 1983a. Multiscale sources of spatial variation in soil I. The application of fractal concepts to nested levels of soil variation. Journal of Soil Science, 34: 577-597.
- Burrough, P.A., 1983b. Multiscale sources of spatial variation in soil II. A non-Brownian fractal model and its application in soil survey. Journal of Soil Science 34: 599-620.
- Burrough, P.A., 1986. Principles of geographical information systems for natural resource analysis. Oxford University Press, Oxford.
- Burrough, P.A., L. Brown & E.C. Morris, 1977. Variations in vegetation and soil pattern across the Hawkesbury Sandstone Plateau from Barren Grounds to Fitzroy Falls, New South Wales. Australian Journal of Ecology 2: 137-159.
- Campbell, N.A. & W.R. Atchley, 1981. The geometry of canonical variate analysis. Systematic Zoology 30: 268-280.
- Carroll, J.D., 1972. Individual differences and multidimensional scaling. In: R.N. Shepard, A.K. Romney & S.B. Nerlove (Editors): Multidimensional scaling. Theory and application in the behavioral sciences. Vol. 1, Seminar Press, New York, p. 105-155.
- Chatfield, C., 1981. The analysis of time series. Chapman and Hall, London. 268 pp.
- Christiansen, F.B. & T.M. Fenchel, 1977. Theories of populations in biological communities. Springer Verlag, Berlin, 144 pp.
- Chutter, F.M., 1972. An empirical biotic index of the quality of water in South African streams and rivers. Water Research 6: 19-30.
- Clarke, G.M., 1980. Statistics and experimental design. 2nd Edition. Edward Arnold, London, 196 pp.
- Clements, F.E., 1928. Plant succession and indicators. Wilson, New York, 453 pp.
- Clymo, R.S., 1980. Preliminary survey of the peat-bog Hummell Knowe Moss using various numerical methods. Vegetatio 42: 129-148.
- Cochran, W.G., 1963. Sampling techniques. 2nd Edition. Wiley, New York, 430 pp.
- Cochran, W.G., 1977. Sampling techniques. 3rd Edition. Wiley, New York, 444 pp.
- Cochran, W.G., 1983. Planning and analysis of observational studies. Wiley, New York, 158 pp.
- Cochran, W.G. & G.M. Cox, 1957. Experimental Designs. New York, Wiley, 630 pp.
- Coetzee, B.J. & M.J.A. Werger, 1975. On associatian analysis and the classification of plant communities. Vegetatio 30: 201-206
- Cohen, J., 1973. Statistical power and research results. American Educational Research Journal 10: 225-229.
- Cohen, J., 1977. Statistical power analysis for the behavioural sciences. Revised edition. Academic Press, London, 489 pp.
- Cook, R.D. & S. Weisberg, 1982. Residuals and influence in regression. Chapman

277

and Hall, London, 230 pp.

- Corley, M.F.V., A.C. Crundwell, R. Dull, M.O. Hill & A.J.E. Smith, 1981. Mosses of Europe and the Azores; an annotated list of species, with synonyms from the recent literature. Journal of Bryology 11: 609-689.
- Corsten, L.C.A., 1985. Current statistical issues in agricultural research. Statistica Neerlandica 39(2): 159-168.
- Corsten, L.C.A. & K.R. Gabriel, 1976. Graphical exploration in comparing variance matrices. Biometrics 32: 851-863.
- Cox, D.R., 1958. Planning of experiments. Wiley, New York, 315 pp.
- Cox, D.R. & E.J. Snell, 1981. Applied statistics. Chapman and Hall, London, 197 pp.
- Cramer, W, 1986. Vegetation dynamics on rising sea shores in eastern central Sweden. Acta Universitatis Uppsaliensis. Comprehensive summaries of Uppsala dissertations from the Faculty of Science 25. Almqvist & Wiksell International, Stockholm, 34 pp.
- Cuppen, H.P.J.J. & W. Oosterloo, 1980. Een oriënterend onderzoek naar de macrofauna van de Nijmolense beek (Gemeente Epe). Report Regionale Milieuraad Oost Veluwe en Zuiveringsschap Veluwe, Apeldoorn, 50 pp.
- Curtis, J.T. & R.P. McIntosh, 1951. An upland forest continuum in the prairieforest border region of Wisconsin. Ecology 32: 476-496.

Dale, M.B., 1975. On objective methods of ordination. Vegetatio 30: 15-32.

- Dargie, T.C.D., 1984. On the integrated interpretation of indirect site ordinations: a case study using semi-arid vegetation in south-eastern Spain. Vegetatio 55: 37-55.
- David, M., 1977. Geostatistical ore reserve estimation. Developments in Geomathematics 2. Elsevier, Amsterdam, 364 pp.
- Davies, P.T. & M.K-S. Tso, 1982. Procedures for reduced-rank regression. Applied Statistics 31: 244-255.
- Davis, J., 1986. Statistics and Data Analysis in Geology. 2nd Edition. Wiley, New York, 550 pp.
- Davis, R.B. & D.S. Anderson, 1985. Methods of pH calibration of sedimentary diatom remains for reconstructing history of pH in lakes. Hydrobiologia 120: 69-87.
- Davison, M.L., 1983. Multidimensional scaling. Wiley, New York, 242 pp.
- Dawkins, H.C., 1983. Multiple comparisons misused: why so frequently in responsecurve studies? Biometrics 39: 789-790.
- Dent, B.D., 1985. Principles of thematic map design. Addison-Wesley, Reading, Massachusetts.
- De Sarbo, W.S. & V.R. Rao, 1984. GENFOLD2: a set of models and algorithms for the general unfolding analysis of preference/dominance data. Journal of Classification 1: 147-186.
- de Wit, A.E., J. Spuij & P.J.C. van Boheemen, 1984. Slootplanten als indicatoren: een onderzoek naar indicatoren en hun bruikbaarheid in een biologisch meetnet. Landschap 1: 219-227.
- Dixon, W.J., 1981. BMDP statistical software. University of California Press, Berkeley, 726 pp.

- Dobson, A.J., 1983. Introduction to statistical modelling. Chapman and Hall, London, 125 pp.
- Draper, N. & H. Smith, 1981. Applied regression analysis. 2nd Edition. Wiley,. New York, 709 pp.
- Dunn, G. & B.S. Everitt, 1982. An introduction to mathemathical taxonomy. Cambridge University Press, Cambridge.
- Dupont, P., 1975. Les limites altitudinales des landes atlantiques dans les montagnes cantabriques (Nord de l'Espagne). In: J.M. Géhu (Editor): La vegetation des landes d'Europe occidentale Lille 1975, Colloques phytosociologiques 2: 47-58. J. Cramer Vaduz.
- Efron, B., 1982. The jackknife, the bootstrap and other resampling plans. SIAM, Philadelphia, 92 pp.
- Egerton, F.N., 1968. Leeuwenhoek as a founder of animal demography. Journal of Historical Biology 1: 1-22.
- Ellenberg, H., 1948. Unkrautgesellschaften als Mass für den Säuregrad, die Verdichtung und andere Eigenschaften des Ackerbodens. Berichte über Landtechnik, Kuratorium für Technik und Bauwesen in der Landwirtschaft 4: 130-146.
- Ellenberg, H., 1979. Zeigerwerte der Gefässpflanzen Mitteleuropas. Scripta Geobotanica 9, Göttingen, 121 pp.
- Ellenberg, H., 1982. Vegetation Mitteleuropas mit den Alpen in Ökologischer Sicht. 3rd Edition. Ulmer Verlag, Stuttgart, 989 pp.
- Everitt, B.S., 1980. Cluster analysis. 2nd Edition. Heinemann, London, 122 pp.
- Everitt, B.S., 1984. An introduction to latent variable models. Chapman and Hall, London, 107 pp.

Faith, D.P., 1983. Asymmetric binary similarity measures. Oecologia 57: 287-290.

Fasham, M.J.R., 1977. A comparison of nonmetric multidimensional scaling, principal components and reciprocal averaging for the ordination of simulated coenoclines and coenoplanes. Ecology 58: 551-561.

- Feoli, E. & L. Orloci, 1979. Analysis of concentration and detection of underlying factors in structured tables. Vegetatio 40: 49-54.
- Finney, D.J., 1964. Statistical methods in biological assay. Griffin, London, 668 pp.
- Fisher, R.A., 1954. Statistical Methods for Research Workers. 12th Edition. Oliver & Boyd, London, 356 pp.
- Fresco, L.F.M., 1982. An analysis of species response curves and of competition from field data sets: some results from heath vegetation. Vegetatio 48: 175-185.
- Gabriel, K.R., 1971. The biplot graphic display of matrices with application to principal component analysis. Biometrika 58: 453-467.
- Gabriel, K.R., 1978. Least squares approximation of matrices by additive and multiplicative models. Journal of the Royal Statistical Society, Series B 40: 186-196.
- Gauch, H.G., 1979. COMPCLUS A FORTRAN program for rapid initial clustering of large data sets. Cornell University, Ithaca, New York.
- Gauch, H.G., 1982. Multivariate analysis in community ecology. Cambridge

278

University Press, Cambridge, 298 pp.

- Gauch, H.G. & G.B. Chase, 1974. Fitting the Gaussian curve to ecological data. Ecology 55: 1377-1381.
- Gauch, H.G., G.B. Chase & R.H. Whittaker, 1974. Ordinations of vegetation samples by Gaussian species distributions. Ecology 55: 1382-1390.
- Gauch, H.G., R.H. Whittaker & S.B. Singer, 1981. A comparative study of nonmetric ordinations. Journal of Ecology 69: 135-152.
- Gause, G.F., 1930. Studies on the ecology of the Orthoptera. Ecology 11: 307-325.
- Gifi, A., 1981. Nonlinear multivariate analysis. DSWO-press, Leiden, 451 pp.
- Giloi, W.K., 1978. Interactive Computer Graphics Data Structures, algorithms, languages. Prentice Hall, Englewood Cliffs, N.Y., 354 pp.
- Giltrap, D., 1983. Computer Production of soil maps I. Production of grid maps by interpolation. Geoderma 29: 295-311.
- Gittins, R, 1985. Canonical analysis. A review with applications in ecology. Springer-Verlag, Berlin, 351 pp.
- Glacken, C.J., 1967. Traces on the Rhodian shore. Nature and culture in western thought from ancient times to the end of the eighteenth century. University of California Press, Berkely, 762 pp.
- Gleason, H.A., 1926. The individualistic concept of the plant association. Bulletin of the Torrey Botanical Club 53: 7-26.
- Goodall, D.W., 1954. Objective methods for the classification of vegetation. III. An essay in the use of factor analysis. Australian Journal of Botany 1: 39-63.
- Goodall, D.W., 1970. Statistical Plant ecology. Annual review of Ecology and Systematics I: 99-124.
- Goodall, D.W., 1973. Numerical methods of classification. In: R.H. Whittaker (Editor): Handbook of vegetation science, Volume 5: Ordination and Classification of plant communities. Junk, the Hague: 575-615.
- Goodall, D.W. & R.W. Johnson, 1982. Non-linear ordination in several dimensions. A maximum likelihood approach. Vegetatio 48: 197-208.
- Gordon, A.D., 1981. Classification. Methods for the exploratory analysis of multivariate data. Chapman and Hall, London, 193 pp.
- Gounot, M., 1969. Méthodes d'étude quantitative de la végétation. Masson, Paris, 314 pp.
- Gourlay, A.R. & G.A. Watson, 1973. Computational methods for matrix eigen problems. Wiley, New York, 130 pp.
- Gower, J.C., 1966. Some distance properties of latent root and vector methods used in multivariate analysis. Biometrika 53: 325-338.
- Gower, J.C. 1974. Maximal predictive classification. Biometrics 30: 643-654.
- Green, R.H., 1979. Sampling design and statistical methods for environmental biologists. Wiley, New York, 257 pp.
- Greenacre, M.J., 1984. Theory and applications of correspondence analysis. Academic Press, London, 364 pp.
- Greig-Smith, P., 1971. Application of numerical methods to tropical forests. In: G.P. Patil, E.C. Pielou & W.E. Waters (Editors): Statistical ecology. Pennsyl-

vania State University. 3: 195-204.

- Grolle, R., 1976. Verzeichnis der Lebermoose Europas und benachbarter Gebiete. Feddes Repertorium 87: 171-279.
- Guillerm, J.L., 1971. Calcul de l'information fournie par un profil écologique et valeur indicatrice des espèces. Oecologia Plantarum 6: 209-225.
- Hajdu, L.J., 1982. Graphical comparison of resemblance measures ¹¹ phytosociology. Vegetatio 48: 47-59.
- Hardonk, R., 1968. Koornmullenaers, Pampiermaeckers en Coperslaghers. Korte historie van de watermolens van Apeldoorn, Beekbergen en Loenen. Historisch Museum Moerman, Apeldoorn, 256 pp.
- Harrell, F., 1980. The LOGIST procedure. In: SAS supplemental library user's guide, 1980 Edition. SAS Institute Inc., Cary, N.C., p. 83-89.
- Hartigan, J.A., 1975. Clustering algorithms. Wiley, New York, 364 pp.
- Harvey, T., 1981. Environmental intervention: The monitoring paradigm. I. The monitoring concept and practice of descriptive monitoring. The environmentalist 1: 283-291.
- Hawksworth, D.L., P.W. James & B.J. Coppins, 1980. Check-list of Britisch lichenforming, lichenocolous and allied fungi. Lichenologist 12: 1-115.
- Heiser, W.J., 1981. Unfolding analysis of proximity data. Thesis. University of Leiden, Leiden, 273 pp.
- Heiser, W.J., 1986. Undesired nonlinearities in nonlinear multivariate analysis. In: E. Diday et al. (Editors): Data analysis and Informatics 4. North Holland, Amsterdam, p. 455-469.
- Heiser, W.J., 1987. Joint ordination of species and sites: the unfolding technique. In: P. Legendre & L. Legendre (Editors): Developments in numerical ecology. Springer Verlag, Berlin, p. 189-221.
- Hill, M.O., 1973. Reciprocal averaging: an eigenvector method of ordination. Journal of Ecology 61: 237-249.
- Hill, M.O., 1974. Correspondence analysis: a neglected multivariate method. Applied Statistics 23: 340-354.
- Hill, M.O., 1977. Use of simple discriminant functions to classify quantitative phytosociological data. In: E. Diday, L. Lebart, J.P. Pagès & R. Tomassone (Editors): First International Symposium on Data Analysis and Informatics. Vol 1: 181-199. Institut de Recherche d'Informatique et d'Automatique, Le Chesnay, 181-199.
- Hill, M.O., 1979a. DECORANA A FORTRAN program for detrended correspondence analysis and reciprocal averaging. Cornell University Ithaca, N.Y., 52 pp.
- Hill, M.O., 1979b. TWINSPAN A FORTRAN program for arranging multivariate data in an ordered two-way table by classification of individuals and attributes. Cornell University Ithaca, N.Y., 90 pp.
- Hill, M.O., R.G.H. Bunce & M.W. Shaw, 1975. Indicator species analysis, a divisive polythetic method of classification and its application to a survey of native pinewoods in Scotland. Journal of Ecology 63: 597-613.
- Hill, M.O. & H.G. Gauch, 1980. Detrended correspondence analysis, an improved ordination technique. Vegetatio 42: 47-58.

- Hocking, R.R. & O.J. Pendleton, 1983. The regression dilemma. Communications in Statistics Theory and Methods 12: 497-527.
- Hogeweg, P., 1976. Iterative character weighing in numerical taxonomy. Comput. Biol. Med. 6: 199-211.
- Howe, S.E. & T.III Webb, 1983. Calibrating pollen data in climatic terms: improving the methods. Quaternary Science Reviews 2: 17-51.
- Hurlbert, S.H., 1984. Pseudoreplication and the design of ecological field experiments. Ecological Monographs 54: 187-211.
- Ihm, P. & H. van Groenewoud, 1984. Correspondence analysis and Gaussian ordination. COMPSTAT lectures 3: 5-60.
- IJzerman, A.J., 1979. Sprengen en sprengenbeken op de Veluwe. Een onderzoek naar beheer, onderhoud en watervoorziening in historisch perspectief. Rapport ten behoeve van de commissie bestudering waterhuishouding Gelderland, Arnhem, 110 pp.
- Innis, G.S., 1979. Letters to the editor. Bulletin of the Ecological Society of America 60: 142.
- Israëls, A.Z., 1984. Redundancy analysis for qualitative variables. Psychometrika 49: 331-346.
- Jaccard, P., 1912. The distribution of the flora of the alpine zone. New Phytologist 11: 37-50.
- Janssen, J.G.M., 1975. A simple clustering procedure for preliminary classification of very large datasets of phytosociological relevées. Vegetatio 30: 67-71.
- Johnson, S.B. & R.D.-Berger, 1982. On the status of statistics in phytopathology. Phytopathology 72: 1014-1015.
- Jongman, R.H.G. & Th.J. van de Nes (Editors), 1982. Beken op de Veluwe. Een onderzoek naar de mogelijkheden voor herstel en behoud. Begeleidingscommissie Proefgebied Nationaal Landschap Veluwe, Arnhem, 112 pp.
- Journel, A.J. & C.J. Huijbregts, 1978. Mining geostatistics. Academic Press, New York, 600 pp.
- Jowett, G.H. & G. Scurfield, 1949. A statistical investigation into the distribution of Holcus mollis L. and Deschampsia flexuosa L. Trin. Journal of Ecology 37: 68-81.
- Kalkhoven, J. & P. Opdam, 1984. Classification and ordination of breeding bird data and landscape attributes. In: J. Brandt & P. Agger (Editors): Methodology in Landscape ecological research and planning. Universitetsforlag, Roskilde Volume III: 15-26.
- Kanal, L., 1974. Patterns in pattern recognition 1968-74. IEEE Transactions on Information Theory 20: 697-722.
- Kant, G.R., 1982. Beken op de Veluwe. Hydrologische aspecten in relatie tot de watervoering van de beken op de Oost- en Zuid Veluwe. Basisrapport 1 ten behoeve van de werkgroep Sprengen en beken. Provincie Gelderland, Arnhem, 84 pp.
- Kendall, D.G., 1971. Abundance matrices and seriation in archaeology. Zeitschrift für Warscheinlichkeitstheorie 17: 104-112.
- Kilic, M., 1979. Soil variability near Follonica, Italy. MSc. Thesis. MSc Course in Soil Science and Water Management, Agricultural University, Wageningen.

- Kirkpatrick, J.B., P.R. Minchin & J.B. Davies, 1985. Floristic composition and macroenvironmental relationships of Tasmanian vegetation containing bolster plants. Vegetatio 63: 89-96.
- Kooijman, S.A.L.M., 1977. Species abundance with optimum relations to en- vironmental factors. Annals of System Research 6: 123-138.
- Krebs, C.J., 1978. Ecology, the experimental analysis of distribution and abundance. Harper and Row, New York, 648 pp.
- Kruijne, A.A., D.M. de Vries & H. Mooi, 1967. Bijdrage tot de oecologie van de Nederlandse graslandplant (with english summary). Verslagen Landbouwkundige Onderzoekingen 696. Pudoc, Wageningen, 65 pp.
- Kruskal, J.B., 1964. Multidimensional scaling by optimizing goodness of fit to a nonmetric hypothesis. Psychometrika 29: 1-27.
- Kruskal, J.B. & J.D. Carroll, 1969. Geometric models and badness-of-fit functions.
 In: P.R. Krishnaiah (Editor): Multivariate analysis II. Academic Press, New York, p. 639-671.
- Laboratory for Computer Graphics, 1968. SYMAP Manual, Graduate School of Design, Harvard.
- Lachenbruch, P.A., 1975. Discriminant analysis. Hafner Press, New York, 128 pp.
- Lam, N.S., 1983. Spatial interpolation methods: a review. The American Cartographer 10: 129-149.
- Lambert, J.M. & M.B.Dale, 1964. The use of statistics in phytosociology. Advances in Ecological Research 2: 59-99.
- Lambert, J.M., S.E. Meacock, J. Barrs & P.F.M. Smartt, 1973. AXOR and MONIT: two new polythetic-divisive strategies for hierarchial classification. Taxon 22: 1733-176.
- Lance, G.N. & W.T. Williams, 1967. A general theory of classificatory sorting strategies. I Hierarchical systems. Computer Journal 9: 373-380.
- Loucks, O.L., 1962. Ordinating forest communities by means of environmental scalars and phytosociological indices. Ecological Monographs 32: 137-166.
- Louppen, J.W.M. & E. van der Maarel, 1979. CLUSLA: a computer program for the clustering of large phytosociological data sets. Vegetatio 40: 107-114.
- Lozano, J.A. & J.D. Hays, 1976. Relationship of Radiolarian assemblages to sediment types and physical oceanography in the Atlantic and Western Indian Ocean. In: R.M. Cline & J.D. Hays (Editors): Investigation of late Quaternary paleoceanography and paleoclimatology. Geological Society of America Memoir 145: 303-336.
- Mandelbrot, B.B., 1982. The fractal geometry of nature. Freeman, San Francisco, 460 pp.
- Mardia, K.V., J.T. Kent & J.M. Bibby, 1979. Multivariate analysis. Academic Press, London, 521 pp.
- Matheron, G., 1971. The theory of regionalized variables and its applications. Les Cahiers du Centre de Morphologie Mathematique de Fontainebleu. Ecole Nationale Superieure des Mines de Paris.
- McBratney, A.B., R. Webster, R.G. McLaren & R.B. Spiers, 1982. Regional variation of extractable copper and cobalt in the topsoil of south-east Scotland.

Agronomie 2 (10): 969-982.

- McCullagh, P. & J.A. Nelder, 1983. Generalized linear models. Chapman and Hall, London, 261 pp.
- McIntosh, R.P., 1981. Succession and ecological theory. In: D.C. West, H.H. Shugart & D.B. Botkin (Editors): Forest succession. Concepts and application. Springer Verlag, New York, p. 10-23.
- Mead, R. & D.J.P. Pike, 1975. A review of response surface methodology from a biometric viewpoint. Biometrics 31: 803-851.
- Meijers, E., 1986. Defining confusions-confusing definitions. Environmental monitoring and assessment 7: 157-159.
- Meijers, E., W.J. ter Keurs & E. Meelis, 1982. Biologische meetnetten voor het beleid. Mededelingen van de Werkgemeenschap Landschapsecologisch Onderzoek 9: 51-58.
- Mertz, D.B. & D.E. McCauley, 1980. The domain of laboratory ecology. Synthese 43: 95-110.
- Meulman, J. & W.J. Heiser, 1984. Constrained multidimensional scaling: more directions than dimensions. COMPSTAT 1984, Physica-Verlag, Vienna, p. 137-142.
- Minchin, P.R., 1987. An evaluation of the relative robustness of techniques for ecological ordination. Vegetatio 67: 1167-1179.
- Montgomery, D.C. & E.A. Peck, 1982. Introduction to linear regression analysis. Wiley, New York, 504 pp.
- Morrison, D.F., 1976.-Multivariate statistical methods. 2nd Edition. McGraw-Hill, Tokyo, 415 pp.
- Mosteller, F. & J.W. Tukey, 1977. Data analysis and regression. Addison-Wesley, Reading, 588 pp.
- Mueller-Dombois, D. & H. Ellenberg, 1974. Aims and methods of vegetation ecology. Wiley, New York, 547 pp.
- Naes, T. & H. Martens, 1984. Multivariate calibration. II Chemometric methods. Trends in analytical chemistry 3: 266-271.
- Neef, E., 1982. Stages in development of landscape ecology. In: S. Tjallingi & A. de Veer (Editors): Perspectives in Landscape Ecology. Proceedings of the International symposium of Landscape ecology at Veldhoven Netherlands. Pudoc, Wageningen, p. 19-27.
- Nelder, J.A. & R.W.M. Wedderburn, 1972. Generalized linear models. Journal of the Royal Statistical Society, Series A 135: 370-384.
- Nielsen, D.R. & J. Bouma (Editors), 1985. Soil spatial variability. Proceedings of a workshop of the ISSS and the SSSA, Las Vegas, USA, 30 Nov. - 1 Dec. 1984. Pudoc, Wageningen, 243 pp.
- Nishisato, S., 1980. Analysis of categorical data: dual scaling and its applications. Toronto University Press, Toronto, 276 pp.
- Noy-Meir, I., 1973. Data transformation in ecological ordination. I. Some advantages of non-centering. Journal of Ecology 61: 329-341.
- Noy-Meir, I., 1974. Catenation: quantitative methods for the definition of coenoclines. Vegetatio 29: 89-99.
- Noy-Meir, I., D. Walker & W.T. Williams, 1975. Data transformations in ecological

- ordinátion. II. On the meaning of data standardization. Journal of Ecology 63: 779-800.
- Odum, E.P., 1971. Fundamentals of ecology. 3rd Edition. Saunders, Philadelphia.
- Orlóci, L., 1966. Geometric models in ecology. I. The theory and application of some ordination methods. Journal of Ecology 54: 193-215.
- Orlóci, L., 1967. An agglomerative method for the classification of plant communities. Journal of Ecology 55: 193-206.
- Parker, R.E., 1979. Introductory statistics for biology. Studies in biology, No 43. Arnold, London, 122 pp.
- Pavlidis, T., 1982. Algorithms for graphics and image processing. Springer Verlag, Berlin, 416 pp.
- Pielou, E.C., 1969. An introduction to mathematical ecology. Wiley, New York, 286 pp.
- Pielou, E.C., 1976. Population and community ecology. Gordon and Breach, New York, 432 pp.
- Pielou, E.C., 1977. Mathematical ecology. 2nd Edition. Wiley, New York, 385 pp.
- Pielou, E.C., 1984. The interpretation of ecological data. A primer on classification and ordination. Wiley, New York, 263 pp.
- Popma, J., 1982. Beken op de Veluwe. Numerieke classificatie van macrofauna gemeenschappen van sprengen en beken op de Veluwe. Basisrapport 2 ten behoeve van de werkgroep sprengen en beken. Provincie Gelderland, Arnhem, 132 pp.
- Popma, J., L. Mucina, O. van Tongeren & E. van der Maarel, 1983. On the determination of optimal levels in phytosociological classification. Vegetatio 52: 65-76.
- Prentice, I.C., 1977. Non-metric ordination methods in ecology. Journal of Ecology 65: 85-94.
- Prentice, I.C., 1980. Vegetation analysis and order invariant gradient models. Vegetatio 42: 27-34.
- Quinn, J.F. & A.E. Dunham, 1983. On hypothesis testing in ecology and evolution. The American Naturalist 122: 602-617.
- Ramensky, L.G., 1930. Zur Methodik der vergleichenden Bearbeitung und Ordnung von Pflanzenlisten und anderen Objekten, die durch mehrere, verschiedenartig wirkende Faktoren bestimt werden. Beiträge zur Biologie der Pflanzen 18: 269-304.
- Rao, C.R., 1964. The use and interpretation of principal component analysis in applied research. Sankhya A 26: 329-358.
- Rao, C.R., 1973. Linear statistical inference and its applications. 2nd Edition. Wiley, New York, 625 pp.
- Ripley, B., 1981. Spatial statistics. Wiley, New York, 252 pp.

Risser, P.G., J.R. Karr & R.T.T. Forman, 1984. Landscape ecology directions and approaches. A workshop held at Allerton Park, Piatt. County Illinois 1983. Illinois Natural History Survey Special Publication Number 2, 16 pp.

Rivas-Martínez, S., 1979. Brezales y jarales de Europa occidental. Lazaroa 1: 5-127.

285

- Rogister, J.E., 1978. De ecologische mR- en mN-waarden van de kruidlaag en de humuskwaliteit van bosplantengezelschappen. Werken Proefstation Waters en Bossen (Groenendaal- Hoeilaart) reeks A 20: 1-29.
- Roux, G. & M. Roux, 1967. A propos de quelques méthodes de classification en phytosociologie. Revue de statistique appliquée 15: 59-72.
- Saaty, T.L., 1977. A scaling method for priorities in hierarchical structures. Journal of mathematical psychology 14: 234-281.
- Salton, G. & A. Wong, 1978. Generation and search of cluster files. Association for computing machinery transactions on database systems 3: 321-346.
- SAS Institute Inc., 1982. SAS user's guide: statistics, 1982 Edition. SAS Institute Inc., Cary, N.C., 584 pp.
- Schaafsma, W. & G.N. van Vark, 1979. Classification and discrimination problems with application, part IIa. Statistica Neerlandica 33: 91-126.
- Scheffé, H., 1973. A statistical theory of calibration. Annals of Statistics 1: 1-37.
- Schiffman, S.S., M.L. Reynolds & F.W. Young, 1981. Introduction to multidimensional scaling. Theory, methods and applications. Academic Press, London, 413 pp.
- Schwartz, L.M., 1977. Nonlinear calibration. Analytical chemistry 49: 2062-2068.
- Seber, G.A.F., 1977. Linear regression analysis. Wiley, New York, 465 pp.
- Sheenan, P.J., 1984. Effects on community and ecosystem structure and dynamics. In: P.J. Sheenan, D.R. Miller, G.C. Butler & Ph. Bourdeau (Editors): Effects of pollutants at the ecosystem level. Wiley, New York, p. 51-99.
- Shepard, R.N., 1962. The analysis of proximities: multidimensional scaling with an unknown distance function. II. Psychometrika 27: 219-246.
- Shepard, D., 1968. A two-dimensional interpolation function for irregularly spaced data. In: Proc. Ass. Comput. Mach. 517-523.
- Sibson, R., 1972. Order invariant methods for data analysis. Journal of the Royal Statistical Society, Series B 34: 311-349.
- Simberloff, D., 1983. Competition theory, hypothesis-testing, and other community ecological buzzwords. The American Naturalist 122: 626-635.
- Slob, W., 1987. Strategies in applying statistics in ecological research. Free University Press, Amsterdam, 112 pp.
- Sneath, P.H.A. & Sokal, R.R., 1973. Numerical taxonomy. W.H. Freeman, San Fransisco.
- Snedecor, G.W. & W.G. Cochran, 1980. Statistical methods. 7th Edition. The Iowa State University Press, Ames, 507 pp.
- Sokal, R.R. & C.D. Michener, 1958. A statistical method for evaluating systematic relationships. University of Kansas Science bulletin 38: 1409-1438.
- Sokal, R.R. & F.J. Rohlf, 1981. Biometry. 2nd Edition. Freeman, San Fransisco, 877 pp.
- Sørensen, T., 1948. A method of establishing groups of equal amplitude in plant sociology based on similarity of species content. Det. Kong. Danske Vidensk. Selsk. Biol. Skr. (Copenhagen) 5 (4): 1-34.
- Southwood, T.R.E., 1978. Ecological methods, with particular reference to the study of insect populations. 2nd Revised edition. Methuen, London, 391 pp.

Strong jr., D.R., 1980. Null hypotheses in ecology. Synthese 43: 271-285.

- Swain, P.H., 1978. Fundamentals of pattern recognition in remote sensing. In: P.H. Swain & S.M. Davies (Editors): Remote sensing: the quantitative approach. McGraw-Hill, New York.
- Swaine, M.D. & P. Greig-Smith, 1980. An application of principal components analysis to vegetation change in permanent plots. Journal of Ecology 68: 33-41.
- Tacha, T.C., W.D. Warde & K.P. Burnham, 1982. Use and interpretation of statistics in wildlife journals. Wildlife Society Bulletin 10: 355-362.
- Taylor, C.C. & P.A. Burrough, 1986. Multiscale sources of spatial variation in soil III. Improved methods for fitting the nested model to one-dimensional semivariograms. Mathematical Geology 18: 811-821.
- ten Berge, H.F.M., L. Stroosnijder, P.A. Burrough, A.K. Bregt & M.J. de Heus, 1983. Spatial variability of physical soil properties influencing the temperature of the soil surface. Agricultural Water Management 6: 213-216.
- ter Braak, C.J.F., 1982. DISCRIM A modification of TWINSPAN (Hill, 1979) to construct simple discriminant functions and to classify attributes, given a hierarchical classification of samples. Report C 82 ST 10756. TNO Institute of Mathematics, Information processing and Statistics, Wageningen.
- ter Braak, C.J.F., 1983. Principal components biplots and alpha and beta diversity. Ecology 64: 454-462.
- ter Braak, C.J.F., 1985. Correspondence analysis of incidence and abundance data: properties in terms of a unimodal reponse model. Biometrics 41: 859-873.
- ter Braak, C.J.F., 1986a. Canonical correspondence analysis: a new eigenvector technique for multivariate direct gradient analysis. Ecology 67: 1167-1179.
- ter Braak, C.J.F., 1986b. Interpreting a hierarchical classification with simple discriminant functions: an ecological example. In: E. Diday et al. (Editors): Data analysis and informatics 4. North Holland, Amsterdam, p. 11-21.
- ter Braak, C.J.F., 1987a. The analysis of vegetation-environment relationships by canonical correspondence analysis. Vegetatio 69: 69-77.
- ter Braak, C.J.F., 1987b. CANOCO a FORTRAN program for canonical community ordination by [partial] [detrended] [canonical] correspondence analysis, principal components analysis and redundancy analysis (version 2.1) Agricultural Mathematics Group, Wageningen, 95 pp.
- ter Braak, C.J.F., 1988. Partial canonical correspondence analysis. In: H.H. Bock (Editor): Classification and related methods of data analysis. North Holland, Amsterdam, p. 551-558.
- ter Braak, C.J.F. & L.G. Barendregt, 1986. Weighted averaging of species indicator values: its efficiency in environmental calibration. Mathematical Biosciences 78: 57-72.
- ter Braak, C.J.F. & C.W.N. Looman, 1986. Weighted averaging, logistic regression and the Gaussian response model. Vegetatio 65: 3-11.
- Titterington, D.M., G.D. Murray, L.S. Murray, D.J. Spiegelhalter, A.M. Skene, J.D.F. Habbema & G.J. Gelpke, 1981. Comparison of discrimination techniques applied to a complex data set of head injured patients (with discussion). Journal of the Royal Statistical Society, Series A 144: 145-175.

- Toft, C.A. & P.J. Shea, 1983. Detecting community-wide patterns: estimating power strengthens statistical inference. The American Naturalist 122: 618-625.
- Torgerson, W.S., 1958. Theory and methods of scaling. Wiley, New York, 460 pp.
- Tso, M.K-S., 1981. Reduced-rank regression and canonical analysis. Journal of the Royal Statistical Society Series B 43: 89-107.
- Tutin, T.G. et al. (Editor), 1964-1980. Flora Europaea. Cambridge University Press, Cambridge.
- van Dam, H., G. Suurmond & C.J.F. ter Braak, 1981. Impact of acidification on diatoms and chemistry of Dutch moorland pools. Hydrobiologia 83: 425-459.
- van de Nes, T.J., 1980. Introduction to a system approach for water management in Gelderland, The Netherlands. In: Water resources management on a regional
- scale. Committee for hydrological research TNO, Proceedings and Information no 27, p. 9-22.
- van der Maarel, E., 1979a. Multivariate methods in phytosociology, with reference to the Netherlands. In: M.J.A. Werger (Editor): The study of vegetation. Junk, The Hague, pp. 161-226.
- van der Maarel, E., 1979b. Transformation of cover-abundance values in phytosociology and its effects on community similarity. Vegetatio 39: 97-114.
- van der Maarel, E., L. Orlóci & S. Pignatti, 1976. Data processing in phytosociology, retrospect and anticipation. Vegetatio 32: 65-72.
- van der Maarel, E., J.G.M. Janssen & J.M.W. Louppen, 1978. TABORD, a program for structuring phytosociological tables. Vegetatio 38: 143-156.
- van der Meijden, R., E.J. Weeda, F.A.C.B. Adema & G.J. de Joncheere, 1983. Flora van Nederland. 20th Edition. Wolters-Noordhoff, Groningen, 583 pp.
- van der Steen, W.J., 1982. Algemene methodologie voor biologen. Bohn, Scheltema and Holkema, Utrecht, 115 pp.
- van Tongeren, O., 1986. FLEXCLUS, an interactive program for classification and tabulation of ecological data. Acta Botanica Neerlandica 35: 137-142.
- Varmuza, K., 1980. Pattern recognition in chemistry. Springer Verlag, Berlin, 217 pp.
- Vevle, O. & K. Aase, 1980. Om bruk ov oekologiske faktortall i norske plantesamfunn. In: K. Baadsvik, T. Klokk & O.I. Rönning (Editors): Fagmoete i vegetasjonsoekologi pa Kongvoll 16- 18.3.1980. Kgl. Norske Videnskabers Selskab Museet, Botanisk serie 1980-5, p. 178-201.
- Vink, A.P.A., 1963. Planning of Soil Surveys in Land Development. Publication of the International Institute for Land Reclamation and Improvement (ILRI) 10, 53 pp.
- von Humboldt, A., 1806. Ideen zu einer Physiognomik der Gewächse. Cotta, Stuttgart.
- Webster. R., 1984. Elucidation and characterization of spatial variation in soil using regionalized variable theory. In: G. Verly et al. (Editors): Geostatistics for Natural Resources Characterization, Part 2, Reidel, Dordrecht, p. 903-913.

Webster, R., 1985. Quantitative spatial analysis of soil in the field. Advances in

Soil Science, Volume 3. Springer-Verlag, New York.

- Webster, R. & T.M. Burgess, 1983. Spatial variation in soil and the role of kriging. Agricultural Water Management 6: 111-122.
- Webster, R. & T.M. Burgess, 1984. Sampling and bulking strategies for estimating soil properties in small regions. Journal of Soil Science 35: 127-140.
- Webster, R. & S. Nortcliff, 1984. Improved estimation of micronutrients in hectare plots of the Sonning Series. Journal of Soil Science 35: 667-672.
- Westhoff V. & E. van der Maarel, 1978. The Braun Blanquet approach, In: R.H. Whittaker (Editor): Classification of plant communities. Junk, The Hague, p. 287-399.
- Whittaker, J., 1984. Model interpretation from the additive elements of the likelihood function. Applied Statistics 33: 52-65.
- Whittaker, R.H., 1956. Vegetation of the Great Smoky Mountains. Ecological Monographs 26: 1-80.
- Whittaker, R.H., 1967. Gradient analysis of vegetation. Biological Reviews 49: 207-264.
- Whittaker R.H., 1973. Approaches of classifying vegetation. In: Whittaker R.H. (Editor): Ordination and classifiction of communities. Junk, The Hague, p. 325-354.
- Whittaker, R.H., S.A. Levin & R.B. Root, 1973. Niche, habitat and ecotope. American Naturalist 107: 321-338.
- Wiens, J.A. & J.T. Rotenberry, 1981. Habitat associations and community structure of birds in shrub steppe environment. Ecological Monographs 51: 21-41.
- Williams, B.K., 1983. Some observations on the use of discriminant analysis in ecology. Ecology 64: 1283-1291.
- Williams, E.J., 1959. Regression analysis. Wiley, New York, 211 pp.

1

- Williams, W.T. (Editor), 1976a. Pattern analysis in agricultural science. CSIRO Melbourne, Australia.
- Williams, W.T., 1976b. Hierarchical divisive strategies. In: W.T. Williams (Editor): Pattern analysis in agricultural science. CSIRO, Melbourne, Australia.
- Williams, W.T. & J.M. Lambert, 1959. Multivariate methods in plant ecology. I. Association analysis in plant communities. Journal of Ecolology 47: 83-101.
- Williams, W.T. & J.M. Lambert, 1960. Multivariate methods in plant ecology. II The use of an electronic digital computer for association-analysis. Journal of Ecology 48: 689-710.
- Williams, W.T. & J.M. Lambert, 1961. Multivariate methods in plant ecology. III Inverse association-analysis. Journal of Ecology 49: 717-729.
- Williams, W.T., J.M. Lambert & G.N. Lance, 1966. Multivariate methods in plant ecology. V. Similarity-analysis and information-analysis. Journal of Ecology 54: 427-445.
- Williamson, M., 1972. The analysis of biological populations. Arnold, London, 180 pp.
- Wishart,*D., 1978. CLUSTAN user manual. Program library unit. Edinburgh University Press, Edinburgh, 175 pp.

288

Wold, H., 1982. Soft modelling: the basic design and some extensions. In: K.G.

Jöreskog & H. Wold (Editors): Systems under indirect observation. II. North Holland, Amsterdam, p. 1-54.

Yarranton, G.A., 1969. Plant ecology: a unifying model. Journal of Ecology 57: 245-250.

Yarranton, G.A., 1970. Towards a mathematical model of limestone pavement vegetation. III. Estimation of the determinants of species frequencies. Canadian Journal of Botany 48: 1387-1404.

Zelinka, M. & P. Marvan, 1961. Zur Präzisierung der biologischen Klassifikation der Reinheit fliessender Gewässer. Archiv für Hydrobiologie 57: 389-407.

Zhang, L., 1983. Vegetation ecology and population biology of Fritillaria meleagris L. at the Kungsången nature reserve, eastern Sweden. Acta Phytogeographica Suecica 73: 1-92.

×

· 291

.

Ş